# An Analysis of Approaches for Developing National Health Information Systems: A Case Study of Two Sub-Saharan African Countries

by

# THINASAGREE MUDALY

# Submitted in fulfilment of the academic requirements of

# **Master of Science**

School of Mathematics, Statistics and Computer Science

College of Agriculture, Engineering and Science

University of KwaZulu-Natal

Westville Campus

South Africa

April 2016

**PREFACE** 

The research contained in this dissertation was completed by the candidate while based in the

Discipline of Computer Science, School of Mathematics, Statistics and Computer Sciences of

the College of Agriculture, Engineering and Science, University of KwaZulu-Natal, Westville

Campus, South Africa.

The contents of this work have not been submitted in any form to another university and, except

where the work of others is acknowledged in the text, the results reported are due to

investigations by the candidate.

Signed: Mr A. Pillay

Date:

Signed: Dr D. Moodley

Date:

ii

### **DECLARATION 1: PLAGIARISM**

- I, Thinasagree Mudaly, declare that:
- (i) the research reported in this dissertation, except where otherwise indicated or acknowledged, is my original work;
- (ii) this dissertation has not been submitted in full or in part for any degree or examination to any other university;
- (iii) this dissertation does not contain other persons' data, pictures, graphs or other information, unless specifically acknowledged as being sourced from other persons;
- (iv) this dissertation does not contain other persons' writing, unless specifically acknowledged as being sourced from other researchers. Where other written sources have been quoted, then:
- a) their words have been re-written but the general information attributed to them has been referenced;
- b) where their exact words have been used, their writing has been placed inside quotation marks, and referenced;
- (v) where I have used material for which publications followed, I have indicated in detail my role in the work;
- (vi) this dissertation is primarily a collection of material, prepared by myself, published as journal articles or presented as a poster and oral presentations at conferences. In some cases, additional material has been included:
- (vii) this dissertation does not contain text, graphics or tables copied and pasted from the Internet, unless specifically acknowledged, and the source being detailed in the dissertation and in the References sections.

Signed: Thinasagree Mudaly
Date:

## LIST OF PUBLICATIONS

1. T Mudaly, D Moodley, A Pillay, CJ Seebregts; *Architectural Frameworks for Developing National Health Information Systems in Low and Middle Income Countries*; Enterprise Systems Conference (ES), 2013, Cape Town, 2013, pp. 1-9. 978-1-4673-6412-6, IEEE

#### **ABSTRACT**

Health information systems in sub-Saharan African countries are currently characterized by significant fragmentation, duplication and limited interoperability. Incorporating these disparate systems into a coherent national health information system has the potential to improve operational efficiencies, decision-making and planning across the health sector. In a recent study, Coiera analysed several mature national health information systems in high income countries and categorised a topology of the approaches for building them as: top-down, bottom-up or middle-out. Coeria gave compelling arguments for countries to adopt a middle-out approach. Building national health information systems in sub-Saharan African countries pose unique and complex challenges due to the substantial difference between the socioeconomic, political and health landscapes of these countries and high income countries. Coiera's analysis did not consider the unique challenges faced by sub-Saharan African countries in building their systems. Furthermore, there is currently no framework for analysing high-level approaches for building NHIS. This makes it difficult to establish the benefits and applicability of Coiera's analysis for building NHIS in sub-Saharan African countries.

The aim of this research was to develop and apply such a framework to determine which approach in Coiera's topology, if any, showed signs of being the most sustainable approach for building effective national health information systems in sub-Saharan African countries. The framework was developed through a literature analysis and validated by applying it in case studies of the development of national health information systems in South Africa and Rwanda. The result of applying the framework to the case studies was a synthesis of the current evolution of these systems, and an assessment of how well each approach in Coiera's topology supports key considerations for building them in typical sub-Saharan African countries.

The study highlights the value of the framework for analysing sub-Saharan African countries in terms of Coiera's topology, and concludes that, given the peculiar nature and evolution of national health information systems in sub-Saharan African countries, a middle-out approach can contribute significantly to building effective and sustainable systems in these countries, but its application in sub-Saharan African countries will differ significantly from its application in high income countries.

#### **ACKNOWLEDGMENTS**

My sincere thanks to the people who assisted me in this study, including, but not limited to: Nishan Singh, for convincing me to take a break from work and study again, and then constantly reminding me thereafter, to hurry up and finish, as there was so much work I still have left to do:

Echo, for waiting patiently by my side for her day in the park;

Cadessa Naicker, for running my errands so I could study;

Nishan and Preeth Singh, for bearing my burden and defending my honour, giving me the time to study;

The numerous people that contributed to my education, without which this level of research would not have been possible. This group, led by my parents, includes all the people in my personal, student and work life from whom I have learnt;

The community of thinkers, whose work has been freely shared, and upon which my research was based and developed. I have gained knowledge and insight from their ideas. I have explored paths of thought I would not ordinarily have. I hope that I, in return for all that I have gained, am able to make some contribution to this body of knowledge, with this study and in future;

Mr. A. W. Pillay, Dr D. Moodley and Prof C.J. Seebregts, my supervisors, for their expert guidance throughout my study, to a light at the end of a tunnel I often found quite dark;

Ravi Ramnath, for his patience and attention to detail, in reviewing my dissertation;

G. Gurudas, for her help with the administration aspects of my study;

Financial support from the HeAL (Health Architecture Laboratory – a research laboratory focused on the application of enterprise architecture and health informatics to resource-limited settings) established at UKZN with funding from the Rockefeller Foundation under Grant Number: 2010 THS 347 (Establishing a Health Enterprise Architecture Lab); and the International Development Research Centre (IDRC) Grant Number: 105708 (Health Enterprise Architecture Laboratory, - Developing Capacity, Sharing Knowledge and Good Principles Across eHealth in Africa.).

# **TABLE OF CONTENTS**

	<b>Page</b>
PREFACE	ii
DECLARATION 1: PLAGIARISM	iii
LIST OF PUBLICATIONS	iv
ABSTRACT	v
ACKNOWLEDGMENTS	vi
TABLE OF CONTENTS	vii
LIST OF TABLES	xi
LIST OF ACRONYMS	xii
CHAPTER 1: INTRODUCTION	1
1.1 Problem Statement	2
1.2 Research Aim and Objectives	2
1.3 Research Questions	3
1.4 Methodology	4
1.5 Contributions	5
1.6 Structure of the thesis	6
CHAPTER 2: MATERIALS AND METHODS	7
2.1 Analysis Framework Design	8
2.2 Analysis Framework Design	9
2.3 Case Study Design	11
2.3.1 Case Selection	12
2.3.2 Interviews	13
2.3.3 Case Study Research Questions	
2.3.4 Analysis	16
2.4 Literature Review and Data Collection	17
2.5 Validity	
2.6 Summary	19
CHAPTER 3: Coiera's Topology of Approaches for Building NHIS	20
3.1 The Role of eHealth	20
3.2 The nature of an NHIS	21
3.2.1 Electronic Health Records	22

3.3	NHIS in High Income Countries	23
3.4	The Challenges with Developing NHIS	24
3.5	Coiera's Topology of Approaches for Building NHIS	25
3.6	An Analysis of the Differences between the Approaches	26
3.6.1	Technical Architecture	27
3.6.2	Organisation	28
3.6.3	Implementation Strategy	29
3.7	Summary	31
СНАРТЕІ	R 4: NHIS Development in SSAC	33
4.1	NHIS in Sub-Saharan African Countries	33
4.2	The Benefits of Developing NHIS in SSAC	35
4.3	Adopting an Approach from HIC	36
4.4	The Challenges with Developing NHIS in SSAC	37
4.4.1	Resource Constraints	37
4.4.2	Socio-political Environment with Distributed Autonomy	38
4.4.3	Dynamic, Heterogeneous Systems	39
4.4.4	Inadequate Systems Design	40
4.5	Key Considerations for Developing NHIS in SSAC	40
4.5.1	Capacity Maximisation and Building-dealing with unstable, constrained	funding
	41	
4.5.2	A systems thinking approach – dealing with system complexity	41
4.5.3	Interoperability – dealing with heterogeneity	42
4.5.4	Collaboration and Coordination – dealing with organisational complex	kity and
autor	nomy	42
4.5.5	Incremental and Iterative Progress – dealing with resource constraints	43
4.5.6	Flexible, Agile and Adaptable Systems - dealing with dynamism	43
4.6	Summary	44
CHAPTEI	R 5: Case Study - South Africa	45
5.1	The Organisation and Governance of the Health Sector	45
5.2	The Organisation and Governance of the NHIS	47
5.2.1	Policies	47
5.2.2	Standards	48
5.3	The Key Information Systems within the NHIS	48
5.3.1	Population-based HIS	49

	5.3.2	Patient-based HIS	50
	5.3.3	Health Information Exchange	51
	5.3.4	Private Sector Systems	51
	5.4	Challenges with Developing the NHIS	52
	5.5	The Evolution of the NHIS	53
	5.5.1	The Past	53
	5.5.2	The National eHealth Strategy	54
	5.5.3	The Future	55
		5.5.3.1 Population-based Systems	55
		5.5.3.2 Patient-based HIS	56
		5.5.3.3 EHR and HIE	56
		5.5.3.4 Standards	57
	5.6	Summary	58
C	HAPTE	R 6: Case Study – Rwanda	59
	6.1	The Organisation and Governance of the Health Sector	59
	6.2	The Organisation and Governance of the NHIS	61
	6.1.1	Policies	62
	6.1.2	Standards	63
	6.3	The Key Information Systems within the NHIS	63
	6.1.3	Population-based HIS	63
	6.1.4	Patient-based HIS	65
	6.1.5	Health Information Exchange	65
	6.1.6	Private Sector Systems	65
	6.4	Challenges with Developing the NHIS	66
	6.5	The Evolution of the NHIS	66
	6.1.7	The Past	66
	6.1.1	The National eHealth Strategy	68
	6.1.8	The Future	70
	6.6	Summary	70
C	HAPTE	R 7: RESULTS AND DISCUSSION	71
	7.1	A Technical Architecture Perspective	71
		7.1.1.1 Evolution of the NHIS in South Africa and Rwanda	72
		7.1.1.2 Dealing with an Evolving and Heterogeneous NHIS	75
	7.2	An Organisational Perspective	78

7.2.1	Dealing with Organisational Complexity and Autonomy	79
7.3	An Implementation Strategy Perspective	80
7.3.1	Dealing with Unstable Funding and Resource Constraints	80
7.3.2	Dealing with Dynamism	82
7.3.3	Dealing with System Complexity	83
7.4	A middle-out approach for building NHIS in SSAC	84
7.4.1	Implementing a middle-out approach in SSAC	85
7.5	Summary	85
CHAPTER	8 8: CONCLUSION	88
8.1	Summary of findings	88
8.2	Limitations and Future Work	89
REFEREN	ICES	91

# LIST OF TABLES

<u>Table</u>				<u>P</u>	age
	Characteristics Building NHIS				
Approaches for	Duilding NAIS	 ••••••	 • • • • •	 •••••	. 31

# LIST OF ACRONYMS

Acronym	Meaning
ACM	Association for Computing Machinery
AIDS	Acquired Immune Deficiency Syndrome
CSRQ	Case Study Research Question
DHIS	District Health Information System
EA	Enterprise Architecture
EHR	Electronic Health Record(s)
EMR	Electronic Medical Record(s)
EU	European Union
GDP	Gross Domestic Product
HIE	Health Information Exchange(s)
HIS	Health Information System(s)
HIV	Human Immunodeficiency Virus
HMIS	Health Management Information System
HMN	Health Metrics Network
ICT	Information and Communication Technology
IEC	International Electrotechnical Commission
IEEE	Institute of Electrical and Electronics Engineers
ISO	International Organization for Standardization
ISO/TC	ISO Technical Committee
IT	Information Technology
LMIC	Low and Medium Income Country(/ies)
МОН	Minister of Health
NDOH	National Department of Health
NGO	Non-Government Organisation(s)
NHIS	National Health Information System(s)
NHS	National Health Service
OpenMRS	Open Medical Record System
RQ	Research Question
SCR	Summary Care Record(s)
SEHR	Shared Electronic Health Record(s)
SSA	sub-Saharan African

SSAC	sub-Saharan African Country(/ies)
ТВ	Tuberculosis
UK	United Kingdom
USA	United States of America
WHO	World Health Organisation

Acronym	Meaning in the Context of Rwanda
RHEA	Rwanda Health Enterprise Architecture

Acronym	Meaning in the Context of South Africa
DHMIS	District Health Management Information System
HNSF	Health Normative Standards Framework
LOGIS	Logistical Information System
NHC/MIS	National Healthcare Management Information System
NHI	National Health Insurance
NHIRD	National Health Information Repository and Data Warehouse
PHC	Primary Healthcare

## **CHAPTER 1: INTRODUCTION**

Health information systems (HIS) manage and provide the information required for improving the effectiveness and efficiency of health systems [1]. However, many HIS in sub-Saharan African Countries (SSAC) are frequently characterized by significant fragmentation, duplication and limited interoperability, making them ineffective in meeting the requirements of the health system. Incorporating these disparate systems into coherent National Health Information Systems (NHIS) has the potential to improve operational efficiencies, decision-making and planning across the health sector. Many SSAC are, therefore, attempting to build their NHIS, but are struggling to overcome specific challenges with its development. Health information systems remain predominantly paper-based in SSAC. In contrast to SSAC, HIC have made substantial progress in developing their NHIS [2]. Patient-based HIS, dealing with detailed, granular records such as the patient's profile and health encounters, are now ubiquitous at facility level in HIC. In addition, HIC with mature NHIS have already started implementing advanced, national patient-based systems such as electronic health record (EHR) systems and self-service portals [3].

One of the key reasons for SSAC struggling with their NHIS is the lack of a sustainable, high-level approach for building an NHIS that effectively meets the requirements of the health system it serves. Building and successfully establishing socio-technical systems, as complex as an NHIS, requires an overarching national approach that includes more than just guidelines for software design and engineering, but also addresses the complex mix of social, economic and political concerns. In a recent analysis of mature NHIS in HIC, Coiera [4] highlighted the importance of the high-level approaches used for building NHIS, and proposed a topology that distinguishes between three types of approaches, viz. top-down, bottom-up and middle-out. He also gives compelling arguments for countries to take a middle-out approach.

Coiera's analysis, however, did not consider the unique challenges faced by SSAC in building their NHIS. Most SSAC [5] share, though in varying degrees, common problems [6] such as: widespread and chronic absolute poverty, increasing disparities in income and between urban and rural standards of living and economic opportunities, and increasing dependence on foreign resources and often inappropriate systems [7]. Health systems are weak and inequitable. A

larger percentage of the population is dependent on the public and rural systems. However, doctors and nurses are concentrated in urban areas, and private and urban health services are often far superior to public and rural health services. Mortality, communicable disease and chronic non-communicable disease levels are high - some at pandemic level [8]. In contrast, in high income countries (HIC), mortality [9] and infectious diseases levels have declined, and health threats now come from chronic and degenerative diseases such as cancer, and stroke [8]. So, while SSAC still struggle with public health issues, many HIC have progressed to focusing on providing patient-centric health care [10].

Heavily overburdened by disease [11], most SSAC have so far focused mainly on implementing population-based HIS that combine public health data from multiple sources [10]. Some SSAC are now trying to implement patient-based HIS for use at point of care for the management of diseases such as HIV. However, given the lack of resources and increasing burden and urgency of public health issues, developing patient-based HIS is of a lower priority to population-based HIS in SSAC.

The selection of an approach for building NHIS in SSAC must take into account the specific socio-economic and political conditions of SSAC and their health landscapes. It must also give due priority to the development of population-based systems and more widespread implementation of patient-based systems.

#### 1.1 Problem Statement

Given the substantial differences between the health landscapes of HIC and SSAC, there is a need to assess the applicability and benefits of Coiera's analysis for building electronic NHIS in SSAC. However, no theoretical framework currently exists for analysing high-level approaches for building NHIS, which makes it difficult to establish the benefits and applicability of Coiera's approaches in SSAC.

## 1.2 Research Aim and Objectives

The research aimed to develop and apply a framework for (1) characterising the high level approach taken by an SSAC in developing its NHIS, (2) determining which approach from

Coiera's topology is followed by an SSAC, and (3) assessing the consequences of an SSAC taking such an approach..

The following main objective of the study covered the four general types of purposes for research viz. descriptive, exploratory, explanatory and improving [12]:

- Develop an analytical framework for characterising the high-level approach taken by an SSAC in developing its NHIS, and use it to describe the evolution of NHIS in SSAC
- Use the framework to determine whether an SSAC is taking an approach from Coiera's topology and explain the differences between the NHIS development landscapes of HIC and SSAC
- Use the framework to assess the consequences of an SSAC taking an approach from Coiera's topology and explore the consequences of taking a top-down, bottom-up and middle-out approach, respectively, for building NHIS in SSAC
- By evaluating these consequences determine which approach, if any, showed signs of being the most sustainable approach for building NHIS that effectively meet the requirements of health systems in SSAC.

#### 1.3 Research Questions

An analysis of these complex and abstract objectives resulted in the following sub-questions that guided the research:

### (RQ1) What is the nature of the approaches in Coiera's topology?

RQ1.1.Which aspects of the systems and software engineering practice are influenced by the approaches?

RQ1.2. Which aspects of the NHIS are influenced by the approaches?

# $(RQ2)\ Are\ there\ fundamental\ differences\ between\ the\ development\ of\ NHIS\ in\ HIC\ and\ SSAC?$

- RQ2.1.What are the distinguishing characteristics of SSAC and HIC in the context of this study?
- RQ2.2.What is the perspective/scope/interpretation of the NHIS and its development in SSAC and HIC?
- RQ2.3. What aspects of the development of NHIS are common to both HIC and SSAC?

RQ2.4.What are the fundamental differences between HIC and SSAC, in terms of (a) health system objectives and strengthening initiatives, (b) the current state and evolution of NHIS, and (c) the development of NHIS?

#### (RQ3) What are the consequences of applying Coiera's framework for building NHIS in SSAC?

- RQ3.1.What are the key characteristics of each approach?
- RQ3.2.Are SSAC naturally evolving towards any of these approaches?
- RQ3.3.Should NHIS in SSAC be developed using any of these approaches?
- RQ3.4. What interventions are necessary to implement such an approach in SSAC?

#### 1.4 Methodology

An analysis of the literature was undertaken and used to develop a theoretical framework for analysing high level approached to building NHIS in SSAC. The approach was validated on two case studies.

Given the relative immaturity and inherent complexity of NHIS in SSAC, the scarcity of studies that analyse approaches for building NHIS in SSAC and the lack of a framework for such analyses, a case study approach was adopted. South Africa and Rwanda were selected as the cases to study because each has made great strides towards building their NHIS. Through an analysis of the literature, a theoretical framework was developed for the analysis of the case studies. It comprises of (1) the key characteristics distinguishing the approaches in Coiera's topology from each other, and (2) the key considerations for building NHIS in SSAC, in comparison to HIC. The development of NHIS in each case was analysed in terms of this framework. The framework was then also used to analyse how well each approach in Coiera's topology supports the key considerations for building NHIS in SSAC. The results of this analysis were then assessed with the aim of determining which approach showed signs of being the most sustainable for building NHIS in SSAC.

Paper based systems still play an important role in existing health information systems in SSAC. This study, however, concentrates mainly on electronic HIS due to their increasing importance in developing NHIS in SSAC. In general, throughout the dissertation, unless specifically qualified to be paper-based, references to HIS are assumed to be referring to electronic HIS.

#### 1.5 Contributions

The key contribution made by the study was the development of a theoretical framework based on Coiera's topology of approaches, for (1) characterising the high-level approach taken by an SSAC in developing its NHIS, (2) determining whether an SSAC is taking an approach from Coiera's topology, and (3) assessing the consequences of an SSAC taking an approach from Coiera's topology.

The framework comprises of (1) the key characteristics for distinguishing the approaches in Coiera's topology from each other, and (2) the key considerations (or development strategies) for building NHIS in SSAC, in comparison to HIC.

The development of the framework was guided by studying the key challenges faced by SSAC when building NHIS. While much literature exists, the generally accepted set of key challenges with developing NHIS in SSAC has not been consolidated. Proposed solutions for building NHIS in SSAC cannot be properly evaluated without first understanding these challenges.

The validity of the framework was established by applying the framework to two case studies. The application of the framework established the consequences of applying Coiera's topology of approaches for building NHIS in SSAC, a previously established theory based only on an analysis of implementations in HIC. While it was found that a middle-out approach can contribute significantly to building effective NHIS in a sustainable manner, given the nature and evolution of NHIS in SSAC, its application in SSAC will be significantly different to its application in HIC.

The study provided an in-depth analysis of NHIS development in South Africa and Rwanda. It also established that NHIS in SSAC generally evolve through three stages. Understanding this evolution may give a SSAC some indication of how its NHIS may evolve in the future. The theoretical framework developed as part of this study can be used to study other SSAC and be extended for application to low and middle income countries (LMIC). The analysis that emerged from the study may assist a SSAC in developing its own NHIS, and aid in the development of frameworks for building NHIS in SSAC. Global health organisations such as the WHO, Ministries of Health in SSAC, donors and research organisations interested in building NHIS in SSAC, may be guided by this study to advise SSAC on how to practically implement their NHIS.

#### 1.6 Structure of the thesis

The dissertation is structured as follows:

Chapter 2 describes the research design and methods used.

The analysis of the literature is covered in chapters 3 and 4.

Chapter 3 describes the nature of NHIS, Coiera's topology of approaches for building NHIS in HIC, and the key characteristics distinguishing the approaches from each other (addressing RQ1).

Chapter 4 describes the nature of NHIS in SSAC, as well as the key challenges with, and the key considerations for developing NHIS in SSAC (addressing RQ2).

Chapter 5 and 6, describe the current state of NHIS and its development in South Africa and Rwanda, respectively (addressing RQ2).

Chapter 7 presents and discusses the research findings, including the current evolution of NHIS in SSAC, and the consequences of adopting an approach from Coiera's topology for developing NHIS in SSAC (addressing RQ3).

Chapter 8 concludes the study (addressing RQ3).

### **CHAPTER 2: MATERIALS AND METHODS**

The aim of this research was to analyse the possible consequences of adopting an approach from Coiera's topology for building NHIS in SSAC. The methodology was designed to first answer the research questions. Then, analysing the answers to these questions determine which approach, if any, showed signs of being the most sustainable approach for building NHIS that effectively meet the requirements of health systems in SSAC.

NHIS are inherently complex, multi-dimensional, socio-politico-technical systems that are relatively immature in SSAC. The significance and contemporaneousness of the field of NHIS development is evident in the current, and high activity in the field – in the way of research, 'industrial' projects and discussion. However, there is a scarcity of studies that analyse approaches for building NHIS [6], specifically in SSAC, and a lack of mature research tools for their study. Thus, a case study approach was deemed most appropriate to achieve the research objectives.

A theoretical framework for the analysis of the case studies was required. However, since theories in the field of NHIS development are underdeveloped at present, an analysis framework was developed for the study by an extensive analysis of the literature. This framework, comprises of (1) the key characteristics for distinguishing the approaches in Coiera's topology from each other, and (2) the key considerations for building NHIS in SSAC, in comparison to HIC. The validity of the framework was established by applying it to two case studies. The framework was used (1) to develop the case study research questions which were used to analyse NHIS development in the two cases studied i.e. South Africa and Rwanda, and (2) to analyse how well each approach in Coiera's topology supports the key considerations for building NHIS in SSAC.

This chapter is structured as follows:

 Analysis Framework Design, describes the development of the analytical framework used in the study

#### 2.1 Analysis Framework Design

A general framework for the analysis of high-level approaches for the development of NHIS is lacking. An analysis framework had to be developed to objectively analyse the cases.

The key characteristics distinguishing the approaches in Coiera's topology from each other formed the first part of the framework. A foundational understanding of the state of the art of the NHIS development landscape was developed through a literature search. (Search term used: "nation health information systems". Generally, when searching for information related to the NHIS the search was also extended to include the following terms: "health information system", "eHealth", "electronic health records".) A study of the middle-out approach, more generally, in software engineering highlighted the key specialisations of the approaches in Coiera's topology, answering RQ1.1. Which aspects of the systems and software engineering practice are influenced by the approach. (Search terms: "middle-out software engineering", "middle-out architecture", "emergent enterprise software systems".) The key differentiating characteristics of each approach were then identified and formalised [13, 14], mainly from an analysis of [4], and papers documenting further insights from NHIS implementations in the UK, Australia, USA, New Zealand, and Greece, answering RQ1.2. Which aspects of the NHIS are influenced by the approaches.

Given that Coiera abstracted the middle-out approach from his analysis of NHIS development in HIC only, it was important to understand the fundamental differences between the NHIS and development of NHIS in HIC and SSAC. A Google search provided definitions for "sub-Saharan African countries" and "high income countries". The nature of the NHIS development landscape in HIC, in which the approaches in Coiera's topology have actually been implemented, and the practical implications of implementing these approaches was studied first. (Search terms: "health systems in developed countries", "national health information systems in developed countries"). Then the landscape of NHIS development in SSAC (a specialised case of developing countries) was studied. (Search terms: "national health information systems in" + "developing countries"). This analysis served to scope-out and clarify the nature of the problem areas that need to be addressed by the study, *answering RQ2.2.What is the perspective/scope/interpretation of the NHIS and its development in SSAC and HIC*.

The key considerations for building NHIS in SSAC, formed the second part of the analysis framework. The key challenges that SSAC face were identified. (Search terms: "challenges with national health information systems in" + "developing countries"/"sub-Saharan Africa"/"Rwanda"/"South Africa"). From the same set of papers, the key considerations for building NHIS in SSAC, in particular in mitigating the identified key challenges were also elicited. The data was evaluated by showing how large numbers of different previously observed challenges could be uniformly explained, and addressed [12, 13]. The data was also analysed for common patterns in NHIS development landscapes in HIC and SSAC. This provided a motivation for considering the approaches tested in HIC for application in SSAC. This analysis provided answers to RQ2.3.What aspects of the development of NHIS are common to both HIC and SSA, and RQ2.4.What are the fundamental differences between HIC and SSAC.

- Case Study Design, describes the design of the case studies, the results of which are presented in chapters 5 and 6 of the dissertation.
- Literature Review and Data Collection, describes the techniques used to collect and analyse the data, the results of which are presented in chapter 7 of the dissertation.
- The chapter concludes with a summary and discussion of the validity of the results of the study.

## 2.2 Analysis Framework Design

A general framework for the analysis of high-level approaches for the development of NHIS is lacking. An analysis framework had to be developed to objectively analyse the cases.

The key characteristics distinguishing the approaches in Coiera's topology from each other formed the first part of the framework. A foundational understanding of the state of the art of the NHIS development landscape was developed through a literature search. (Search term used: "nation health information systems". Generally, when searching for information related to the NHIS the search was also extended to include the following terms: "health information system", "eHealth", "electronic health records".) A study of the middle-out approach, more generally, in software engineering highlighted the key specialisations of the approaches in Coiera's topology, answering RQ1.1. Which aspects of the systems and software engineering practice

are influenced by the approach. (Search terms: "middle-out software engineering", "middle-out architecture", "emergent enterprise software systems".) The key differentiating characteristics of each approach were then identified and formalised [13, 14], mainly from an analysis of [4], and papers documenting further insights from NHIS implementations in the UK, Australia, USA, New Zealand, and Greece, answering RQ1.2. Which aspects of the NHIS are influenced by the approaches.

Given that Coiera abstracted the middle-out approach from his analysis of NHIS development in HIC only, it was important to understand the fundamental differences between the NHIS and development of NHIS in HIC and SSAC. A Google search provided definitions for "sub-Saharan African countries" and "high income countries". The nature of the NHIS development landscape in HIC, in which the approaches in Coiera's topology have actually been implemented, and the practical implications of implementing these approaches was studied first. (Search terms: "health systems in developed countries", "national health information systems in developed countries"). Then the landscape of NHIS development in SSAC (a specialised case of developing countries) was studied. (Search terms: "national health information systems in" + "developing countries"). This analysis served to scope-out and clarify the nature of the problem areas that need to be addressed by the study, *answering RQ2.2.What is the perspective/scope/interpretation of the NHIS and its development in SSAC and HIC*.

The key considerations for building NHIS in SSAC, formed the second part of the analysis framework. The key challenges that SSAC face were identified. (Search terms: "challenges with national health information systems in" + "developing countries"/"sub-Saharan Africa"/"Rwanda"/"South Africa"). From the same set of papers, the key considerations for building NHIS in SSAC, in particular in mitigating the identified key challenges were also elicited. The data was evaluated by showing how large numbers of different previously observed challenges could be uniformly explained, and addressed [12, 13]. The data was also analysed for common patterns in NHIS development landscapes in HIC and SSAC. This provided a motivation for considering the approaches tested in HIC for application in SSAC. This analysis provided answers to RQ2.3. What aspects of the development of NHIS are common to both HIC and SSA, and RQ2.4. What are the fundamental differences between HIC and SSAC.

### 2.3 Case Study Design

The aim of the case studies [12], was to gain an in-depth understanding of the NHIS development landscape in SSAC, giving due consideration to the contextual factors that influence it and to validate the analysis framework developed by the study. NHIS are complex systems with many, functionally as well as structurally, interconnected human and non-human components, which may interact unpredictably [15]. The development of NHIS in SSAC is also heavily influenced by social, political, policy, as well as technical issues, making them difficult to study [16].

Case studies have been criticized for being of little value and not generalizable [12]. However, the development landscape of NHIS in SSAC is complex, non-deterministic and emerging – one that cannot be carefully controlled for, say, a statistical evaluation. Lucas [17] found that initial small-scale success in HIS development, from artificially controlled research, does not guarantee sustainable large-scale benefits [18-20]. The growing number of failed pilot projects in SSAC confirm that the issues are themselves complex, and cannot be feasibly controlled or replicated [21]. Furthermore, this study is not focused on the static design of the NHIS for or at a particular point in time. Its concerns include the manner and method in which the NHIS changes over time. For example, the study allows for knowledge of how and why the NHIS in SSAC have changed in the past, to inform the way in which they will evolve in the future [22-24]. The case study approach [12] was therefore appropriate for understanding and explaining the contemporary complexities involved in building NHIS in SSAC, and their impact on the resulting NHIS [19].

Researcher bias [12] is also an important factor that may impact negatively on the reliability of case study research. The researcher's bias was generally mitigated by triangulation throughout the study:

- Multiple research methods, viz. literature analysis and case studies, were used to research the emerging phenomena associated with complex ICT initiatives such as building NHIS in SSAC.
- More than one data source was used for critical data. This is evident in the multiple references supporting key statements.
- More than one method of data collection was used. Data was collected from documented text, as well as formal and informal interviews and discussions with experts in the field.

Different viewpoints and rival theories were taken into account and investigated. For
example, some sources claimed that an NHIS could hugely benefit SSAC despite their
low resources, while others were sceptical about the value that NHIS could offer SSAC.
Rivals theories were evaluated in answers to each of the research questions e.g. in
comparing the consequences of implementing alternative approaches for developing
NHIS in SSAC.

#### 2.3.1 Case Selection

Rwanda and South Africa were the cases chosen to be studied because they provide a good cross-section of contrasting features which help to generalise the common findings to other SSAC. The following distinguishing features of each country influenced their choice:

- In terms of availability of resources, Rwanda is a low income country, while South
  Africa is a middle-income country. This characteristic also makes them critical cases,
  as, in terms of resource availability, most other SSAC would fall in the range between
  them.
- Rwanda and South Africa generally exhibit varying levels of maturity with respect to their NHIS and their development. Each case also offered valuable data not available in the other. For example, Rwanda has already adopted an EMR system for country-wide rollout, and moving towards connecting them through its national health information exchange. On the other hand, South Africa has moved ahead in linking its aggregateddata systems across the country.
- Both countries generally share the challenges with building NHIS, as well as the key considerations for mitigating these challenges, identified in the literature review.

The lack of clear boundaries between the phenomenon (e.g. the approach for building the NHIS) and context (the NHIS) of the study was evident, from the lack of clear, accepted definitions, specifications and frameworks in the fields of eHealth and ICT. The case studies helped to clearly and explicitly define for SSAC, the key characteristics of the NHIS environment, the NHIS, the evolution of NHIS, and the approach for developing NHIS. This ability to make explicit the boundary of the cases proved the case studies to be exemplary in the context of SSAC [25].

#### 2.3.2 Interviews

Interviews were conducted to help corroborate some of the findings of the case study research questions, with the experience of experts in the field, who generally know what's actually happening in practice. The goal of the interviews was to 'see' the interviewee's perspective on the NHIS, its evolution and its future. The field experts provided insights from their first-hand experience in planning for and building national-level HIS in the cases studied. They were able to explain how and why they came to have particular perspectives, comment on the generalizability of previously obtained qualitative findings [26], help to correct some researcher bias, and verified to some degree the quality of the documents studied i.e. to what extent interviewees agree with the contents [12]. The interviews also allowed for specific conflicting or ambiguous topics related to the case study to be explored in order to make explicit the boundary of the cases [26]. For example, it was found that experts differ in their interpretation of the terms NHIS and eHealth.

Skype-interviews were done because a postal survey would likely have had no response [26]. While the aim of the interviews was to focus on eliciting information on specific situations in the world of the interviewee, given the limited information available on the research topic, abstractions and general opinions were also allowed [26]. Since the nature and range of participants' opinions about the research topic could not easily be quantified in a structured open-response interview, the interviewee was encouraged to actively shape the course of the interview rather than passively respond to the interviewer's questions [26]. This semi-structured approach was taken because the interview itself was not testing any formal hypothesis, and the researcher did not know in advance the amount and type of factual information the participants would be able to provide.

The interviews were guided by the following questions:

1. What has been your involvement in eHealth related projects in <Rwanda/South Africa>

(CSRQ3) What is the nature and state of the NHIS IT architecture?

- 2. Does < Rwanda/South Africa > have an NHIS? If so, what are the main information systems comprising the NHIS?
- 3. How widespread are electronic medical record (EMR) systems deployed across the country -<30% or >90%?
- 4. Do these systems exchange EMR across facilities?

# (CSRQ2) How is the NHIS organised and controlled

- 5. What are the key roles and responsibilities of these entities in terms of the NHIS
- 5.1. each level of the health sector the states/provinces, local government and individual facility level and
- 5.2. the private and non-government organisations,
- 6. How much autonomy do these entities have in the procurement, designing and implementing HIS that manage data critical to an NHIS?

### (CSRQ4) How is the NHIS evolving?

- 7. What were the significant milestones in the evolution of the NHIS?
- 8. Does < Rwanda/South Africa > have a vision for its NHIS? If so,
- 8.1. What are the main components of this vision?
- 8.2. What will national EMR systems look like in 5 years' time?
- 8.3. Is a shared health record desirable? How much of a priority is it?
- 9. What are the 5 main challenges with NHIS development?
- 9.1. Is a shared health record achievable?
- 10. What are my best sources of information related to < Rwanda/South Africa >'s NHIS and its development documents and/or people?

The following people, with key knowledge of the Rwandan and South African NHIS, were interviewed:

- Mr Emmanuel Rugomboka, Jembi<sup>1</sup> country coordinator in Rwanda, for his knowledge of HIS development in Rwanda, on 13/05/2015
- Mr Gaurang Tanna, Director of HIS, NDOH, for his knowledge of HIS development in SA, on 25/06/2015

The transcripts of interviews were sent to the respective interviewees for review, to confirm that their responses were not misunderstood and still stand.

<sup>&</sup>lt;sup>1</sup> Jembi Health Systems NPC, is a nonprofit organisation that works in developing countries in Africa and focuses on the development of eHealth and HIS.

#### 2.3.3 Case Study Research Questions

The main objective of the case studies was to gain the knowledge required to answer the research questions (search terms: "health systems in" / "national health information systems in" + "Rwanda"/"South Africa"). In particular, the case studies provided evidence to support answers to RQ2.3. What aspects of the development of NHIS are common to both HIC and SSA, and RQ2.4. What are the fundamental differences between HIC and SSAC.

The analysis framework was used to translate the research questions into the following case study research questions:

- (CSRQ1) How is the health sector organised and controlled?
  - What is the country's regional breakdown?
  - o Do patients move between care centres?
  - What resource constraints does the health sector experience?
  - What are the key problems faced by the health sector?
  - What are the objectives of the health system
  - What are the key roles and responsibilities in managing the health sector both official and operational?
  - What is the role of the non-government and private organisations in the health sector?
- (CSRQ2) How is the NHIS organised and controlled?
  - o Is the control of the NHIS centralised or decentralised?
  - Who controls the budget for HIS?
  - o Is the NHIS implemented by the execution of a plan?
    - If a plan exists, (a) does the plan involve large scale rollouts or smaller incremental changes, (b) is the plan controlled and executed by the centralised organisation, (c) how much autonomy does each level of the health sector have in the execution of the plan, (d) which of the following are in place or planned: an eHealth strategy, an interoperability framework, an enterprise architecture, HIS standards and policies?
  - o How are HIS standards and policies enforced? (Not just the planned enforcement, but what is actually happening on the ground)
    - Who decides which systems are deployed?

- How much autonomy does (1) each level of the health sector the states/provinces, local government and individual facility level and (2) the private and non-government organisations, have in the procurement, designing and implementing HIS?
- Are private and non-government organisations at liberty to implement their own systems or do they need to implement a government or government-endorsed system?
- (CSRQ3) What is the nature and state of the NHIS IT architecture?
  - O What HIS are in place?
  - What is the nature of these HIS?
    - Are they regional or disease specific?
    - Are they based on a uniform IT architecture?
    - Are NHIS architecturally centralised or decentralised?
  - What are the drivers for exchanging information between systems within the NHIS?
  - What is the health sector's requirements for the integration and interoperability of systems within the NHIS?
  - o Is there a health information exchange? What is the level of integration between these systems with respect to the exchange of EMR?
- (CSRQ4) How is the NHIS evolving?
  - How did the NHIS evolve to its current state?
  - O What are the key challenges with developing NHIS?
  - What is the future of the NHIS?

#### 2.3.4 Analysis

The study of each case, based on the case study research questions, was reported on using the following linear-analytic report structure:

- 1. Introduction
- 2. The Organization and Governance of the Health Sector (addressing CSRQ1)
- 3. The Organization and Governance of the NHIS (addressing CSRQ2)
- 4. The Key Information Systems within the NHIS (addressing CSRQ3)
- 5. Challenges with developing the NHIS (addressing CSRQ4.2)
- 6. The Evolution of the NHIS (addressing CSRQ4)

#### 7. Conclusion

Abstractions and inferences were then drawn from the cases studies to critically analyse the current NHIS development practices in SSAC, with an aim of improving them [12], (addressing (RQ3) What are the consequences of applying Coiera's framework for building NHIS in SSAC). By analysing events related to HIS development in SSAC over time [12], a model of the evolution of NHIS in SSAC was formulated. The model highlights the introduction of different types of HIS at each stage in the evolution of NHIS in SSAC. The various previously observed implementations of the different types of HIS, could now be uniformly explained [13] by this model. This analysis answered RQ3.2.Are SSAC naturally evolving towards any of these approaches.

The key characteristics distinguishing the approaches from each other, were analysed against the development of NHIS in SSAC, based on the South Africa and Rwanda case studies. The top-down, bottom-up and middle-out approaches were then critically evaluated in terms of how well each supports the key considerations for building NHIS in SSAC. Conclusions were then drawn from this analysis on the approach that showed signs of being the most sustainable for developing NHIS in SSAC, and the interventions necessary to implement this approach in SSAC. This addressed RQ3.3.Should NHIS in SSAC be developed using any of these approaches and RQ3.4.What interventions are necessary to implement such an approach in SSA.

#### 2.4 Literature Review and Data Collection

Qualitative secondary data was mainly collected using an independent, third-degree method [12] from focused internet searches and experts in the field of HIS development in SSAC. The following repositories were searched, with preference given to data from the last ten years: Google & Google Scholar, Science Direct, IEEE Xplore, ACM Digital Library, Pubmed, amongst others on a more ad-hoc basis. A potential limitation is the bias in collecting data (the limited number databases searched, and the exclusion of non-English data) and the inclusion of grey literature (documents and web pages).

An unstructured, appreciative, mode of inquiry was adopted in reviewing data, focusing on the nature of a problem, without quantifying it. Analysis was carried out in parallel with the data

collection, to allow the analysis to guide the collection of additional data [12, 27]. An objective, open coding scheme was applied to the data, with the aim of identify patterns, sequences and relationships [12, 28]. The data and annotations were clustered around their relevant codes in a word document. The initial codes (short phrases depicting themes or concepts [29]) were based on the search terms defined in the analysis framework design. The less formal, more intuitive initial approach evolved to a more systematic and objective discovery during subsequent content analyses, as the themes developed [29]. The key themes eventually became the headings within the dissertation [12].

The iterative analysis of the literature allowed for the inclusion of observations that were not anticipated. To ensure that the analysis did not only include material supporting the hypotheses [29], a negative case analysis technique was used to find alternate explanations and theories. Cases were sought that did not fit an observation, and discarded with a documented reason for its exclusion or the observation reformulated to account for it, until "practical certainty" of the observation was attained [29].

No audit trail was maintained. No ethical issues needed consideration.

#### 2.5 Validity

Since this study relied primarily on qualitative data and only theoretical research [30], a key weakness of the study is the possible influence of the opinions of the researcher, supervisors and those of the field experts on the outcome of the study. Using data already interpreted means that the data may also have been coloured by the bias of the writers.

External validity was addressed by the description of the context in the introduction, including the relevance of the study for SSAC. Construct validity was addressed by the detailed documentation of the study's research methodology in this chapter. Internal validity and reliability were addressed by using triangulation to increase the precision of the empirical research, reduce bias by individuals and increase the innovation in the analysis.

### 2.6 Summary

A case study approach was used to achieve the research objectives. An extended literature review was conducted to (1) develop the framework for analysing the cases, (2) highlight the specialised and complex nature of the NHIS development landscape, (3) support answers to the research questions, as well as (4) acknowledge the large body of existing knowledge upon which this research was based and developed. The analysis framework comprises of (1) the key characteristics for distinguishing the approaches in Coiera's topology from each other, and (2) the key considerations for building NHIS in SSAC, in comparison to HIC. The framework was used (1) to analyse NHIS development in the two cases studied i.e. South Africa and Rwanda, and (2) to analyse how well each approach in Coiera's topology supports the key considerations for building NHIS in SSAC.

# **CHAPTER 3:** Coiera's Topology of Approaches for Building NHIS

An NHIS is described as

"the nationwide organized and integrated network of resources and processes that contribute to the overall production and communication of health-related information, composed by multiple specific information systems (maintained by various programs, offices and institutions) functionally interacting at different levels of a country. This network is coordinated by the national health authority, with the primary goal to support evidence based decision and action in the health sector" [31].

This chapter describes the state of the art of NHIS and their development in high income countries. It starts with descriptions of the role of eHealth, the nature of NHIS, NHIS in HIC and the challenges with developing NHIS. This is followed by a description of Coiera's topology of approaches for building NHIS. The chapter concludes with the key characteristics differentiating these approaches, which will be used later in the analysis of the approaches in SSAC.

### 3.1 The Role of eHealth

Access to comprehensive clinical and operational health information is more critical than ever, given the challenges faced in delivering high-quality and accessible care under increasing fiscal pressures [32, 33]. Given the drive to reform health systems, one of the most important uses of HIS today is to track trends in health systems performance and certain diseases. Patients, health professionals and policy-makers rely on good quality information to make resource, policy and planning decisions [34-36],[37]. Events such as bioterrorism [33], SARS and Avian flu are driving the need for information from the grass-roots [8]. Driven by the millennium development goals there is a need to share health knowledge globally. The rapid growth of the private sector is also driving the need for information to monitor policy, regulatory and operational guidelines [38]. The general public have also become consumers of health information, in ensuring their well-being through access to more reliable, accredited health information [33, 39]. There is also a growing need for patient-based systems to improve the delivery and continuity of care, of patients who are more often, now, visiting multiple centres of care.

"the cost-effective and secure use of information and communications technologies in support of health and health-related fields, including health-care services, health surveillance, health literature, and health education, knowledge and research" [40].

It is widely recognised for its potential to improve health systems safety, quality and efficiency [40, 41]. It is an all-encompassing term which includes all electronic HIS, including, but not limited to (1) population-based HIS (2) patient-based HIS, and (3) the national HIS. The advancement of technology has made possible some innovative solutions to improve the quality of health services provided [42], especially evidence-based practices [36]. It has also reduced the number and skill level of human resources required in managing health information [43], medical errors and costs in delivering care [44]. Information amassed over time can also provide valuable know-how for process improvements [45], which will in turn improve the quality, availability and use of health information [46, 47]. The better the quality of the information (right, accurate, appropriate, meaningful, timely and fast), the more effectively can HIS support the health system [32, 48].

#### 3.2 The nature of an NHIS

The broad goal of an NHIS is to deliver information to users when and where they need it [33]. It must ease access to the diverse national-relevant health-related information coming from multiple sources, as well as drive the generation of new knowledge of health and the health sector [4, 15, 31, 49]. An NHIS can be viewed as a system of heterogeneous sub-systems, that encompasses different HIS and applications, that span across: public and private healthcare [4], formal and informal organizations, different geographical levels (district to national), different health delivery services (primary, public, personal, research), different health domains (workforce, financing, governance, medical products, vaccines and technologies), different diseases, different health specialties, as well as different sectors in government (e.g. armed forces' health services, environmental services) [31].

An NHIS must satisfy the requirements of the various types of information users across the health sector, including: patients/citizens/consumers, health system managers, policymakers,

payer institutions, and researchers [50, 51]. The functional coverage reached by the NHIS also extends beyond the boundaries and functions of the health sector of a country [31]. An NHIS supports sharing of information across national boundaries to facilitate cross-border learning, and to enable global public health surveillance and medical research to combat global issues like HIV [33]. The health organisational hierarchy spans from global through national, region, district, facility/community, to the patient level. Doctors require detailed patient information, districts require aggregated community-level information, while the national authority requires indicators to prepare budgets and take strategic decisions [52]. Typically, the granularity of the information required decreases as one moves up the hierarchy [52].

An NHIS encompasses a number of electronic HIS for collecting and sharing health information. An NHIS architecture will comprise mechanisms by which various HIS communicate with each other to facilitate information sharing, generation, and use. NHIS are complex systems with many, functionally as well as structurally, interconnected human and non-human components [53]. Interoperability is the ability for heterogeneous, autonomous, systems to function jointly, and to share their resources - information and services - in a reciprocal way [52]. An NHIS is a conceptual system that facilitates a single "source of the truth" through nationwide interoperability [54, 55].

## 3.2.1 Electronic Health Records

An NHIS can connect different silos of data by sharing health records [16, 56]. The key types of health records are described in the following paragraphs.

An electronic medical record (EMR) is an electronic record of an episode of medical care within and owned by a single institution e.g. a general practitioner practice or a single hospital [57]. They ease the access to historical data, over trying to wade through a paper-based patient folder. They are used to manage chronic patient care, to exchange information between the patient's healthcare providers, for research, legal purposes, quality-assurance, and the management of healthcare facilities and services [58].

Continuity of care is greatly enabled by access to distributed patient health records [59]. While EMR are described as provider-centric, electronic health records (EHR) are described as person-centric [57]. An EHR has been described as

"a longitudinal electronic record of patient health information generated by one or more encounters in any care delivery setting. Included in this information are patient demographics, progress notes, problems, medications, vital signs, past medical history, immunizations, laboratory data, and radiology reports" [57, 58, 60, 61].

A personal health record (PHR) is a specialisation of an EHR under the custodianship of a person(s) e.g. a patient or family member [57]. A shared-EHR (SEHR) is an implementation of an EHR, in which a single instance of each EHR is stored centrally and shared nationally. The type and extent of EHR varies from one implementation to another, and unfortunately, is sometimes also referred to inter-changeably as EMR in the literature [59, 61].

## 3.3 NHIS in High Income Countries

By 2005 [62] many HIC had already initiated plans to address the lack of flexibility, scalability and reusability of the complex and heterogeneous HIS that resulted from siloed development programmes [22, 63]. They started to leverage the advances in health architectures and frameworks to enable the delivery of affordable, scalable health services, as well as better personalized care.

The lead responsibility of many NHIS sits with the MOH. To address challenges posed by political and legal systems, advisory bodies involving e.g. health provider and patient representatives have been established as part of the eHealth governance [64], and supporting organisations like Canada Health Infoway and Australia's National eHealth Transition Authority (NEHTA) have been funded by the government to lead the eHealth initiatives at a national level [65, 66].

Many HIC eHealth strategies generally include policies, legislation and governance to support the NHIS. HIC, such as Australia, Canada, Spain, Sweden and Denmark, developed health enterprise architecture frameworks that focus on standards and guidelines for improving the interoperability of their HIS in an incremental and progressive manner [63, 66]. The frameworks are used to create a common understanding of a country's NHIS design [64] and to provide guidance to business and IT experts delivering eHealth systems to the country [65, 67]. Some countries, like the UK, have made substantial investment in developing central

infrastructures. Others, like Canada and Australia, promote the use of a service-oriented architecture (SOA) approach [66, 68, 69], allowing the development of individual IT systems at the local level [70]. Some countries have detailed architecture blueprints e.g. Canada's EHRS Blueprint [69] and Australia's NEHTA Blueprint [65, 66].

In developing their national health architecture frameworks, HIC have leveraged existing architecture frameworks. For example, Canada's EHRS Blueprint has been created by adapting and using the Zachman Framework [69]. Australia's processes and principles are aligned with the Open Group Architectural Framework (TOGAF) [65, 66, 71] and its interoperability framework is based on ISO/IEC RM-ODP [65, 72]. The Health Informatics Service Architecture (HISA) (a European standard), Generic Component Model (GCM) and Reference Model of Open Distributed Processing (RMODP) standards have been implemented in many European countries [73].

To drive eHealth reform across its region, the EU agreed on an eHealth Action Plan with the objective of addressing common challenges through pilot projects and disseminating best practices [3]. This common action by the Member States did not materialise largely due to the differences among the Member States and regions in the Union [3]. The EU has since adopted a less formal, more flexible approach to cater for its federated 'organisation' and complex policy development [10]. The EU countries are now independently evolving to a set of common targets [10]. This centralised action led by the EU has had considerable success [3]. All states have strategies for eHealth, many are working on legal frameworks [15], standards for EHR have been developed, and the EU public health and best practice portals offers reliable and timely information for a variety of users.

# 3.4 The Challenges with Developing NHIS

While the scope and implementation of an NHIS generally differs per country [74], building an NHIS involves connecting the EMR systems [33] through a combination of technology, standards and agreed upon rules and processes [75]. NHIS have, generally, proven difficult to realize on a nation-wide scale, usually as a result of high costs, complexity, uncontrollability, and unintended consequences [15, 33, 53]. Despite funding and resources, implementing EHR remains a stumbling block for many HIC [69, 76, 77] due to the continuous change of system characteristics, the politically driven relationships among healthcare providers [2] and the fact

that local organisations have different starting points, goals and resources [61, 78]. Some, like Coiera [78], question the value of share EHR and summary care records (SCRs) vs distributed but nationally shareable records, in the early phases of the NHIS [2, 4, 78, 79]. They advocate for an NHIS to be driven by country needs and policies, and be justified by feasibility, readiness and cost benefit [31], rather than by pressure from technology producers [80].

# 3.5 Coiera's Topology of Approaches for Building NHIS

NHIS architectures in HIC range from unified centralized systems like the National Health Service (NHS) in the United Kingdom (UK) to decentralized systems like that of the United States (US) [4]. Coiera [4] suggests that the issues to consider when comparing development approaches are: the role and control of government, a strategy for system architecture, common functions, standards compliance, and procurement. Coiera also distinguishes three approaches for building NHIS: top-down, bottom-up and middle-out [4].

In a top down approach, central government builds and controls an NHIS, including its system architecture, standards compliance, procurement process and implementation strategy [4]. The NHIS is made up of standardised IT systems, and centrally stored and shared EHR [22]. It is often implemented through a "rip and replace" strategy, where HIS compliant with national standards replace existing non-compliant systems. The UK initially adopted this approach for building its National Health Service (NHS) [4].

In a bottom-up approach, there is no nation-wide implementation plan for the NHIS. It develops as more local health organizations share more locally stored EMR with each other. There is also no national governance of the resulting NHIS nor its implementation. The development of the NHIS is in complete control of de-centralised local authorities. The USA initially adopted this approach [4]. Systems at regional level connect through regional HIE, but there is no national plan for connecting these systems. Instead of a shared EHR, the system enables the aggregation of information from regional systems.

In a middle-out approach, in collaboration with local organisations, the government drives the development of a national set of goals, policies and standards. Local organisations are encouraged with incentives to meet national goals, and for their health systems to comply with the national policies and standards [78]. To cater for its complex and heterogeneous HIS,

Australia adopted a middle-out approach. They started by developing architecture frameworks that focus on standards [66] and interoperability [63, 65, 69]. They have now progressed on to developing its Person Controlled EHR system [4].

# 3.6 An Analysis of the Differences between the Approaches

In architecting a system, it is important to first understand the principles governing its design and evolution [81]. Conceptualisation of a system's architecture and adherence to architectural precepts assists the understanding of the system's behaviour, composition and evolution [81]. This, in turn, provides insights into its feasibility, utility and maintainability [81]. Being able to constantly and smoothly re-engineer an NHIS is a complex IT, strategic and organizational challenge [54].

The literature describes Coiera's topology, the characteristics of each approach in the topology, the characteristics of the exemplars of each approach [4], lessons learnt from the exemplars [2], the possible reasons the examplars took their respective approaches [22], and what is being achieved by the countries taking such approaches [82]. However, it does not provide a consolidated framework for (1) characterising the high level approach taken by an SSAC in developing its NHIS, (2) determining whether an SSAC is taking an approach from Coiera's topology, and (3) assessing the consequences of an SSAC taking an approach from Coiera's topology. This section aims to formalise such a framework, by a detailed analysis of the existing literature on the characteristics of each approach in Coiera's topology.

Building on the differentiating components Coiera identified, this study identified the following key perspectives from which the approaches in Coiera's topology may be differentiated:

- Technical Architecture: This relates to the technical architecture of the NHIS, with specific focus on the location, access and exchange of EMR.
- Organisation: This relates to the (non-technical) organisational structures that influence and control the NHIS, and the standards and policy frameworks that support its governance. The role of government is of central concern here.
- Implementation Strategy: This relates to the strategies by which the NHIS is implemented, including strategies for setting and enforcing goals, standards and policies.

The differentiating characteristics of the approaches were then abstracted from an analysis of their technical architecture, organisation and implementation strategy.

#### 3.6.1 Technical Architecture

The connectivity of HIS is a key aspect of an approach for building NHIS [83]. Of particular importance is the electronic exchange of health information among organizations, usually facilitated by a technical infrastructure referred to as a health information exchange (HIE). An HIE may be implemented as part of an NHIS to facilitate the exchange of health related data between HIS, including EMR and other HIS dealing with patient-level data, according to recognized standards [84]. It links data across systems using unique patient identifiers, and authorises access to the data using unique provider and facility identifiers [64]. The HIE is likely to consist of many, often geographically-dispersed, networks that are capable of communicating and exchanging information with each other [85].

Privacy and security are key in fostering trust in an NHIS [64]. An HIE's long-term success depends on the level of trust within the networked community and the strategic advantage offered by the network to its users [86]. The choice of HIE must therefore balance privacy and security against the ability share information [86]. Overall, an HIE architecture may be categorised as having one of the following governance/operational models [86, 87]: centralized, decentralized (peer-to-peer, federated) and hybrid.

Information exchange in a top-down approach is facilitated by a centralised system, in which all data is stored in a single central repository [86, 87]. Any patient medical data captured in local sources is collected and stored in the central repository, for sharing across points of care in the network connected by the HIE. All information exchanges are routed through the central repository [86, 87]. To facilitate information exchange, systems within the NHIS are standardised in terms of data and technology, and some processes in which the EHR play a role may also be integrated and standardised nationally.

In a bottom-up approach, through voluntary collaboration, some local authorities share data between their heterogeneous systems, via proprietary interfaces - resulting in a de-centralised HIE with an inter-dependent subset of systems with shared services. Individual organizations have control of the patient medical data [86, 87]. There is no central repository [86, 87]. The

individual systems are linked through record locator services that enable them to exchange information [86, 87].

In a middle-out approach, local authorities may share data between their, respective, heterogeneous systems, via nationally-standardised interfaces, connected through a hybrid HIE. The patient medical data is usually stored and managed at organizational or regional levels. The information-exchange is enabled through a central hub and supporting central repositories [78] that store a minimal set of specific data that is shared across the network [86, 87]. The central repositories include vital registration and unique identifier look-up systems. The concrete technical implementation of the central hub or service bus is sometimes also referred to as an HIE. Systems that comply with the national standards and policies may seamlessly access and share data through the HIE. Some systems will be capable of conforming to the technical standards imposed by an NHIS; those not able to can be connected through "middleware" technologies [85].

## 3.6.2 Organisation

In HIC, NHIS architecture and the approach to its development has largely been influenced by the structure and nature of its health system, especially its policy and economic nature [22]. The culture of the NHIS formed by its people and leadership - including the culture of ICT use, government stewardship and organisational systems - can have a significant impact on its achieving its goals [16, 83, 88]. The approach for building an NHIS influences key decisions such as the level of local autonomy allowed, governance to be enforced, and level of involvement of key stakeholders in the NHIS development process [4, 78].

In a top-down approach, complete accountability, control and leadership for the NHIS, including the local systems of which it comprises and connects, is centralised at national-level, limiting local autonomy and collaboration.

In contrast, in a bottom-up approach there is no accountability for the NHIS at national-level. Local authorities have complete autonomy over their respective local systems. Collaboration is voluntary and generally only occurs at the local level, and is made possible only by the high level of trust between collaborating entities. The limited governance at national level is mainly to protect privacy and equality of access.

A middle-out approach aims to balance local and national interests and in order to improve connectivity across the NHIS [22]. The middle-out approach is not a completely new approach. Rather, it is a 'hybrid' of the top-down and bottom-up approaches, embracing and extending the activities, processes and structures that is good in each.

The term 'middle' in 'middle-out' does not refer to a position in the organisational hierarchy, as the terms 'top' and 'bottom' do. Rather it refers to a negotiated, operational stance or 'middle-ground' between the top and the bottom of the hierarchy, generally facilitated by an organisation delegated with the responsibility for the NHIS. The term 'out' in 'middle-out' indicates that guidance is provided from this negotiated middle-ground to both the top and bottom of the hierarchy. This guidance is the stance resulting from the negotiation of requirements coming from the top and bottom of the hierarchy. The rules governing this negotiation are highlighted below.

In a middle-out approach, overall accountability and leadership for the NHIS is centralised at national-level. [4] advises that government should not be embedded in the IT process, and instead, must outsource this activity to experts in the field, through a single joint well-resourced entity [4]. This responsibility is generally delegated to a single public-private collaborative - a government function, an independent public/private partnership, or a private organization [85]. Governments' role is to (1) define the policy framework to converge public and private, as well as local and central systems, into a functional national system, (2) fund public sector and provide incentives for private sector initiatives to join the NHIS, (3) develop the platform including standards frameworks, ICT infrastructure, health informatics workforce, progress monitoring and evaluation mechanisms, (4) institute legislation to protect the interests of citizens [4, 78]. Local authorities have complete autonomy over their respective local systems. They are, however, accountable to the national authority for sharing, at national-level, critical data generated by their local systems, in accordance with the national goals, policies and standards [2].

## 3.6.3 Implementation Strategy

The capacity to sustainably build and maintain an effective NHIS is another key aspect of an approach for developing NHIS [83].

In a top-down approach, the NHIS is developed typically through national-scale projects with planned fixed timelines and deliverables, driven by national goals, such as public health [4]. With an emphasis on technical implementation and change management [89], and policy, procedural, and technical standardisation [90], a detailed strategy, detailed enterprise architecture specification, and detailed projects plans direct the development of the NHIS. National-scale rollouts replace non-compliant with compliant systems [4]. These rollouts are few and far between to minimise the impact of changes, especially on core business functionality, and to maximise returns on existing investments [4].

In a bottom-up approach the NHIS evolves organically and indirectly through the plans of local authorities to enhance local systems, driven by the goals of the respective local health systems. The NHIS develops in an ad-hoc manner, through voluntary, locally-initiated projects to enable interoperability between specific local systems. While interoperability across the health sector may be a national desire, the national authority has no formal plan or framework for developing its NHIS, and local authorities consider it of secondary importance in comparison to their patient care services.

In a middle-out approach, through facilitated stakeholder collaboration, the national authority develops a set of common technical goals, policies and standards within a strategy framework, which guides the development of the NHIS as a whole. This approach ensures quality control and protects the interests of the nation and patient. The government may be required to drive the development of some central systems, such as its national ICT infrastructure, health registries and the HIE infrastructure. But, the development of the local systems and their connection to the NHIS, are driven by local authorities who have a vested interest in meeting the common national goals, which may be associated with high-level timelines. Interoperability across the health sector is a common goal, enabled by the national authority, achieved incrementally over time by the local authority. The NHIS evolves incrementally through voluntary, locally-initiated projects to make local systems compliant with national standards and driven by the goals of the respective local health systems, but balanced by common national goals. Various parts of the NHIS may develop independently, at different rates, without affecting core business functionality. Government does not dictate (rather encourages) compliance to the common standards and goals [4]. Local organisations will gradually bring existing systems up to national, as their resources permit, through an "embrace and extend"

strategy [2]. Typically, solutions to a focused subset of NHIS requirements are developed by participants drawn from across the sector [91].

# 3.7 Summary

The literature review highlighted the absence of a detailed framework for establishing the benefits and applicability of Coiera's analysis of approaches for building NHIS in SSAC. This chapter formalised such a framework. Coiera's topology of approaches for building NHIS, differ along the following perspectives: technical architecture, organisation, implementation strategy. The table below summarises the key differences between the approaches, along these perspectives.

Table 1 Key Characteristics Distinguishing the Approaches in Coiera's Topology of Approaches for Building NHIS

Perspective	Technical	Organisation	Implementation Strategy
Approach	Architecture		
Top-Down	Centralised HIE, and standardised systems.  System interoperability is inherent in the	Centralised leadership and governance.  Limited, if any,	Detailed strategy, architecture and project plans drawn up by the national authority, upfront.
	architecture of an NHIS.	local autonomy.	Planned national-scale rollouts, few and far between, replace noncompliant with compliant ones.
Bottom-up	De-centralised systems, with no enforced national standards.  System interoperability is achieved as a side-effect of uncoordinated, ad-hoc, local integrations.	De-centralised.  Local authorities have complete autonomy.	No overall, upfront, implementation plan.
Middle-out	Hybrid HIE, with some centralised systems and some not-necessarily-	Centralised leadership.	Common goals, policy and standards, evolve over time, driven by the national

standard, de-centralised systems.

System interoperability is achieved through local collaboration facilitated by the national authority through common goals, policy and standards (particularly data and interface standards).

Centralised governance of common goals, policies and standards.

Local authorities have autonomy over local systems, but must strive towards goals and adherence to national policies, and standards authority, in collaboration with local authorities.

Incremental evolution towards this common, evolving, target state, mainly through local projects, encouraged and supported by the national authority.

Despite the lack of empirical evidence of successful NHIS implementations developed through a middle-out approach, its flexibility often makes the middle-out approach the best initial strategy for a country [4, 22]. It is also possible for a country to evolve from a bottom-up or top-down to a middle-out approach over time [4]. Hence, [2] recommends a middle-out approach for Greece, [3] for all EU countries, and the UK and USA are also now migrating towards a middle-out approach [22, 92]. The US's change in approach was motivated by the realisation that direction and support from the national-level was essential to achieve national connectivity [22]. In the UK, the central control of the NHS migrated, over time, to complex, devolved governance and funding structures consisting of multiple, diverse and partially autonomous organisations with varying resources and IT capabilities. It became far less uniform and amenable to the top-down approach initially taken [22]. The UK, therefore, have now decided to allow NHS organizations some level of autonomy in the development of their local systems [93].

# **CHAPTER 4: NHIS Development in SSAC**

Despite the declaration of Alma Ata in 1978 to protect and promote the health of all people, SSAC, and within countries poor people, tend to have worse health outcomes than higher-income countries, and people, respectively [94, 95]. Furthermore, it is predicted that, globally, the health system in 2020 will have to treat proportionately more patients and illnesses, with less funds and workers [78]. Consolidating the currently fragmented, domain-specific health information systems in SSAC, into a coherent NHIS, can improve the quality of information available to improve patient care, enable policy-making and increase operational efficiencies throughout the health sector [33, 38, 96]. This will, in turn, lead to better health outcomes for the population.

This chapter describes the development landscape of NHIS in SSAC, highlighting in particular the key differences between NHIS development in HIC and SSAC. It concludes with the key challenges with- and considerations for developing NHIS in SSAC.

# 4.1 NHIS in Sub-Saharan African Countries

The promise of a more co-ordinated, well-architected, interoperable NHIS may benefit SSAC [5], where (1) systems are distributed over remote, geographically-dispersed environments and (2) government often do not have sufficient funding for public HIS and (3) donor driven projects implement HIS for the purposes of monitoring and evaluating only their specific programs, e.g. an HIV system used to monitor and evaluate an HIV treatment program at a few clinics. While NHIS planning and implementation in many SSAC is still largely absent, some SSAC have initiated programmes to develop their NHIS, including artefacts such as standards and interoperability frameworks [97]. SSAC must simultaneously try to introduce more patient-centric technologies, and connect existing systems, while still focusing on public health issues [10].

Despite resource limitations, success stories towards the development of NHIS include the deployment of aggregated-data systems such as the District Health Information System (DHIS) [98, 99] and patient-based systems such Open Medical Record System (OpenMRS) [100-103], both of which are implemented widely in several SSAC. The DHIS is a free and

open source application for sharing aggregated data collected from facilities with higher levels in the public health sector. It derives a combination of health statistics on health service delivery in sub-districts, districts, provinces and nationally [104-106]. OpenMRS [101, 102, 107] is a multi-institution, non-profit collaborative with a growing global network focused on creating EMR systems, and enabling local HIS development and self-reliance within resource-constrained environments [108].

However, the quality, timeliness and relevance of health information needs much improvement [21, 34, 48]. High-quality health information is currently not available when and where it is needed due to a number of critical challenges that make its collection, access and application problematic. Even the most basic administrative tasks in the health sector are more complex than their non-health equivalents [109]. Managing the exponential growth of complex health information, and implementing the ever-changing rules from the various sources that govern the information, has become an increasingly resource-intensive activity [33]. Lost information, poor documentation, and lack of access to available information, impede the delivery of high quality health services in SSAC [38].

There is still a predominance of paper-based systems. In the absence of ICT infrastructure such as computers, electricity and mobile phone coverage, paper-based systems are often the most feasible and cost-efficient solution [110]. However, limitations of paper-based systems include illegibility, ambiguity, incomplete data, poor availability and data fragmentation [110, 111]. Hence, a variety of program-specific electronic HIS are being used e.g. for the care of HIV patients [112]. This siloed and ad hoc development of HIS, result in a complex, fragmented NHIS made up of a number of pilot HIS that do not scale, are not interoperable and overlap in function [100, 113] [21].

There is currently insufficient information on the users of health services. Health plans are, therefore, often based on expressed sentiments, estimations and guess work. This situation directly contributes to inefficiency, inequity, and waste in the provision of health care services [21, 34]. As a result, SSAC are often caught unprepared by fairly predictable epidemics, which place an even heavier burden on health systems that could have been prevented if there was adequate planning and monitoring of trends [34]. Health information available to citizens also varies in quality and is often conflicting [33].

Despite the various initiatives and effort invested in strengthening HIS in SSAC, progress is slow and unsustainable [114]. SSAC developing NHIS generally have some leadership and sponsorship for their NHIS, and many are working towards a defined strategy [112]. Many SSACs claim that their HIS are inadequate or non-existent, because they do not have the financial, human, and infrastructure resources, and policies in place to implement and manage an NHIS. These SSACs also agree that they need help with developing strategic plans for their NHIS [47]. Other high-level issues with current approaches for developing NHIS include: inconsistent planning methods, "techno-centric" rather than "health system-centric" focus, insufficient collaboration & cooperation, regulation and security complexities [115], and over-ambitious goals [31].

## 4.2 The Benefits of Developing NHIS in SSAC

Many commentators have raised concerns that ICT has yet to deliver on its promises, and that the rapid adoption of ICT is a risk [78]. Furthermore, not everyone is convinced that ICT development will advantage SSAC. Most eHealth solutions have been developed in high income countries (HIC) contexts to improve diagnosis and reduce costs [41]. While public HIS are a priority for SSAC, current HIC architecture frameworks have a greater focus on developing patient-based HIS. Developing and maintaining the architecture frameworks used by HIC generally depends on a high level of financial, human and infrastructure resources.

[41] suggests that ICT offers no immediate benefit to SSACs struggling with inadequate service provision, as opposed to just high costs. In particular, some question the ethics of advocating for large-scale EHR implementation in SSAC, especially against the proven alternative healthcare interventions, such as safe water and immunizations, that compete for the same resources [116]. The evidence of the benefits, risks, costs and social value of EHR in developing countries and how these change over time is scant [116].

Even a well-built NHIS IT solution cannot guarantee that it will be used effectively [31]. There is limited capacity, institutional culture and political will for using information to support policy- and decision-making [31, 75]. Decision-making continues to be carried out on the basis of tradition, ignoring available information and evidence [31]. In particular in SSAC, the transition from paper-based records to EMR represents a paradigm shift for the health sector

[59]. Many health workers still document information on paper during patient encounters, recapturing them later in electronic format [59].

Despite its complexity and the challenges with its development and use, many other experts believe that the benefits to be gained by implementing an NHIS far outweighs the initial difficulties encountered [59]. [96] motivates the need for an integrated NHIS in SSAC to support management, planning, and policy development. Since patient care, today, involves several different types of healthcare providers, it is increasingly necessary for the various providers to exchange patient records to accurately diagnose and treat complex conditions [58, 117]. There is also a need for the simultaneous access to patient records by multiple users to inform decision making at various levels and areas across the health sector [118].

SSAC, donors and other stakeholders involved in health have agreed that strengthening NHIS is essential to achieve and sustain improvements in the health sector [21, 36, 96].

# 4.3 Adopting an Approach from HIC

SSAC and HIC share some common characteristics. For example, some of the key goals of the health systems in HIC and SSAC are to achieve Universal Health Coverage, facilitate more evidence-based decision making, offer patient-based solutions and reduce non-communicable diseases. SSAC also have complex organisations with multiple, distributed authorities, and their patients also move between care centres. Health system connectivity is important, for continuity of care of patients moving between facilities and for the country-wide collection of data to monitor and control epidemics. HIC have also had to address the existence and interoperability of heterogeneous systems giving due consideration to (1) security and privacy, (2) the inherent complexities of the health domain, such as the high rate of change, non-standard clinical terminology, and (3) the lack of globally accepted standards. Given the common challenges they face, SSAC may benefit from adopting an approach for developing NHIS tested in HIC.

SSAC, however, have further challenges, not relevant to HIC, which influence the approach they take in developing their NHIS. For example, SSAC are struggling to reduce communicable disease and poverty related health issues. They are struggling to meet the Millennium Development Goals, a pre-requisite to achieving Universal Health Coverage. NHIS in SSAC are weak and fragmented, while HIC have advanced infrastructure, advanced data-aggregating

systems and large coverage of patient-based HIS. The main focus of HIC is on facilitating widespread interoperability between their already ubiquitous EMR systems, through the implementation of EHR systems. In contrast, SSACS, while promoting interoperability between its EMR systems, are still struggling with the deployment of EMR systems, especially in rural areas.

# 4.4 The Challenges with Developing NHIS in SSAC

Developing a national information system of the scale, complexity and heterogeneity of an NHIS, poses complex, unique challenges in SSAC. In this section, the key challenges faced by SSAC, over and above the common challenges faced by all countries with developing NHIS, are described.

## **4.4.1** Resource Constraints

Inadequate resources hinder the development of the NHIS in SSAC [36, 47].

Huge investments and national budgets in excess of US\$36 billion have enabled the great advancement of NHIS in HIC [44, 119-121]. In contrast, less than US\$2 billion a year, almost half sponsored by donors, is spent on different eHealth initiatives in SSAC [21]. Even this investment is becoming more difficult to raise in a struggling economy [122, 123]. Furthermore, much of the investment is used in a manner that is uncoordinated duplicative and fragmented [21], mostly due to the lack of HIS policy [47], planning [124], local ownership and leadership [125, 126]. Long-term national government funding and political support [52] for HIS fluctuates, as HIS initiatives compete with political and social goals such as the reduction of poverty and health service delivery [38, 127] [109, 128].

Many health workers in rural areas, do not understand English, are not computer literate or skilled [44], and have limited access to training [35, 129]. Acquiring specialised [34, 125] resources locally is expensive [34, 38] and difficult [35, 47, 48, 75, 129]. As a result, there is a strong dependence on external resources to implement HIS [17, 114, 125].

The limited, unstable resource availability [129] will often inhibit the distribution and development of infrastructure [36], especially if state-owned. The limited infrastructure, even access to power

[129], [96] is mostly deployed in urban areas [130]. The disconnected remote areas, supporting the majority of the population [44], are a particular challenge [131] when developing HIS for country-wide deployment [103], and can account for why many 'pilot projects' [41] are not implemented nationally [17]. While mobile phone access is prolific [132], bandwidth, where available, is expensive, and network speed is low, and therefore, impractical for use in most rural areas [44, 132].

# 4.4.2 Socio-political Environment with Distributed Autonomy

NHIS development is influenced by the politics of diseases and sponsorships, national and local authorities, vendors and users, and the public [128]. The lack of policies and strong leadership often results in a misalignment of their priorities [114]. This manifests in complicated and unclear governance processes [125], too many diverse and sometimes conflicting requirements for the NHIS [75], overlaps and gaps in the NHIS architecture, a general lack of focus and a waste of already limited resources [31, 124].

The complex processes and relationships of the NHIS spans multiple government departments [38]. On recommendation from the WHO and the World Bank [133], many SSAC have implemented a variety of decentralisation mechanisms to transfer responsibility for health service delivery from the MOH to local authorities [134, 135]. Health care delivery is also highly dependent on the private sector which generally shares only a limited amount of its data and not involved in driving NHIS development [75]. Health systems influence, and are also influenced by the departments of agriculture, labour, education, water and sanitation [21, 48].

Many development programmes run in parallel silos, each developing their own HIS [52, 96, 124, 136]. Since half the HIS budget originates from multiple donor organisations [21], the driving force behind many of these programmes are individuals or groups with diverse political interests, the money they bring in, and the particular diseases in which they are interested [128]. When donors withdraw, they typically do not leave in place a sustainability strategy to ensure that the benefits, if any, are sustained and can be further strengthened over time [35, 114, 128].

HIS development is also greatly influenced by the social and professional cultures of healthcare organisations [36, 109, 131]. Employees often resist new systems, with a fear of being replaced

or restricted by them [131]. The adoption of eHealth is also constrained by a concern for privacy, security and confidentiality of personal health data [39, 85].

As a result of the multiple units of autonomy, local implementation often contends with national HIS policy, goals and control, and. Available resources, especially those already burdened with data collection, are stretched beyond their limits as a result of sometimes duplicative and conflicting administrative, economic, legal, vendor or donor pressures [16, 38] [34, 46].

# 4.4.3 Dynamic, Heterogeneous Systems

NHIS in SSAC comprise of HIS that are heterogeneous technically (in terms of application, platform, protocols, language), in relation to funding mechanisms (from government, donor agencies, the World Bank), and with respect to institutional grounding (central ministries, district administration, local health clinics, vertical programs) [128].

Supporting the varied information needs from district to national level [135], and keeping pace with new developments and technologies [52] is a challenge when an NHIS comprises such varying levels of maturity [137]. Infrastructure range from no electricity to mobile networks and the Internet [11]. HIS are, therefore, also characterised by various, evolving levels of maturity, driven also by changing health priorities [53, 109]. Different systems within the health sector are, at the same time, maturing: 1) from paper-based to electronic records, 2) from standalone computers to networked systems, 3) from local to global (mobile and internet-based) HIS architectures [138], 4) from EMR to EHR systems [103] and to supporting 5) new technologies e.g. monitoring centres and wearable devices and 6) new types of data e.g. DNA or protein data [138].

Replacing the heterogeneous HIS that are currently providing value, with homogeneous alternate solutions is not viable in SSAC, as it can result in either marginalising the low resource settings, or retarding the advancement of high-resource settings. Due to ease of use and legal constraints, even the paper-based forms will not be eliminated in the foreseeable future.

## 4.4.4 Inadequate Systems Design

In a recent survey, most SSAC requested help with developing strategic plans for their NHIS [47]. Many do not have a systematic, high-level, national approach to design an NHIS. [21]. The lack of cohesive planning, an eHealth policy [47, 139, 140], thorough costing analysis, research and requirements gathering, limit the opportunity, for example, to develop common architectural solutions [125].

HIS are often implemented on an ad hoc basis, by foreign software developers, to meet the specific needs of disease-specific programs [48]. There is limited consideration for interoperability at a national level [11, 38, 96], and reusability across systems [11, 31, 44]. Too often, ineffective systems are developed as a result of the lack of understanding, and underestimation of the complexity of health processes and the environment in which the HIS are to be deployed [31, 109, 124]. Many electronic HIS are a direct computerisation of the paper-based systems, perpetuating existing problems [52] [48].

Few implementations from one setting are reused at another [78]. 'Pilot' projects are necessary to enable innovation, but scarce resources have been wasted on too many pilot projects that cannot be effectively scaled-up [17]. Too many artificially controlled research projects do not go further than the 'pilot project' phase [141], because their long term success, typically, depends on implementing environments not characteristic of SSAC [17, 128]. The substantial investment in such heterogeneous legacy systems have made health organisations now reluctant to spend more money on proposed solutions [39].

Existing frameworks for developing NHIS, mainly implemented only in HIC, tend to be voluminous, focused on only a subset of the requirements for an NHIS, difficult to apply and require a great amount of configuration for the health sector in SSAC. For example, the Health Metrics Network framework prescribes resources that are not widely available in SSAC [103].

# 4.5 Key Considerations for Developing NHIS in SSAC

For an effective NHIS to be sustainable [142], it must

- "(1) meet its' present needs, without compromising its' ability to meet its future needs,
- (2) be able to identify and manage risks threatening its long-term viability and

(3) not be dependent on outside support to continue its existence" [114].

The following section describes the key considerations for developing effective NHIS in SSAC in a sustainable manner. These considerations or development strategies are focused specifically on mitigating the key challenges just described.

4.5.1 Capacity Maximisation and Building – dealing with unstable, constrained funding SSAC must take ownership of, and lead their NHIS development efforts. Capacity must be continuously built. At the same time, economic and political feasibility, and the capacity of national management to implement the NHIS [31, 36] must be balanced with a realisation of benefits [33, 35, 51, 78]. For example, development could focus first on simplifying current processes, and use the gains in the efficiency for further improvements e.g. (1) reducing the overhead of data capture procedures allows users to capture additional data and (2) instead of a replacing entire systems with seemingly superior systems, retaining existing systems with which users are familiar, and gradually introducing enhancements, may reduce training costs and resistance to changes [36]. SSAC must also encourage locally developed and maintained IT solutions [137, 139]. Fortunately, many international donors are making greater efforts to involve country partners in their projects [21, 52].

## 4.5.2 A systems thinking approach – dealing with system complexity

SSAC must adopt a systems thinking approach to developing its NHIS [143-145]. The relationship between the NHIS and its environment is a key component in the design of an NHIS [146]. The interconnected human and non-human components of an NHIS may interact unpredictably, with change in one components having an impact on others [15]. Ineffective systems can have a disruptive effect on an already overburdened health sector [11]. Solutions must be grounded in the realities of practice – both socially and technically. ICT plans must be accompanied by parallel plans to optimise performance in other areas of the health sector such as the redesign of processes, and reimbursement incentives that influence behaviours, and skills development [33, 96]. The work already done must be embraced and extended by applying the lessons learned from one case to the next [11, 18, 40].

# 4.5.3 Interoperability – dealing with heterogeneity

Some of the factors that influence an NHIS, including the decentralised nature of its health care system and its heterogeneous HIS, are entrenched and not likely to change [16, 85]. Policies, standards and technical goals must serve to improve the interoperability of heterogeneous systems and organisations [2]. They must guide the long-term convergence of the existing and new, public and private, local and central systems into a functional NHIS [2, 33, 47, 78, 147]. Most existing standards have been developed for HIC, largely without influence from SSAC. Standards more relevant to SSAC must be developed and adopted [39]. Due to the high cost involved in the generation of Patient Unique Identifiers (PUI) [44], SSAC can use citizen unique identifiers to distinguish the patient and its records from all others [44, 148].

# 4.5.4 Collaboration and Coordination – dealing with organisational complexity and autonomy

SSAC need a comprehensive and collaborative national strategy for developing an NHIS, responsive to the needs of stakeholders at all levels of the health sector [38, 40, 128, 149]. Implementing NHIS involves a complex set of relationships among individuals and organizations with competing goals and priorities. The politics of negotiating changes among the actors at various levels in government between private, public and non-profit organisations, donors and development partners is key to ensuring the nation-wide use and development of the NHIS [16, 31, 40, 51, 52, 64, 75, 114, 150]. NHIS plans must generally also be aligned with the country's strategies for eHealth, non-health national information systems, as well as global information systems. Country plans to improve an NHIS can help bring together international donors, and health and ICT resources in a shared mission to strengthen the NHIS [38].

The NHIS must engender trust at all levels [59, 82, 151], in order for the introduced processes and systems to establish and embed themselves [82]. With support from the NHIS leadership, protagonists can spearhead and coach their counterparts to improve NHIS implementations across a greater scope of the health sector [20, 47]. Collaboration, coordination and alignment will enable the development of holistic solutions, policies and standards, reduce the duplication of effort, avoid gaps in requirements and disparity in solutions, and make possible access to the skills and capabilities lacking in government [38, 52, 68, 131].

Such coordination requires strong, committed, local and national, health sector and IT, ownership and leadership. Coiera [78] advises government to delegate this responsibility to a senior IT capability that can drive and vet the development of the NHIS, while outsourcing the development of the system to experts in the IT industry. The leadership of the NHIS must strive to take away constraints, and establish a network of stakeholders, empowering them to innovate [152], increase the reuse of software components and address interoperability issues. Its high-level governance mechanisms ensure that projects achieve both local and health-sector-wide objectives [153].

## 4.5.5 Incremental and Iterative Progress – dealing with resource constraints

It is difficult and risky to implement systems requiring rapid adoption of ICT or substantial change to existing operations and workflows [11, 78]. Problems cannot be completely and sufficiently solved in a single cycle. Iterations enable the systematic identification of problems and evaluation of solutions to address them [52, 148]. An iterative and adaptive development approach, in which benefits and challenges are continuously assessed [22, 53, 64], allows for better risk management. An incremental approach to NHIS development [128] allows countries to optimise their use of scarce resources by starting with simple solutions, and increasing in sophistication as resources become available [36, 114]. Initially, the level of detail of the architecture will be very low - giving the designers much flexibility in their designs. Over time the level of detail will increase, as standards are adopted and enforced, and the design evolves to address the changing needs of the increasing numbers of users [85]. The continuous improvement process must embrace and extend existing systems and knowledge [38]. Standards will also evolve through ongoing negotiations between actors [52, 136].

## 4.5.6 Flexible, Agile and Adaptable Systems - dealing with dynamism

The nature of diseases and health sector operations is changing at a high rate. Connecting the silos is the initial focus of development in SSAC. Over time an NHIS must become agile enough to adapt to the changes in its environment. Solutions must be flexible enough to be enhanced and adapted as more resources become available and more mature/complex techniques are adopted. Solutions must be able to scale to national level, in an incremental manner, without fundamental changes to its design [11]. Systems developed for SSAC must cater for the users' limited, but improving, skill level. For example, it may be pragmatic to initially provide simple,

non-feature-rich systems that can be used by people with limited skill and experience with computers, with plans to enhance features as the users become more skilled in the use of those systems [36]. To support the changing health environment, an NHIS must support customization and re-configuration [152]. It may be necessary for the NHIS to support different solutions for the varying environments, yet fostering re-use across those systems where possible. [103]. For example, solutions must be able to cater for electronic and paper-based systems as well as connected and disconnected environments [11]. Systems, their architectures and the approach to their architecture must therefore be flexible, agile and adaptable.

# 4.6 Summary

Building an NHIS poses unique and complex challenges in SSAC. NHIS development is usually associated with high costs, complexity, uncontrollability, and unintended consequences. In addition, power outages and broken equipment, as well as the human factors associated with working in technologically, cognitively, and socially complex environments are common-place in SSAC [78]. Despite the inadequate systems design, and limited availability of resources that challenge the development of systems [11], NHIS need to become stronger, in order to achieved and sustained improvements in the health sector gains. Given the challenges they share, SSAC may benefit from adopting a high level approach for developing NHIS that has shown to be effective in HIC. The approach, however, must support the following key considerations for developing NHIS in SSAC: (1) maximising and building capacity to deal with unstable, constrained funding (2) taking a systems thinking approach to deal with system complexity (3) improving interoperability with the NHIS to deal with heterogeneity (4) facilitating collaboration and coordination to deal with organisational complexity and autonomy (5) making progress incrementally and iteratively to deal with resource constraints, and (6) developing flexible, agile and adaptable systems to deal with dynamism.

# **CHAPTER 5:** Case Study - South Africa

South Africa is a middle-income country, with one of the strongest economies in Africa [154]. Despite this, the South African health system is weaker than other SSAC that spend a smaller percentage of their GDP on health care [83].

# 5.1 The Organisation and Governance of the Health Sector

South Africa has a health profile of communicable and non-communicable diseases, and incidences of injuries and trauma [154]. Over 85% of its 52.98 million people have access to health care [154] mainly through public health facilities – with 400 of the over 4000 facilities being hospitals [155]. However, delivery of high quality health services is constrained, mainly due to the shortage of financial resources (particularly staff, infrastructure and supplies) [156], and to the residence of a large majority of its population in rural areas [157].

While primary healthcare is provided free at public primary care clinics, a major inequality in healthcare funding exists [158]. 60% of the total health spend in South Africa is consumed by the 15-18% of the population who have medical insurance and can access the private medical sector [154, 157]. Furthermore, urban health services are far superior to rural services. It is common place for patients to move between care centres [117], due in some part to doctors in rural areas using urban public or private facilities as referral centres [159]. The costs associated with clinic and hospital visits (specifically transport and opportunity costs) are a serious deterrent to health system utilisation [158].

Public health services in South Africa are coordinated by the National Department of Health (NDOH), funded mainly through the National Treasury [106, 160], and is organized as four main areas: facility-based, outreach, community, and emergency preparedness and response [154, 161]. Government provides resources, develops policies and coordinates health strengthening initiatives [154]. The MOH and the Members of the Executive Council (MEC) of the nine provinces are committed, by the negotiated service delivery agreement (NSDA) with the President, to deliver [162] on its current priorities [163], consistent with the health-related millennium development goals [164]:

- The National Health Insurance (NHI), a financing system for the provision of essential healthcare for all citizens of South Africa and legal long-term residents. It is an initiative towards achieving universal health coverage.
- To address HIV, AIDS and TB in an integrated manner, and to combat Malaria (as they each contribute significantly to the burden of disease faced by South Africans, particularly by the poor and vulnerable groups).
- To strengthen the Primary Health Care system.
- To reduce the high maternal and child mortality rates in the country.

While the MOH sets the strategy and policies for the country, the Provincial Departments of Health are responsible for the delivery of healthcare, including the HIS, processes and procedures, in their province [135, 157, 160].

South Africa is composed of 9 provinces, which are further divided into 52 districts [165]. In 1997 South Africa committed to a district health system based on the Primary Health Care approach. Decentralization is achieved through the delegation of authority to health districts. The responsibility for managing health service delivery is delegated to districts [154], inkeeping with the district-based model recommended by the WHO, which promotes empowering districts to manage health. In general, national and local governance, while crucial, is poor [159, 166].

The NDOH also has over 25 development partners directly involved in the funding, implementation and rollout of mainly HIV and AIDS programmes to districts [154]. The siloed programmes has resulted in fragmentation and low health outcomes [154]. To avoid duplication and overlap in effort and resources, all partners have pledged their commitment to the 2005 Paris Declaration and the Joint Assistance Framework aimed at promoting consistency and alignment in sustainable support [154]. However, a recent evaluation of the Paris Declaration shows that there is still an increasing misalignment between vertical partner programmes with that of the NDOH, in terms of, but not limited to, accountability, roles and responsibilities [154].

## 5.2 The Organisation and Governance of the NHIS

The NDOH is required in terms of the National Health Act, 2003 (Act 61 of 2003) to facilitate and coordinate the establishment, implementation and maintenance of an NHIS, which will be operational at all levels of the health sector; including the private sector [160]. Although there is no specific allocation, funding for the NHIS must come from provincial and municipal budgets [160]. Individual health departments are responsible for improving their effectiveness through eHealth [160]. For example, in the selection of patient-based HIS at facility level, it is understood that the NDOH has taken the approach of evaluating all systems available in South Africa and creating a short-list from which provinces may select a particular system. The results of the evaluation are yet to be announced. The individual facilities have very little if any ability to make autonomous decisions around HIS - it may need to be approved at the provincial level at least [167].

Various other organisations support the NDOH in establishment, implementation and maintenance of the NHIS [157, 168]. The Technical Advisory Committee of the National Health Council (TAC-NHC) functions as the technical implementation forum of the NHC which is responsible for the implementation of strategic health priorities [169]. The National Health Information System (NHISSA) Directorate in the NDOH, consisting of senior managers for HIS and ICT in the 9 provinces [160], is responsible for policy, regulation and coordination of eHealth [157]. The Health Data Advisory and Coordination Committee (HDACC) is tasked with establishing a monitoring and evaluation system for the negotiated service delivery agreement (NSDA) [170]. The State Information Technology Agency (SITA) is mandated to "consolidate and coordinate the State's information technology resources" [160]. Local eHealth research and development by government bodies and universities is often funded by government [157, 168]. In the private sector, in accordance with the Medical Schemes Act, the Council for Medical Schemes (CMS) is responsible for disseminating private health information [171].

#### **5.2.1** Policies

Polices and legislation are not fully implemented or enforced [157, 168], due to problems with the interpretation of legislation. Enforcement is especially problematic in the private sector [168]. [166] highlights the contrasting understanding of policy implementation between the top and bottom of the health hierarchy.

#### 5.2.2 Standards

As with policies, there is a lack of advocacy, adoption and governance of standards [160]. South Africa, with its observer ("O") membership in ISO/TC 215 Health Informatics [157], is unable to vote on the development of international eHealth standards [160]. However, [160] recognises that technical standards are essential for the successful implementation of an eHealth strategy. Several standards have been adopted to promote interoperability[157]. The NDOH Data Dictionary, developed to support the District Health Management Information System (DHMIS) policy, provides a reference point for selected health information standards such as data set specifications (data elements, indicators, and data validation rules) and the organisational unit hierarchies. The role of the strategy outlined in the Health Normative Standards Framework (HNSF) is to enable the desired network effect of an NHIS [57]. The HNSF provides guidance in navigating through the multitude of eHealth standards, and assists in the assessment of the applicability of international eHealth standards to healthcare information systems deployed in South African healthcare institutions [57]. The MOH has also approved the establishment of an eHealth Standards Board for South Africa to support governance and regulation [169].

The Private Healthcare Information Standards Committee (PHISC) also monitors health information standards in the private sector, and facilitates cooperation related to eHealth standards across the country [160, 171].

## 5.3 The Key Information Systems within the NHIS

Some of the key strengths in South Africa's development of HIS include: its capacity to report on notifiable diseases; the use of health information for strategic planning; and the work being undertaken on EMR systems [168]. In terms of the five stages of HIS maturity identified in [113], South Africa is placed at Stage 3, with various regions within the country ranging from Stages 1 to 4.

[57] provides a comprehensive report on patient-based systems; [168] lists documented reviews by a number of HIS stakeholders of (a) various aspects of the South African HIS and (b) policies, strategies and legislations that support the development of the NHIS. The following

sections briefly describe the country's public and private sector systems as they relate to the development of the NHIS.

## **5.3.1 Population-based HIS**

The Department of Health routinely disseminates statistics on notifiable medical conditions, health status and operational aspects of national and provincial health [209]. Health data is usually generated from censuses, civil registration and surveys, as well as individual, service and resource records [168] [171]. Donors and research projects, and other government departments also continually generate valuable health information [183]

All this information, however, is scattered in many different, geographically-dispersed vertical data collection systems, and managed by various different organisations – not always under the control of the NDOH [104]. For example, though often related, the data generated directly from populations is stored in systems stored and managed separately from vital (birth and death) registrations [168]. The numerous disease-specific registers and surveillance systems are also managed by different organisations in various parts of the country [168]: the National Electronic TB Register used by all provinces, the Department of Health's Clinical Trials Register, National Health Laboratory Service's National Cancer Registry, UNISA and the Medical Research Council's Injuries surveillance system. The multiple (often donor-driven) health monitoring and evaluation systems has resulted in a lack of information ownership, variable data quality, often confusing data variations, lags between collection and analysis, and the lack of granular demographic, social status and locality data [154, 157, 168, 172].

On a positive note, South Africa has done some good work in establishing a data flow through the health system hierarchy [173]. It has adopted the District Health Information Software (DHIS), implemented by each of the provincial Departments of Health [135, 157, 160] and now synonymous with the District Health Management Information System (DHMIS), for managing routine data in support of the NHIS [174]. Well-established [157, 173], it provides a large proportion of the information used for decision-making at all levels of the health sector. [104] provides an official regulatory framework for the DHMIS.

The National Electronic TB Register, the National Electronic HIV Register, the Supply Chain Management System and the National Health Laboratory Service are also operating country

wide (albeit with gaps in certain areas) and delivering data to the national level. A National Health Information Repository and Data Warehouse (NHIRD) was developed to integrate data from various specialist information systems [183]. To procure, control and regulate optimal stock levels of supplies national and provincial departments use Logistical Information System (LOGIS); district and local municipalities may use various financial and procurement systems such as excel, Pastel or even SAP depending on the resources available to at the facility [175].

### 5.3.2 Patient-based HIS

While electronic HIS in most hospitals generally support patient-centric administrative processes such as registration of visits and client statistical surveys (many even making use of recognised standards), fewer than half meet typical patient administration requirements and fewer than a third meet typical patient care requirements [57]. In the private and public sector, the financial management and laboratory modules incorporated in EMR systems have increased the rate of acquisition of EMR systems [59]. Large hospital systems include Meditech, Medicom and Clinicom [157]. Several smaller applications exist, such as PAAB. Unfortunately, no system has reached 100% coverage in any one province [61]. Despite the use of electronic records for administration, and the benefits associated with EMR, patient records remain mostly paper-based [57, 58], due to several factors including illiteracy, electricity, broadband coverage [61] and the high cost of EMR systems.

Some vertical national level patient-based HIS do exist, though not exactly EMR systems. For example, the National Health Laboratory Service (NHLS) supports over 80% of the population through a national network of laboratories [176]. It is inherently a patient-based HIS by its management of individual patient specimens. The Western Cape has implemented the patient-based Primary Health Care Information System (PHCIS) for its primary health care facilities and Patient Record and Health Management Information System (PREHMIS) for the City of Cape Town Metropolitan Municipality [177]. MomConnect [178] is an example of a new era of systems that are truly at a national level; even though it is still being rolled out nationally to all sites, because of the fact that it is a solution leveraging mobile technologies and is one of the first applications based on the HNSF, it is actually accessible country-wide [179]. The government-owned patient administration and billing system (PABS) is used in three provinces [169].

Patients moving between public or private hospitals have no means by which their health records can be transferred between these hospitals, due in some part to patient's health records not being stored electronically in most public hospitals, while private hospitals store patient health records in electronic format [180].

## **5.3.3** Health Information Exchange

The NHIS is fragmented, unsustainable, unwieldy and inoperable [57]. A national data dictionary is available [181], but, data is generally confined to the private or public HIS that generated the data and cannot be accessed by any other HIS [180]. In many health care facilities data collection and management are computerised [105, 157], but, the various HIS are often not standardized nor interoperable [57, 157, 173]. Of those systems that do share data, few abide by international health data-exchange standards [57]. An exception is the health information exchange technology on which the MomConnect and National Pregnancy Registry is based, which fully implements the requirements of information exchange laid out in the South African National Department of Health's Normative Standards for Interoperability Framework [182]. The system (including the OpenHIM on which the Rwandan HIE is based) delivers accurate data, securely throughout South Africa [182].

### **5.3.4** Private Sector Systems

In general, HIS in the private sector support mainly administrative functions [160]. There is no nationally connected network of HIS that supports patient care across the private healthcare sector [171]. Some networks of health specialists such as pathologist groups have their own HIS, with varying capabilities, as there is currently no standard to which they must comply [171].

The White Paper on Healthcare Reform: 1997 [183] proposes the integration of private health sector HIS into the NHIS [168]. However, there is limited, if any, interoperability between HIS in the private and public sectors [58, 171], due in part to the absence of a strategy for assessing the quality of information in the private sector [160, 168, 171]. Fortunately, the requirements for medical scheme claims have prompted significant consultation and joint decision-making by the NDOH, the Council for Medical Schemes and relevant stakeholder groups, laying the foundation for further collaboration on HIS issues [160].

# 5.4 Challenges with Developing the NHIS

South Africa has faced many challenges in implementing its eHealth strategy. The NDOH's uncoordinated eHealth investments have yielded local benefits but not the interoperability required between the heterogeneous systems of the NHIS [57, 173]. The increased demand for accurate and relevant routine information has revealed [104, 168] a range of problems [157]. Without high-level strategic and policy support from government challenges include: (1) a misalignment of goals across the health sector, (2) inadequate involvement of authorities at various levels of the sector in the management and use of information, (3) inefficient and effective HIS, and (4) inadequate governance, management and accountability [83], [104, 160].

Another key challenge is the inequity in resources and services between rural and urban districts [157]. South Africa's HIS landscape includes both high and low resource settings [11], with a marked variance in eHealth maturity across the country [160]. While the larger cities have advanced HIS technologies, rural settings have little or no electronic HIS [11]. Compounding this, the lack of resources makes even the paper-based systems ineffective [168].

Many doctors and nurses in state hospitals also have limited or no access to Internet, either because there are insufficient computers or because hospital management views internet access as a potential source of distraction which will reduce productivity [157]. Furthermore, while the South African telecommunications market is the largest in Africa (mobile phone penetration is estimated at 75% with approximately 90% of the country covered by mobile telephony), bandwidth is still very expensive and broadband penetration is low [157].

Inadequate ICT infrastructure development and management makes a shift to the use of web-based systems difficult [104], and, coupled with a lack of health resources in may facilities, also hinders clinical data being captured at the point of care directly by healthcare professionals [57]. The shortage of computer literate health workers and medical informatics skills, especially in rural areas, has also created a huge dependence on foreign experts [104, 157, 168]. [59] assessed the technology challenges with EHR system implementation as less complex than the people and cultural issues, such as privacy and security of patient information, and the reluctance of practitioners to adopt new systems that interfere with their workflow [59].

### 5.5 The Evolution of the NHIS

This section describes South Africa's approach to building its NHIS in terms of Coiera's topology of approaches, by understanding (1) how the NHIS evolved to its current state and (2) the possible future of the NHIS.

#### **5.5.1** The Past

During the apartheid era, superior services were provided for whites than blacks, especially in comparison to those living in rural areas [133]. In fierce opposition to apartheid's racial fragmentation of health services, and deregulation of the health sector, a range of organisations (mainly Churches and NGO) and individuals were organised, in the 1980s, to develop and promote a national primary healthcare (PHC) strategy for South Africa, based on a system where services in a 'health ward' are integrated with the district hospital at its core [158]. The National Health Plan, the few years prior to 1994, sought to (1) make comprehensive, community-based health care accessible to all South Africans, by decentralising the organisation and management of health services and establishing PHC centres as the foundation of the national health system, and (2) eliminate the fragmentation and duplication of services by integrating all health services under a single Ministry of Health [158]. Post 1994 the public HIS were re-developed to support the new PHC approach. The core modules of the National Healthcare Management Information System (NHC/MIS) were originally envisaged to be: patient registration, a basic summary care record, patient billing, appointment scheduling, clinical pharmacy, a patient master Index (PMI) [160].

In 1996 the NHC/MIS was decentralised to provincial level [160]. The DHIS was also developed in 1996. In 1999 the first national minimum data set for PHC was rolled out to all public primary level facilities [104]. The DHIS has since been implemented by most provinces. Instead of a government mandate, its adoption was enabled by the supporting processes that government developed around the DHIS product which made it attractive for provincial authorities [155]. In the absence of a national eHealth strategy with policy related to eHealth procurement from the government, and enabled by the decentralised management of the health system, in an uncoordinated bottom-up fashion, local authorities acquired different EMR systems for of their choice. Because of varying degrees of capabilities and maturities within and across the provinces, several disparate systems - neither compatible nor interoperable with

each other - have been implemented in the provinces [57, 160] Also as a result of the decentralisation, many of the envisaged components of NHC/MIS were not implemented [57].

Unfortunately the under-performance of PHC-based health system increased the reliance of the South African population on the complementary use of public and private (including traditional healers, herbalists, inyangas, faith healers and prophets) services [158]. This further fragmented the NHIS, as patient records, if captured at all, was now also being stored in private systems with no interoperability with public systems.

The government, showing political will, have since moved to more centralised control. It has assigned the responsibility for eHealth policy, regulation and programme coordination to the Policy Co-ordination and Integrated Planning Cluster within the NDOH [157]. In response to the uncontrolled growth of EMR systems, it has placed a moratorium on EMR systems [57] and attempted to link the various, provincial DHIS [157, 173, 184]. It also undertook a national HIS assessment based on the Health Metrics Network framework [168], the results of which suggested: (a) the development and implementation of strategic plans for an NHIS; (b) the enforcement of the legal framework; and (c) the active promotion of standards in health information production [168].

### 5.5.2 The National eHealth Strategy

Strengthening the NHIS, will enable the use of information to improve planning, management, and the quality of care provided across the health system, both public and private [154]. The NDOH recognises that for such eHealth benefits to exceed the costs over time, requires an approach that includes (1) the adoption of appropriate architecture, standards and policies that will ensure long-term interoperability and quality, (2) the identification of opportunities and setting of priorities, and (3) the development of feasible plans [160]. The national eHealth Strategy provides a roadmap for NHIS development [163]. It identifies ten strategic priorities: Strategy and Leadership, Stakeholder Engagement, Standards and Interoperability, Governance and Regulation, Investment, Affordability and Sustainability, Benefits Realisation, Capacity and Workforce, Applications and Tools to support Healthcare Delivery, Monitoring and Evaluation of the eHealth Strategy, and eHealth Foundations (infrastructure, connectivity, registration of patients, facilities and providers, and a basic national EHR) [160]. The new strategy is for the NHIS to evolve incrementally, building on what exists, with a focus on

standards, interoperability and collaboration [160]. The strategy defines common goals and mentions the development of an enterprise architecture, but only elaborates implementation plans for common infrastructure components [160]. The private sector has not been involved in the development of this strategy [157, 168]; it is proposed that the private sector be included as key stakeholders when strong policies (especially around governance, procurement and standards) are in place and practised [160]. South Africa's strategy also includes plans for developing an enterprise architecture. The country acknowledges that the development of the NHIS must be a partnership between the NDOH and Universities, private sector, NGO and civil society [155]. The NDOH will guide the development with the expertise that they have around social development, and the tools that they have been using and developing over the years [155].

The current strategy also gives provincial authorities autonomy over their systems, but with the requirement that they adhere to policies, comply with standards and strive towards common country goals in order to connect to the NHIS. South Africa aims to encourage (rather than mandate) compliance to their goals and frameworks. Incremental evolution towards common goals, mainly through local projects, is being encouraged and supported by the national authority. The common goals, policy and standards, are also intended to evolve over time, driven by the national authority, in collaboration with provincial authorities.

#### 5.5.3 The Future

South Africa's national HIS has many facets that can now be managed in a more integrated manner [174], through the co-ordination and synergy offered by its national eHealth strategy [185]. However, considerable efforts are required to address the country's NHIS requirements [185].

## 5.5.3.1 Population-based Systems

There is a need to improve South Africa's capacity to monitor and rapidly respond to public health emergencies [160]. A new civil registration system is being developed [185]. National indicators and targets have now been agreed for priority areas [185]. The NHI will require a 'National Health Data Dictionary' for the various coding systems that are to be developed [185]. Information generated through vertical disease-specific systems should be made available to facility and district users, perhaps through the DHIS [106]. Similarly, the information required

by the NHI that resides in the NHIRD must also be made available to facility and district users [183]. In addition, health care providers at clinic level should be encouraged to influence the development of the systems that they use [106].

## 5.5.3.2 Patient-based HIS

The envisaged National Health Insurance (NHI) is expected to increase the need for improved continuity of care information [58]. Functionally, it requires the details of individual encounters - what treatment was received, what payment methods were used [185]. Technically, it requires an effective, national, automated, transaction-based, standards-based, patient-based HIS, that supports interoperability across the public and private sectors [160, 170, 185]. Given that many SSAC (including Kenya, Rwanda) have adopted OpenMRS for all their primary care facilities [148], including HIV, TB and Malaria, it may be worthwhile for South Africa to consider its adoption.

## 5.5.3.3 EHR and HIE

More boldly, [57] recommends a "person-centric healthcare", "cloud-based shared national eHealth infrastructure" be established, with an enterprise architecture that "will define how eHealth solutions, across all levels of healthcare, in both the public and private health systems, will interoperate with each other to support person-centric continuity of healthcare". While researchers, such as [61], suggest that the current lack of eHealth capability cannot justify a national initiative of the magnitude of a national EHR implementation, driven by the NHI [59, 160] includes plans for a basic National EHR and information exchange. Integrating patient health records across the health sector will involve great effort and must leverage innovative solutions from across the country [185]; for example, the Western Cape's Primary Healthcare Information System (PHCIS) and Patient Master Index (PMI) module, which maintains a unique reference number across all its primary health care facilities for every individual patient [177]. The National EHR will be interoperable with legacy provincial electronic health repositories, and allow patient tracking wherever patients present themselves [157, 160, 184].

Architecting a SEHR at this stage is very much in its infancy [186]. The Patient Register and the Master Patient Index (MPI), identified as the main repositories of the core patient records to which all patient related systems and applications should be linked, do not currently exist [160]. For this purpose the NHIS/SA adopted the national ID number as the unique identifier

in health [169]. All systems must use the ID number in building databases making it possible for the sharing of data between hospitals in the province as well as between provinces. A Health Population Registry System (HPRS) is being developed to be deployed at all national health insurance pilot sites [179]. Smart cards are expected to cater for mobile/offline access to the EHR from facilities or providers that will not always be connected to the online EHR services [157, 184].

To increase the feasibility of EHR systems, [58] promotes a phased adoption of standards-based EMR across South Africa. In most cases, the current EMR systems in hospitals can be scaled up to interoperate with a shared national infrastructure [57]. The further adoption of standards-based EMR systems by primary healthcare providers, however, continuously increases the feasibility of an EHR system [58]. The country-wide TIER.Net EMR rollout is another example of such an approach [160], using a configurable middle tier to support the migration towards EMR systems [187]. Paper-based and electronic systems are combined in a unified system to produce common nationally required indicators [187]. For example, in the Western Cape, HIV reports are based on a combination of reports from paper-based antiretroviral registers, TIER.Net (the offline tier-2 software) and EKAPA (the tier-3 networked EMR) [187]. As resources and infrastructure improve, facilities can transition to the more technologically sophisticated tiers [187].

Taking a similar evolutionary approach in the private sector, the effective administrative HIS, that already support a large proportion of private sector, could provide patient records in the future [171]. [180] further recommended that the services of the PAAB system in public hospitals be extended to incorporate patient health records, and be integrated with the EMR systems found in private hospitals through the use of a service hub [180].

### 5.5.3.4 Standards

Compliance with nationally adopted standards may reduce HIS implementations costs [160]. At the highest level, the planned eHealth enterprise architecture will need to be aligned with the Government Wide Enterprise Architecture (GWEA), Minimum Interoperability Standards (MIOS) and Government IT Architecture (GITA) frameworks, which guide the development of public and private national information systems in South Africa [57, 115, 188]. The NHIS/SA sub-committee on Architecture and Standards is working on a national guideline for

uniquely identifying patients of public facilities. As a prerequisite to integrating private and public data, the role of private health sector in the NHIS must be clarified, as well as the rules governing the interoperability of the public and private HIS [168]. The actual development of standards will be an ongoing process. Since data interchange standards alone are insufficient for a national-scale information infrastructure [189], the CSIR [57] makes the following recommendations: (1) the localisation of the document content standards , (2) make the HNSF applicable to facility-based EMR systems, (3) the establishment of a data dictionary for eHealth in South Africa, (4) set up a mechanism to conduct interoperability conformance testing of vendor/supplier products and existing HIS against the HNSF [57].

## 5.6 Summary

In this chapter, South Africa was studied using the analysis framework defined in the methodology section. South Africa's NHIS has historically been implemented in an uncoordinated manner, leaving the NHIS fragmented and unsustainable. South Africa first introduced aggregated-data public health surveillance systems (most notably the DHIS), vital registration systems and management systems (PAAB). They then started introducing EMR systems at provincial level, but like other HIS across the country, these systems are heterogeneous in nature, and generally not interoperable.

# **CHAPTER 6:** Case Study – Rwanda

Rwanda is a low-income, small, landlocked East African country with few natural resources and the highest density of people in mainland Africa.

# 6.1 The Organisation and Governance of the Health Sector

Rwanda has a population health profile that includes communicable diseases – notably acute respiratory infections and tuberculosis, a generalized HIV/AIDS epidemic and periodic outbreaks of malaria, cholera, meningitis, measles and bacillary dysentery [190]. 75% of the population lives within 5 km of a health facility. However, 87% of Rwanda's 11.78 million people live in rural areas [94, 191]. Mainly due to their extreme poverty and sheer difficulty in reaching care centres, less than a third of citizens requiring treatment are able to see a healthcare professional [94, 192]. Still developing its health care infrastructure, access to health is further threatened by the shortage of experienced healthcare professionals, poor facility infrastructure and inefficiencies of the healthcare system [193]. Since Primary Health Centres (PHC) only address basic health problems and Secondary Health Care Centres (SHC) only certain diseases [194], patients must generally access care from different health care centres for a full range of care [195].

The Ministry of Health (MOH) is responsible for supervising healthcare delivery - whether public, private or not-for-profit [190, 193]; and through its programs and institutions, guides and oversees technical, and policy and standards decisions throughout the health sector [196]. With the goal of providing universal health care [193], the MOH has the following priorities [196]:

- Achieve millennium development goals 1 (nutrition), 4 (child mortality), 5 (maternal health) and 6 (disease control) by 2015.
- Improve accessibility to health services, quality of service provision, and quantity and quality of human resources for health
- Reinforce institutional strengthening

Rwanda is divided into 4 provinces, which are subdivided into districts (the basic politico-administrative unit), then into sectors, and again into units and finally into villages [190].

Rwanda's health sector is decentralized and operates within a 3-tier pyramid of responsibility [190]:

- The central level, comprising the central departments of the Ministry of Health and national reference hospitals, have overall responsibility for the health system at national level, including health policies, strategic planning, high-level technical supervision, and resource coordination.
- The intermediate level, comprising the administrative district health departments, is responsible for developing and providing administrative, logistical, technical and political supervision to the operational level.
- The operational level, comprising the district hospitals, health centres, dispensaries and health posts, have a limited role in deciding how health finances are allocated, but their allocated funds go directly to them [193].

Community health workers, elected by their community, deliver basic health care and promote lifestyles [197]. They also need to provide monthly health indicator and service delivery reports to district-level using a CHW web-based HIS hosted on the national PBF server [197]. Traditional healers and faith-based organisations are recognised and being incorporated in the health policies and strategies [190, 197]. The health sector is also supported by other government ministries, non-governmental organizations (NGO), professional associations, and regulatory bodies [196].

The government has implemented a "top-down and bottom-up" planning, budgeting and monitoring process [196]. Performance standards for the district are set from the top, delinking the accountability of district officials to their citizens; so, while having some degree of autonomy, districts are weak in comparison to central government [198]. Within the district, however, the community participates in joint planning, execution and monitoring of primary health care [193].

Development partners, including international organisations, are included in local and national planning and implementation [197]. Although all activities are controlled centrally, through the health hierarchy, decentralisation has allowed partners to interact directly with local authorities, eliciting their requirements and spreading ideas and good practice [197]. More than half of the health budget comes from external sources and about a third private health schemes, with the

governments contribution including loans and grants [197]. Instead of the vertical and disease-specific funding mechanisms characteristic of SSAC, pooled funds finance the integration of primary health care services [193], minimising overlap and duplication, and ensuring support for rural areas as well [197]. However, while some donor partners are fully aligned to the sector structures, others still maintain parallel vertical systems for budget execution and accounting [196]. Despite its heavy dependence on aid being a risk to sustained progress, the effective use of resources has had a widespread benefit to Rwanda's health sector [197]. The poorly organised private sector, with a greater presence in urban areas, is inadequately regulated by-or aligned with the public health sector [190, 197, 199, 200].

The MOH's creation of the Rwanda Biomedical Centre (RBC) and the closely linked reorganization of the pharmaceutical supply and management system will fundamentally change the landscape of Rwanda's health sector [198, 201]. The RBC is a central autonomous body, with the responsibility for drug procurement and supply chain management for the public sector [201]. It is responsible for addressing issues of quality, cost and sustainability of health care services, through innovation and development of evidence-based practices [201].

## 6.2 The Organisation and Governance of the NHIS

Rwanda's eHealth industry is growing. Stakeholders, including the government, non-government, and private sector partners are jointly developing and implementing Rwanda's eHealth strategy [202]. High-level management and governance is centralised at national level under the leadership of the Ministry of Health (MOH), with support from the Ministry of Science, Technology, and Scientific Research, and the Rwanda Information Technology Authority (RITA); but local authorities have autonomy over their local systems [203]. The RITA is responsible for facilitating the private sector in the development of ICT in Rwanda [112, 202]. The government facilitates fundraising, and the design and implementation of all major systems [202]. As a result, Rwanda was nominated as the eHealth centre of excellence for the East Africa community, and was the beneficiary of a 5 year grant from the USA government to continue building national eHealth systems for HIV/AIDS [186]. The government and its partners, control the incentives for eHealth innovation and implementation, and have hence been the main driver of health ICT [202].

A single government sponsored entity (the eHealth Steering Committee) has been tasked with standards and policy, regulation and enforcement of eHealth in Rwanda [195, 203]. The eHealth Department, comprising of a skilled team of people, spearheads the planning and implementation of eHealth initiatives [112, 195, 202]. However, collaboration is encouraged on a national scale, at all levels of the NHIS. Discussions and plans for NHIS implementation are developed through collaborations between government, academic institutions and NGO [202] e.g. the Rwanda Health Enterprise Architecture (RHEA) and Rwanda Health Information Exchange projects.

#### 6.1.1 Policies

Rwanda has a progressive and expansive eHealth policy. 2008 introduced a National Policy for Quality Health Care, which focuses on strengthening the supervisory system at facility and community levels [197]. The Health Sector Data Sharing Policy is aimed at enhancing access to quality data [204].

Decision-making power in Rwanda is highly centralized [198]. Health facilities belonging to the government cannot procure any HIS without authorisation from the central level [205]. According to [198],

"there is a tendency at all levels for government to sidestep effective participation by giving citizens and their organizations a minimal role in the policy-making and planning processes. Although some policy and programmatic decisions are appropriately made by the central government structures (e.g., national health priorities, regulation, and technical standards), the system runs the risk of missing feedback on health needs and on uptake of health programs from the district and its citizens."

However, in contradiction, [197] claims that

"supported by development partners, the MOH has developed and applied a regulatory framework at all levels to ensure that rules and regulations are followed and objectives are accomplished. The system in place to operationalise accountability ensures that there is constant feedback from the bottom to the central level and vice versa. To formulate policies that match the needs of the population, Rwandan policymakers apply a participatory approach, involving communities in identifying their own problems. Policies and strategies implemented in Rwanda often start as pilot initiatives, usually under NGO supervision. If they show results, the government scales

them up at national level. For instance, good results from pilots spurred rollout of the Mutuelle and Performance-Based Financing, tailored to the national context. Effective governance, strong accountability mechanisms and zero tolerance of corruption are essential to the successful implementation of strategies and policies. "

### 6.1.2 Standards

A standardized data quality assessment methodology at national and district levels has addressed issues of data quality [204]. Rwanda promotes the adoption of international standards to enable interoperability between systems in its NHIS without significant re-engineering of existing systems [195].

## **6.3** The Key Information Systems within the NHIS

Rwanda has made significant progress in developing its NHIS [203]. In terms of the five stages of HIS maturity identified in [113], the various regions or districts in Rwanda are generally at Stages 1, 2 and 3; with evidence of a move towards stage 4 in the Rwamagana district [203]. The following sections briefly describe the country's public and private sector systems as they relate to the development of the NHIS.

### **6.1.3 Population-based HIS**

Rwanda has made Health information is publicly accessible [193]. This has enabled the prioritization of evidence-based policies and strategies [193] for more effective decision-making across the health sector [193]. Monitoring and evaluation systems are key enablers of this strategy. The information reported on, however, is scattered in many databases [202]. The HSSP III document [196] outlines Rwanda's health sector performance indicators and the sources of data for the monitoring and evaluation of the sector. The various data are entered separately in siloed systems [202]: nurses fill out different, mutually exclusive forms [202], community health workers fill out daily paper-based reports, and community supervisors enter the data from the paper-based reports into the Community Health Information System (SISCom), where it is aggregated at the health centre, district, and national levels [94].

The Rwanda Health Management Information System (R-HMIS), based on DHIS-2, provides access to information consolidated from over 700 different public health facilities [206]. Nearly 70 private clinics and dispensaries have also started to share their information through the R-HMIS [206]. An indicator for Mutuelle has now been built into the system and SISCom is in transition to be included in the R-HMIS platform [201]. SISCom is a text message system that allows maternal health workers to send patient records via text messages and identification numbers of pregnant mothers as well as any incidents during the pregnancy to the district hospitals [201].

Mutuelle is the national public health insurance scheme, subscriptions to which are subsidized by the government in accordance with the Ubudehe categorisation [201]. The Ubudehe classification of citizens into socio-economic categories is a key tool for the implementation of social protection programs and the targeting of beneficiaries [207]. A common national Ubudehe database is used to stratify the contribution of the population to Mutuelle according to their capacity to pay [207]. The Ubudehe categorization database can be used in villages without internet or computers [201].

The web-based District Health System Strengthening Tool (DHSST), managed by the Decentralization & Integration unit of the MOH, consolidates data from a various sources to cater for the evidence-based performance evaluation for health facilities [208] [196]. A national data warehouse and dashboard portal consolidates data from the HMIS, SISCom, Demographic and Health Survey, and other sources, to provide indicator data [196, 209]. It is managed by the HMIS Department, which controls the access to data across all HIS based on a Data Sharing and Confidentiality Policy [196].

TRACnet, managed by the central government, enables clinics to report aggregated HIV data [202]. It uses fast and cost effective interactive voice response technology to gather information through the web and mobile phones [201]. The data is then recorded and reported by CHWs, and later aggregated and analysed on a national level [201]. The ability to send data through SMS is significant as many centres do not have internet connections or electricity; especially as it allows critical tests to be sent to and analysed by the national laboratory [201]. TRACplus adds TB and malaria indicators as well [201].

The MOH, together with NGO, with financial and strategic support from the government, developed a cloud-based medical supplies distribution system and has plans to monitor its blood bank online [202, 210].. The Logistics Management Information System (LMIS), together with Electronic Medical Records System (e-MRS), is used to maintain and control adequate drug stocks [201]. Other primary routine data collection systems include the HRIS (Human Resource Information System), and the Resource Tracking Tool (RTT) which provides data related to financial resources committed to and disbursed to districts by donors and the government [196]. Telemedicine is also currently used to connect three major hospitals in Rwanda [201].

#### 6.1.4 Patient-based HIS

There is no standard format for medical records across hospitals, though [194] Rwanda has endorsed a rollout of OpenMRS, effectively standardising on the OpenMRS patient management system as the application to support the patient care needs of primary health sites [112]. OpenClinic GA, an integrated hospital information management system, that provides support for EHR, and administration has also been implemented in Rwanda [211]. To date EMR systems have mainly been used to manage HIV patients under chronic care with anti-retroviral treatments [202]. Facility-level information management, which, at this stage, mainly includes a basic link to the DHIS to generate report indicators, is being enhanced to service facilities better [94]. RapidSMS is an SMS-based alert system used by Community health workers to track pregnant women, and improve communication with other health and district level facilities [201, 212].

### **6.1.5** Health Information Exchange

Interoperability between the various heterogeneous HIS in Rwanda is limited [202]. The MOH and its partners developed a pilot health information exchange (HIE), through which existing point of care systems interoperate, to share patient demographic and clinical information between different health facilities in the Rwamagana district [182].

#### **6.1.6** Private Sector Systems

There is no policy governing the procurement of HIS in private facilities. These facilities are, however, expected by law to send certain data in specific formats to the ministry [205].

### 6.4 Challenges with Developing the NHIS

Developing an NHIS in such a complex and harsh environment is constrained by the costs and risks involved [213]. The sheer remoteness of facilities, lack of electricity, and significant staff turnover in remote areas are major hurdles [190, 193, 201].

The structural and functional weakness of the HIS [190, 193] is due in large part to (1) too many eHealth interventions being implemented [201] and (2) different software platforms established with partners support [204]. Many government systems work independently of one another, which creates duplication and inefficient resource management [213]. There is thus a need to harmonize the systems in order to avoid duplication and additional costs and to strengthen the capacity to support the roll-out of the eHealth system in order to make it sustainable [196, 201, 204, 214].

Rwanda does not have an affordable national data centre [204] and in comparison to other SSAC, has fewer users of technology [202]. The insufficient technical capacities of health facilities is due the limited access and high costs of energy, ICT and skills, and insufficient international bandwidth [213]. The need for further investments has unfortunately also delayed the establishment of a nationwide fibre-optic network in Rwanda, which will pave way to more web-based applications that will in turn enable fast and effective data exchange in Rwanda [201].

#### 6.5 The Evolution of the NHIS

This section describes Rwanda's approach to building its NHIS in terms of Coiera's topology of approaches, by understanding (1) how the NHIS evolved to its current state and (2) the possible future of the NHIS.

#### **6.1.7** The Past

From the 1960s, health care was provided for free in Rwanda, with public subsidies through infrastructure, equipment, personnel, drugs and other supplies [215]. The economic problems of the 1980s and 1990s led to a dilapidated system, with poor quality of care [215]. So in 1992,

the government introduced community participation to ease the financing and management of health care [215]. Between 1994 and 1999 assistance with emergency humanitarian situations and rehabilitation where channelled mainly through non-governmental organizations from the donor countries [190]. Since 1999, when Rwanda began to stabilise politically and economically, donor assistance is channelled directly through the government [190]. Post 2006 of the number of donors increased [190]. In an effort to control how the donor funds was being spent, the government began, in a top-down manner to channel all eHealth investment through the MOH. The government is considered as one of the most "effective" in sub-Saharan Africa, with Rwanda ranking on "Control of Corruption" behind only Botswana and South Africa [202].

To bring first-line medical care closer to every Rwandan, in 2005 the Ministry of Health initiated an ambitious plan to train three individuals as CHWs in each of the country's 15,000 villages [94]. Though CHWs generally have little formal education, through training and support from facility-based supervisors, Rwanda's voluntary workers provide an effective first line of health care and referral [94]. As Rwanda's CHW program matured, demand for quality facility-based services grew, but many could not afford it [94]. In 2006, Rwanda decentralized its health sector, granting managerial autonomy to facilities [208], and launched a national community-based health insurance (CBHI) scheme [94].

The decentralisation increased the demand for quality data at the facility level [208]. To monitor the performance of community healthcare and reduce the high turnover of CHWs, the MOH developed the community HIS (SISCom), and a community performance-based financing (PBF) systems [94]. However, a great portion of Rwanda's information systems were developed, from the bottom-up, by NGO from donor countries, developing systems for their specific vertical programmes. The health ministry had multiple systems to gather data from the country's health services, but the systems used to aggregate and analyse the information were weak and not interoperable [94]. Each month, to aggregate data, each of the country's 450 health centres entered their information in local databases and sent a flash drive to one of the 40 district hospitals, which consolidated and sent district data to the central level on a flash drive [94]. Multiple systems existed to gather data on the country's 45,000 community health workers, HIV services, human resources, and other special programs—but these systems could not interact[94]. Despite spending considerable time and money in collecting immense amounts

of data, the MOH was unable to effectively use that information for strategic planning or immediate action [94].

The eHealth initiatives to strengthen and coordinate its NHIS in large measure began in 2007 [94]. The Rwandan government took a top-down approach in developing a full Health Management Information System (HMIS) to address the lack of interoperability between various existing systems [202]. In recent years the reporting indicators were simplified and standardized, and the R-HMIS, in 2012, replaced the HMIS [94]. The R-HMIS allows health centres at all levels to enter interact directly with the national database to capture data and use data [94]. They also moved all of the performance-based financing databases and the national disease surveillance and response system to the DHIS-2 platform [94]. The Rwandan Data Warehouse now also provides the ministry with a clear overview of the health care situation [94]. According to the [94], the Rwandan health system "has the information it needs to adapt and quickly react to changes or stressors in its environment".

The eHealth Steering Committee has more recently become responsible for regulation and enforcement of standards and policy, to accelerate and coordinate the development and implementation of NHIS projects and programmes for the country. Rwanda has also involved the key health stakeholders in the development of its eHealth strategies.

### **6.1.1** The National eHealth Strategy

Rwanda has developed a National eHealth Strategic Plan, which aims to

"provide and maintain highly effective, reliable, secure, and innovative information systems to support clinical decision making, patient management, education and research functions of the health sector in Rwanda in a bid to improve healthcare service delivery",

with the common goal of having

"an integrated, interoperable eHealth system that spans and supports the entire continuum of care across all of the many settings and locations where healthcare is provided and accessed" [195].

The national eHealth coordination unit is strengthening the NHIS with the phased implementation of independent projects [112, 182, 202] such as the Rwanda Health Enterprise Architecture (RHEA) project, which developed an enterprise architecture specification that includes the following components [112]:

- The National Registry Service caters for access to the common resources.
- The Client Registry caters for the management of identity and demographic information for persons who receive healthcare services within Rwanda.
- The Facility Registry caters for the management of identity and related information (e.g. location, capabilities, etc.) of the health care facilities. This registry serves 2 purposes. The first is to identify the various significant locations, e.g., health care provider, warehouse, village, and to provide geographic coordinates so that spatial relationships can be managed. The second is to identify regions, and to locate places within regions to support creation of maps and to allow summary reporting based on geographic entities.
- The Provider/Practitioner Registry caters for the management of identity and related information of health care providers (practitioners), including their specialties and associated capabilities and facilities.
- A Data Extraction Layer on the National Data Warehouse makes available information through reporting and ad hoc queries.
- The Shared Health Record caters for the management of clinical data that has been recorded for patients within the health care system.
- The Terminology Server caters for the management of different code systems and value sets (vocabulary) that are needed to support processing by the HIS components.
- The RapidSMS application provides a single point of contact for receiving care and managing information that is provided by community health workers. Communication between the individual community health worker and the RapidSMS system takes place via cell phone.
- A hybrid HIE, with nationally-standardised interfaces, facilitates interoperability between its existing and future heterogeneous HIS, allowing its NHIS to evolve incrementally [203].

The architecture developed by the RHEA project [203] may be seen as the eHealth Strategy Framework. The rules governing the integration with the HIE could effectively become Rwanda's Integration Framework.

#### **6.1.8** The Future

A strong NHIS could reduce the paperwork burdening health workers and allow the government to more effectively evaluate health service performance, to identify health service challenges, and to improve quality of service across the health sector [202]. The government continues to provide strong leadership in developing the NHIS which will [195]:

- support country-wide access to health insurance information
- enable the delivery of medication and other health supplies through the national drug procurement and tracking information system
- share human resource information through the Human Resource Management Information System

Ideally, government would like existing and future systems to be interoperable with the R-HMIS [202]. The HIE will facilitate information sharing between the connected systems [182], especially of continuity of care records that will enable access to a patient's clinical information from different health facilities and levels in the health system [182]. A National HIE will enable interoperability between the systems making up the NHIS by encouraging compliance to its goals and frameworks – but not mandating this compliance, yet. The intention is to develop the NHIS in an incremental manner, giving provincial and district authorities complete autonomy over their HIS.

### 6.6 Summary

In this chapter, Rwanda was studied using the analysis framework defined in the methodology section. Rwanda's NHIS has historically been implemented through isolated projects, leaving the NHIS fragmented and unsustainable. Its NHIS now comprises aggregated-data and patient-based HIS which are heterogeneous in nature, and generally not interoperable. Despite the challenges, Rwanda still has some of the most advanced HIS in Africa.

### CHAPTER 7: RESULTS AND DISCUSSION

The aim of this study was to develop and apply a framework for (1) characterising the high level approach taken by an SSAC in developing its NHIS, (2) determining whether an SSAC is taking an approach from Coiera's topology, and (3) assessing the consequences of an SSAC taking an approach from Coiera's topology.. The framework, based on the analysis of Coiera's topology, comprises three perspectives viz. technical architecture, organisation and implementation strategy. The framework was validated by applying it in case studies of national health information systems development in South Africa and Rwanda. The results of the case studies were compared and analysed, with the aim of identifying their key similarities and differences that could affect the selection of an approach for building NHIS. A formalisation of the evolution of NHIS in these countries emerged from this analysis. It highlighted specifically the influence of the health systems requirements for public health information and patient-based records on the development of its NHIS. By applying the framework, it was possible to then also assess how each approach in Coiera's topology supports the key considerations for building NHIS in SSAC. This chapter is a discussion of this analysis. It concludes by suggesting which approach shows signs of being the most sustainable for building NHIS in SSAC, highlighting the value of the framework developed for analysing the high-level approaches for developing NHIS in SSAC.

### 7.1 A Technical Architecture Perspective

The technical architecture perspective, explained in more detail in section 3.6.1, deals with the location and movement of health-related electronic information across the health sector.

An analysis of the technical architectures and the history of HIS implementation in South Africa and Rwanda revealed the development of two key distinct types of HIS in SSAC: population-based and patient-based HIS. It also revealed the need to exchange information between the various heterogeneous HIS to effectively support health systems in SSAC. The analysis suggests that NHIS in SSAC generally evolve through three different stages of maturity, with each stage (a) addressing a different management requirement, (b) characterised by different types of systems, and (c) often demanding different strategic approaches. A description of these stages in the next section is followed by an evaluation of how each approach in Coiera's

topology supports the inherent characteristics of the technical architecture emerging in NHIS in SSAC.

### 7.1.1.1 Evolution of the NHIS in South Africa and Rwanda

A combination of planning and events have influenced SSAC evolution to their current state, for example: (1) the increased demand to manage data related diseases like HIV, and (2) plans to take advantage of opportunities that could be gained from the very rapid improvement in information and communication technology, and the adoption of health information standards. An analysis of the evolution of the NHIS in South Africa and Rwanda provides valuable insights on a tried and tested transition path from which lessons can be learnt for other SSAC that are developing NHIS.

## Stage 1 - Population-based HIS

Both South Africa and Rwanda initially embarked on population-based HIS aimed at supporting public health management and policy-making. These systems include birth and death registration systems, disease surveillance systems (including aggregated indicators), health management information systems (monitoring for example bed counts in hospitals), and logistics systems.

The case studies reveal that both Rwanda and South Africa first deployed national birth and death registration systems as part of their civil registration systems. The threat of diseases such as TB and HIV on public health in South Africa and Rwanda then led to the implementation of disease surveillance systems. Rwanda implemented TRACnet for HIV reporting, and South Africa implemented its National Electronic TB Register.

Both countries also implemented health management information systems. Rwanda implemented its HMIS and South Africa its provincial district HIS, both based on the DHIS software. The 'official' national public HIS are usually initiated by the government, with implementing partners from the private sector. Systems, such as those based on DHIS and the national data warehouses in South Africa and Rwanda, were implemented to support planning and policy-making across the health sector by reporting on essential country wide health indicators. Patient-based records still largely remain paper-based. To minimise the burden of routine data collection, the national minimum core indicators in Rwanda and South Africa

exclude some determinants of health private sector indicators. Starting with only the essential indicators on national and district data repositories, helped to improve quality of data collection and use. Aggregated data is manually captured at facilities and sent via districts to national levels where they are imported into systems that use them to produce the country wide indicators.

As its medical supply-chains grew in complexity, logistics systems become essential to effectively manage health logistics. The Logistics Management Information System (LMIS) in Rwanda controls drug stocks. Similarly, in SA, various different supply chain management systems such as excel, Pastel or SAP were implemented at district level and LOGIS was implemented for use by the provincial and national levels.

Rwanda and South Africa both show that population-based HIS are typically the first subsystems of NHIS implemented in SSAC. The study characterised this phase as Stage 1 in the evolution of NHIS in SSAC. A typical stage 1 NHIS will include birth and death registration systems, disease surveillance systems, health management information systems, and logistics systems.

#### Stage 2 -Patient-Based HIS

In both South Africa and Rwanda, the next phase includes the proliferation of public, private and donor sponsored patient-based HIS aimed at using information at the point of collection to directly support health care delivery. Patient-based HIS provide access to historic episodes of medical care with the primary purpose of managing patients, with for example HIV, who are managed over a long period of time. Patient-based HIS are also used for sharing patient information between healthcare providers, research, legal and quality-assurance activities, and management of healthcare facilities and services.

Both South Africa and Rwanda have implemented a variety of different patient-based HIS to support local authorities, the private sector, donors and NGO's for their specific programs. Rwanda has implemented systems based on OpenMRS. In South Africa provincial authorities have implemented various different types of patient administration systems including the Western Cape's Primary Health Care Information System and Primary Health Care Management Information System. In both countries the private sector have also implemented a variety of proprietary patient-based HIS mainly for billing and administration, used in

conjunction with paper-based medical records. Due to resource constraints, though, only less than half of the public facilities in South Africa and Rwanda are able to implement these patient-based HIS.

There is also little or no interoperability across facilities between the patient-based HIS, as they each maintain their EMR in their own formats, usually specific to a disease or health service. There is, however, some aggregated data export from the patient-based HIS for import into the aggregated-data systems. Patient-based HIS such as OpenMRS in Rwanda and Meditech in South Africa allow for some aggregated data to be exported. This data is then sent from facilities to district and national levels, where it is imported into the aggregated-data systems described in stage 1. Both South Africa and Rwanda have, for example, national data warehouses that are fed from various information systems. Rwanda's data warehouse forms part of its HMIS.

Rwanda and South Africa both show that patient-based HIS are typically the second subsystems of NHIS implemented in SSAC. This study characterised this phase as Stage 2 in the evolution of NHIS in SSAC.

### Stage 3 – Interoperable Systems

South Africa and Rwanda are currently taking steps to improve the interoperability between the heterogeneous systems in stage 2, driven by the need for (1) more optimal use of resources allocated to the development of sub-systems of the NHIS (2) greater data usage, especially local authorities' use of patient-based records, (3) greater access to patient records, especially across facilities, and (4) improved quality of data. Both countries are currently working on national eHealth strategies, and adopting policies and standards to improve interoperability between the sub-systems of their NHIS. They are also in the process of designing and developing foundational health information exchange architectures to facilitate the exchange of patient-based records across patient-based HIS, with automatic feeds into population-based HIS.

Despite having greater resource constraints, Rwanda has made more progress towards interoperability within its NHIS than South Africa. Rwanda is working with its donor, NGO and private sector partners to improve the interoperability between the existing as well as new systems within its NHIS. The Rwanda Health Information Exchange is being piloted to improve interoperability between maternal and child HIS in a single district, with plans for other domains and districts. In the maternal health sector in South Africa the MomConnect health

information exchange has been implemented nationwide. To improve the interoperability between the some of its HIV systems within their NHIS, South Africa implemented TIER.Net, a system that caters for the export of HIV data from patient-based HIS to support eKapa, the national HIV/AIDS HIS, and iteratively enable further interoperability between these systems.

South Africa and Rwanda have both also developed national eHealth strategies and promoted the adoption of many international standards such as HL7 for their NHIS. South Africa has also defined the Health Normative Standards Framework, which provides guidance in navigating through the multitude of eHealth standards, and assists in assessing the applicability of international eHealth standards to HIS deployed in South Africa. Future system interoperability in both countries is planned to be achieved through local collaboration facilitated by the national authority, in a phased manner.

Rwanda and South Africa both show that HIE architectures typically follow the implementation of patient-based HIS in SSAC. The study characterised this phase as Stage 3 in the evolution of NHIS in SSAC. A typical stage 3 NHIS will include technical architectures to share patient-based records, and supporting policies and standards.

## 7.1.1.2 <u>Dealing with an Evolving and Heterogeneous NHIS</u>

In stage 1 of the evolution of NHIS in SSAC, development of the aggregated-data systems are driven by requirements for less detail by users higher up the organisational hierarchy. Their requirements can typically be achieved with integrated systems such as the HMIS in Rwanda, exhibiting characteristics of the centralised technical architecture described in 3.6.1, or homogeneous HIS such as the provincial DHIS in SA, as the data being captured is usually (in most cases, de-identified) aggregated data, standardised across the country. A top-down approach can generally support the development of these aggregated-data systems.

In stage 2, South Africa's implementation of patient-based HIS and Rwanda's donor system implementations shows local authorities and organisations driving the development of the patient-based systems. Managing the various types of data (e.g. across disease types and health services), with varying levels of granularity, with homogeneous HIS is extremely difficult. Hence the proliferation of various heterogeneous patient-based HIS by local authorities and organisations. A key characteristic of a top-down approach is its centrally architected and

standardised systems. Taking a top-down approach can limit the incidence of heterogeneous systems, conflicting with the manner in which patient-based HIS are usually implemented in typical SSAC. A bottom-up approach, void of any rules governing the implementation of systems and any requirements for systems to be homogeneous, can support the heterogeneous patient-based systems implemented in typical SSAC. Preserving such systems that are already entrenched and adding value in existing processes, automatically gets the buy-in of "owners and supporters" of these systems, who already understand their value proposition.

The model of the evolution of NHIS in South Africa and Rwanda highlights an increase in granularity of the data in focus from stage 1 (mostly population-based aggregated data) to stage 2 (granular patient-based records). Left uncontrolled, the 'unofficial' patient-based HIS can divert resources from coherent plans for the NHIS aimed at making more optimal use of scarce resources for the greater good of the health system as a whole. They also often create confusion with inconsistent data. Storing data in different locations, in different formats and in different datasets adds complexity to the processes providing the single view of the health status of the country sought in stage 1. Also, any interoperability between the patient-based HIS in stage 2 is implemented by donors and local authorities, exhibiting characteristics of the de-centralised technical architecture described in 3.6.1. Allowing this ad hoc implementation of proprietary interfaces, without national standards, can make the system even more complex with each integration project.

Many SSAC, including SA, Rwanda and Ghana, have now prioritised interoperability between components of their NHIS as a key element of future development. By stage 3, in both South Africa and Rwanda, the complexity of systems increases exponentially, as the focus moves to connecting the systems in a manner that caters for more effective use of the various types of available data for the benefit of both the individual patient and public health. Designs for introducing interoperability between the existing systems can get convoluted. Incompatible data models increases the complexity of interoperability and data reconciliation across HIS, making it difficult to align these systems with national priorities and goals. In addition, the geographical and administrative regions within SSAC may evolve at different rates depending on their access to resources. South Africa encountered such a problem with the implementation of its various provincial and donor-driven patient-based HIS. To prevent further fragmentation of its NHIS South Africa has placed a moratorium on the introduction of 'unofficial' new patient-based HIS. This type of moratoria has been implemented in other SSAC as well e.g. Uganda.

A health information exchange (HIE), one approach to interoperability, is an important technical infrastructure of an NHIS that facilitates the electronic exchange of health-related information among the organizations that need access to it. It is implemented to facilitate the exchange of health related data between HIS, including EMR and other systems dealing with patient-level data, according to recognized standards. Each approach in Coiera's topology is associated with a different technical architecture depending on their respective governance and operational models: (a) centralized in the top-down approach, (b) decentralized in the bottom-up approach, and (c) hybrid in a middle-out approach.

Specialised expertise, which is not easily available in SSAC, is required to manage the technical complexities of NHIS in stage 3. The implementation of the Rwanda HIE pilot required the expert assistance of foreign organisations and donors to enable interoperability between the patient-based HIS it connects. Even SA, one of the more technologically advanced countries in SSA, is currently struggling with the development of its NHIS.

Through common goals, standards and policies, the middle-out approach can cater for the existence and interoperability of heterogeneous systems with different transition strategies, addressing the varying ICT capability and maturity across the health sector in SSAC. Architectures such as the Rwandan HIE and South Africa's TIER.Net will facilitate interoperability between the new and existing heterogeneous systems of the NHIS. Preserving existing systems makes optimal use of resources allocated to the development of systems. Standards and policies allow for the re-use of data and systems which also improve resource usage. Being able to automatically share and exchange records reduces the duplication in capturing information allowing the saved resources to be used to capture other information which increases the access to information. Capturing more information of value to the local authorities can improve their usage of information. The automatic sharing and exchange of data can also reduce the errors associated with the capture and storage of the same data in multiple systems by multiple people, greatly improving quality of data.

The analysis of the evolution of NHIS in South Africa and Rwanda highlighted the heterogeneous nature of NHIS in SSAC. It confirmed the importance of improving interoperability between the sub-systems of these NHIS in order to effectively evolve into an

NHIS which serves the country equitably at all levels of the hierarchy through a well-functioning health information exchange.

From a technical perspective, Coiera's top-down approach will not inherently support the heterogeneous systems of NHIS in SSACS, and the bottom-up approach will not facilitate interoperability between these systems. The middle out approach will most likely enable interoperability between these heterogeneous systems.

## 7.2 An Organisational Perspective

The organisational perspective, explained in more detail in section 3.6.2, deals with the non-technical organisational structures that influence and control the NHIS, especially the role of government. Each approach in Coiera's topology is associated with a different organisational structure: (a) in the top-down approach leadership and governance is centralised at national level, with minimal local influence, (b) in the bottom-up approach local authorities have complete autonomy, and (c) in the middle-out approach centralised leadership and governance is coupled with local autonomy.

South Africa and Rwanda show that the health systems in SSAC comprise of a mix of public and private services. Furthermore, South Africa and Rwanda, like many SSAC, have decentralised their health systems. In South Africa provinces are run autonomously under the leadership of the MOH, while in Rwanda, complete control remains under the MOH. However, in Rwanda in particular, the local authorities have also in the past left donors of vertical programmes in control of various health data.

Transformational projects of the scope of an NHIS require holistic planning and design. The NHIS strategy must, therefore, be aligned with the agendas and concerns of all levels of the decentralised health sector. Effective collaboration and coordination will be needed for developing holistic solutions, negotiating interoperability, policies and standards, reducing duplication of effort, avoiding gaps in requirements and disparity in solutions, and enabling access to the skills and capabilities lacking in government. Each approach is analysed below for its ability to deal with the level of organisational complexity and local autonomy characteristic of NHIS development in SSAC.

## 7.2.1 Dealing with Organisational Complexity and Autonomy

The centralised management of the top-down approach attempts to control all aspects of the NHIS from the top of the health system hierarchy, leaving little room for influence from the local authority. Central control is useful when starting the architecting process for political and policy negotiations, but, localised decision-making is equally important to sustain the change process. Even in Rwanda where control is centralised under the MOH, the MOH is heavily influenced by donors and encourages high levels of local autonomy. Developing systems without the co-operation of donors and local authorities can lead to local disaffection. While the standardised systems and processes of a top-down approach seem to imply a reduction in complexity, in the top-down approach users are rarely able to effectively adopt systems implemented for them without their involvement.

While the health systems in many SSAC are decentralised, national public health issues are driving the demand for population-based HIS at national level. Unfortunately, however, their highly political environments make access to data a challenge. The bottom-up approach pays little attention to support relationships between the various systems that support the health system. Any relationship forged is voluntary and ad-hoc. Nation-wide change can be hampered by lengthy joint decision-making involving the autonomous local authorities. This could account for the lack of interoperability between the provincial EMR systems implemented in South Africa.

While the top-down and bottom up approaches fail to acknowledge the impact of influential actors and their relationships, the middle-out approach acknowledges and supports these relationships within the NHIS. In a middle-out approach, the government, with policies and incentives, is better able to create a collaborative environment where a network of communities, including donors and the private sector, behave in a manner that benefits public good and their own agendas. South Africa and Rwanda show that in an effort to curb the proliferation of disparate systems in stage 3, a government sponsored entity can assume control of the sector, and create various approaches to national coordination and interoperability to improve resource utilisation and the quality and use of information. South Africa has delegated the responsibility of developing its NHIS to the Technical Advisory Committee of the National Health Council, and Rwanda to its eHealth Steering Committee. Their national departments of health have been tasked with the responsibility of developing policy and regulation of eHealth. By government driving the development of the common infrastructure, it is also better able to protect privacy,

security and equality of access for all, enabling trust. A decentralized and federated technical architecture, governed by standards and policies, while allowing accurate and timely access to information, greatly reduces the risk of data misuse.

From an organizational perspective, the top-down approach will not generally cater for the requirements of the local authority. And, the bottom-up approach will not cater well for aggregating data across systems for public health reporting at national level. However, in the middle-out approach, central management facilitates coordination and collaboration across the health system in order to (1) cater for, and balance, the requirements of the local and national, and public and private users, and (2) enable private partners and donors to invest resources to develop the NHIS.

## 7.3 An Implementation Strategy Perspective

The implementation strategy perspective, explained in more detail in section 3.6.3, deals with the strategies for setting and enforcing the goals, standards and policies required for developing the NHIS. Each approach in Coiera's topology is associated with a different implementation strategy. In the bottom-up approach there is no overall, nationally driven, high-level strategy. In the top-down approach a detailed strategy, including architecture and project plans for national-scale rollouts, is drawn up by the national authority. In the middle-out approach the national authority develops common goals, policy and standards in collaboration with local authorities to incrementally evolve the NHIS, mainly through local projects encouraged and supported by the national authority.

Each approach is analysed below for its ability to sustain the development of an NHIS despite unstable, constrained resources, dynamism and the level of complexity characteristic of NHIS development in SSAC.

## 7.3.1 Dealing with Unstable Funding and Resource Constraints

Governments in SSAC do not generally have the funds or technical expertise to drive the development of an NHIS without external assistance, hence their high dependence on donor sponsored foreign experts as seen in Rwanda. Maximising and building capacity is critical in developing NHIS in SSAC.

While a top-down approach seems to offer economies of scale, massive top-down projects are associated with greater upfront costs, time and skills for planning, implementation and maintenance in comparison to bottom-up projects. In contrast to the huge budget (almost US\$20 billion) available in the UK for its NHIS implementation, South Africa's entire health sector received US\$9 billion from the 2013 National Budget. A nationwide design and rollout of new systems, apart from being extremely expensive and wasteful, is a complex task. The limited IT infrastructure in SSAC does not support the development of a central technical architecture that can immediately offer national interoperability. The top-down approach with its high initial upfront investment and the promise of a long term, risky return on investment is often not a compelling motivation to leverage funding from an already constrained health budget in SSAC.

The bottom-up approach requires no upfront resource commitments and is, thus, more easily implementable in a resource-constrained environment. Retaining existing systems that satisfy local requirements, instead of procuring new systems that will also require additional staff training can be cost-effective. The NHIS may be developed incrementally, through the implementation of smaller projects as resources become available. The primary focus of the local authority remains on primary health, with a secondary focus on public health. Local systems can evolve independently, without central coordination. However, at national level, this approach leads to unpredictable costs and adhoc use of limited resources, which in the long term may have a negative overall impact in resource-constrained environments. For example, predicting the cost of interoperability between systems that were implemented independently, or how much information can feasibly be shared with other providers is an enormous task. In South Africa, prior to the development of the national eHealth strategy and the moratorium placed by the national authority on EMR acquisitions, limited resources were being spent on various standalone patient-based HIS. In Rwanda, which is heavily dependent on donor funding, predicting and justifying a return on donor investment can become a major hurdle. Furthermore, in an environment where public health issues add a huge burden on the health system, SSAC may be better served giving greater priority to national goals that make better use of existing capacity. South Africa having come to this realisation, placed a moratorium on the acquisition of EMR systems at provincial level.

In a middle-out approach both local and national stakeholders agree on common goals and standards. This allows both national strategic goals and local practical conditions to be balanced. This helps to mitigate over-ambitious expectations by national stakeholders, a common problem with the top-down approach. The middle-out approach may bear similar costs to the bottom-up approach, but the common goals and central co-ordination offer enough predictability and potential optimisation in the use and acquisition of resources. This makes it easier to convince donors on the returns on their investments and enables SSAC, like Rwanda, to take advantage of economies of scale. In a top-down approach there is also a high risk of vendor lock-in. In a middle-out approach, competitive markets are encouraged. Rwanda's NHIS funding initiatives show that a competitive market can make available private sector resources to the national authority. South Africa's NHSF provides a basis for a policy and standards framework, to regulate competition.

### 7.3.2 Dealing with Dynamism

The nature of diseases and health sector operations is changing at a high rate. An NHIS must be agile enough to adapt to the changes in its environment.

The bottom-up approach, free of any central management, is adaptable to local requirements, since new technology and system design decisions are made locally. Local systems only need to take into account local conditions which eases the risk and cost of design and implementation. Moreover, systems can be changed more rapidly to deal with changes in conditions at local sites. This characteristic allows local authorities and donors to implement systems which may not have been possible if their implementation was controlled from national level. System requirements driven from national level would be broader and complex in nature, in order to take into account the different conditions and requirements of multiple facilities across different regions. The focus on locally-driven requirements is probably the driving factor behind the implementation of effective systems like OpenMRS and DHIS in many facilities in SSAC. However, scaling to national level from the bottom-up is difficult, as local systems often do not cater for requirements outside of the environment in which they are deployed.

The large, long-term, top-down projects are able to execute nation-wide implementations. However, the high rate of change in the health system and unavailability of funding in the short term may make initial upfront designs obsolete. By the time the systems get to implementation stage, requirements for those systems, and health systems priorities may have changed and lead to a waste of scarce resources. Even small changes have a ripple effect across the health sector.

Transitioning from existing IT systems to new ones, as well as upgrades to the new systems, is especially challenging. The dependence on costly outsourced resources at the central level to implement changes, makes system updates in a top-down approach more costly than a bottom-up approach and also hampers innovation and agility.

The middle-out approach enables innovative local change from the bottom up, and caters for nation-wide change through its centralised leadership. This makes the middle-out approach less dynamic than the bottom-up, but enables it to support nation-wide change that is not supported by a bottom-up approach. Innovation increases as a result of the variety of ideas surfacing through collaboration within the NHIS. Central leadership can align the various initiatives started at different times in different places for the progression of the NHIS as a whole. Standards and policy frameworks, such as the NHSF in SA, allow for systems to evolve independently to attain national standards, giving priority to local use over national interoperability, without unnecessarily increasing the overall complexity of the NHIS.

## 7.3.3 Dealing with System Complexity

SSAC need help with developing their eHealth strategies. Many do not have a systematic, high-level, national approach to design an NHIS. Complex systems theory and documented field experience suggests that NHIS in SSAC need evolutionary development strategies. An evolutionary approach enables large, complex problems to be solved in more predictable, incremental iterations.

The evolutionary management style of the middle-out approach allows for patterns of problematic behaviour to be observed, and possible solutions to such problems to be tested, iteratively, over time. The feedback loop, which is a key component of an evolutionary development strategy, is lengthy in a top-down approach and may be biased in only supporting feedback from the national authority. No structure exists to facilitate this feedback loop in a bottom-up approach, hence the implementation of various disparate patient-based HIS by various local authorities in South Africa.

In a middle-out approach, policies, goals and standards are not all developed up-front. They are refined over time through collaborations with experts across the health system to meet current requirements, and to accommodate health and technology changes, guiding the incremental

convergence of local systems into an evolving NHIS. Also unlike the top-down approach, the absence of solutions to complex problems, such as the development of systems for the limited infrastructure in SSAC, does not hinder the delivery of achievable solutions that are immediately useful. The NHIS becomes more useful over time and more reactive to change. By starting small, early successes can incrementally build confidence and buy-in from all stakeholders.

From an implementation strategy perspective, the middle-out approach is (1) more predicable than the bottom-up, but less predictable than the top-down, in terms of resource usage for planning, implementation and maintenance and the resulting architecture, and (2) associated with lower and more manageable costs than the top-down approach. The middle-out approach's evolutionary strategy enables (1) national change not catered for by the bottom-up, in a more sustainable manner than the top-down approach, (2) dynamic change at local level, unlike the top-down approach, and (3) better prioritisation of resource usage to cater for the needs of national-level and local authorities.

# 7.4 A middle-out approach for building NHIS in SSAC

The middle-out approach seems better able to address the key challenges with developing NHIS in SSAC. Comparing how well each approach supports the key considerations for building NHIS in South Africa and Rwanda, it was found that the middle-out approach combines the best attributes of the top-down and bottom up approaches. The population-based HIS generally require the predictability of a top-down approach. The productivity and economies of scale of a top-down approach are necessary in situations where timing is crucial to the success of implementing change, and the national health authority needs a greater guarantee that specific instructions will be executed. On the other hand, the health care delivery information systems benefit from the flexibility and support for local autonomy of a bottom-up approach. Furthermore, SSAC with stage 3 NHIS are already characterised by centralised organisation, high-level of local autonomy, some centralised and some non-standard de-centralised systems. They may be able to migrate to a middle-out approach without completely overhauling their NHIS and thus reduce resistance from key stakeholders.

## 7.4.1 Implementing a middle-out approach in SSAC

Many HIC with mature NHIS are adopting a middle-out approach. However, their implementation of a middle-out approach will differ starkly to a SSAC implementation. Some considerations for adopting a middle-out approach in SSAC, especially in managing the risks associated with the unpredictable nature of a middle-out approach, are discussed below.

A clear, well-articulated strategic framework that builds on existing efforts and allows the goals, policies and standards to evolve over time, is an important starting point, as it will accelerate the development of the NHIS. Although a detailed strategic framework is impractical in the early stages, it may help guard against an unnecessary increase in complexity that results from dealing with heterogeneity, autonomy and dynamism. Initial high-level, long-term strategies will develop and evolve and increase in detail over time.

Strategic frameworks like the Health Metrics Network framework, ISO eHealth architecture maturity model and the WHO eHealth strategy toolkit can guide the common goals set for the NHIS as it evolves. They also provide a common collaboration point across to leverage resources available across the world. The system evolves to some level through trial and error through goal setting and feedback, so regular appraisal of the goals, standards and policies is necessary to keep the NHIS current and effective. [15] and [38], amongst others, have developed frameworks for evaluating the progress and effectiveness of an NHIS.

Instead of developing the NHIS from first principles alone, it is proposed that the NHIS be developed additionally through the adoption of specific methods and models from existing frameworks and implemented solutions, where they are found to be suitable in addressing the SSAC requirements. Strong leadership will be required to balance and co-ordinate the requirements from all levels of the health system. Since attainment of common goals must often be incentivised, novel approaches to encourage a convergence towards common goals will be required in the resource-constrained environment of SSAC.

## 7.5 Summary

In this chapter, the framework developed in the study was used to analyse how well Coiera's approaches for building NHIS support the key consideration for developing NHIS in SSAC. Results were drawn from an analysis of the case studies, and substantiated by the literature

review. Even though South Africa and Rwanda are in stage 3 of the evolution of NHIS development in SSAC, and already developing health information exchanges, the analysis is not dependent on the existence of an exchange in these countries. Furthermore, since the analysis was based on the key challenges faced by South Africa and Rwanda, the findings of the study have relevance to other SSAC that share similar challenges.

From a technical perspective, it was found that the top-down approach would not inherently support the heterogeneous systems of SSAC, and the bottom-up approach would not facilitate interoperability between these systems. The middle-out approach seemed most likely to enable interoperability between these systems.

From an organizational perspective, it was found that the top-down approach would not generally cater for the requirements of the local authority, and the bottom-up approach would not cater well for aggregating data across systems for public health reporting at national level. The middle-out approach seemed most likely to (1) cater for, and balance, the requirements of the local and national, and public and private users, and (2) enable private partners and donors to invest resources to develop the NHIS.

From an implementation strategy perspective, the middle-out approach seems more predicable than the bottom-up, but less predictable than the top-down, in terms of resource usage for planning, implementation and maintenance. It also appears to be associated with lower and more manageable costs than the top-down approach. Furthermore, its evolutionary strategy enables (1) national change not catered for by the bottom-up, in a more sustainable manner than the top-down approach, (2) dynamic change at local level, unlike the top-down approach, and (3) better prioritisation of resource usage to cater for the needs of national-level and local authorities.

The results show that both South Africa and Rwanda are migrating towards a middle-out approach. The middle out approach appears to be more effective in dealing with the key challenges faced by SSAC, than the top-down and bottom-up approaches. Fortunately, SSAC may be able to naturally migrate towards a middle-out approach when expanding their NHIS. A middle out approach can allow SSAC to improve the effectiveness and quality of information provided by their NHIS and optimise the use of limited resources available for NHIS

development. However, specific customisation for each SSAC, to account for individual conditions, is required for practical implementation of the middle-out.

The results of this analysis confirms the value of the framework developed for analysing the high-level approaches for developing national health information systems in SSAC.

### **CHAPTER 8: CONCLUSION**

## 8.1 Summary of findings

An effective NHIS can help strengthen a weak and overburdened health system in a SSAC. SSAC are, however, complex environments for building NHIS, and hence, are currently characterized by disparate and fragmented HIS. In a recent analysis of mature NHIS in HIC, Coiera [4] highlighted the importance of the high-level approaches used for building NHIS, and gives compelling arguments for countries to take a middle-out approach. While the overall long terms benefits of taking a middle out approach in high income countries have been analysed and explored [22, 91], it is not apparent that this approach is also applicable for SSAC. However, given the substantial structural differences between NHIS in HIC and SSAC Coeira's approach cannot be directly applied to analyse NHIS in SSAC. Based on an analysis of Coiera's topology of approaches, a framework to perform such an analysis in sub-Saharan Africa, was developed, for (1) characterising the high level approach taken by an SSAC in developing its NHIS, (2) determining whether an SSAC is taking an approach from Coiera's topology, and (3) assessing the consequences of an SSAC taking an approach from Coiera's topology. The framework was validated by applying it inan in-depth analysis of NHIS development in South Africa and Rwanda, supported through interviews with experts in the HIS development in these countries.

Through the application of the framework, it was established that NHIS in SSAC generally evolve through 3 stages. The key considerations or development strategies necessary for building NHIS in SSAC were also elicited from an analysis of the cases and the literature, and were focused specifically on mitigating the key challenges highlighted by the study. Coiera's topology of approaches for building NHIS were then evaluated for their support of these key considerations. The scope of this evaluation spanned not only the development of national patient-based HIS – but also national aggregated-data systems. It was found that the top-down approach works well when applied to projects with a single, clearly-defined national purpose. The inefficient and often fragile bottom-up approach works well when no future-proof planning is required, but dynamic local change is important. However, the middle-out approach showed signs of being the most sustainable approach for building complex systems, like the NHIS, with rapidly changing national and local requirements that do not behave and evolve in a predictable way.

While lessons may be learnt from NHIS development in HIC, their approaches, in their entirety, may not be appropriate for SSAC, due to differences in the socio-political and economic landscapes. The timeline and path for building NHIS in SSAC may not be as direct as the path taken by well-financed HIC. It can be concluded that while a middle-out approach can contribute significantly to building effective NHIS in a sustainable manner, given the nature and evolution of NHIS in SSAC, its application in SSAC will be significantly different to its application in HIC. It is recommended that SSAC review their current approaches for building NHIS in terms of Coiera's topology of approaches and consider a move towards a middle-out approach.

Given the resource constraints in SSAC, it is unlikely that their NHIS will reach the technological sophistication and maturity of those in HIC. For example, in SSAC the minimum infrastructure may not include nationwide real-time access; as and where necessary, real-time access may be prioritised for monitoring public health risks. However, with a competitive market, SSAC have the opportunity to develop innovative and effective NHIS; for example, new, possibly simpler, more streamlined technical paradigms, such as systems supporting mobile technologies, which can then be fed back into the global networks, to benefit other countries, including HIC.

## 8.2 Limitations and Future Work

The problem addressed by the study is an on-going one. The health sector will change over time, and new challenges may appear in SSAC over and above those that were identified in the study. As such the main limitation of the study is that the predicted success of a middle-out approach has not been practically confirmed – neither in HIC nor SSAC. The results of the research were validated only analytically, with only a very limited number of interviews with experts in the field. The results should be validated by practical, field implementations. By observing formal implementations of a middle-out approach in SSAC, the limitations and shortcomings of this study may be highlighted, and further research can address the problem more completely. The time and resources available for this study did not make this feasible. Since the NHIS in SSAC are still far from mature, the case studies should be repeated periodically as the NHIS in SSAC develop. By studying the research questions over time, analysis can be based on a richer data set that more accurately represents the evolution of NHIS

in SSAC. Only multiple, successful national implementations of the middle-out approach in SSAC will eventually validate the results of this study.

The research is biased by the cases selected. For example, South Africa and Rwanda are both in stage 3 of NHIS evolution and both are developing health information exchanges, while most SSAC are still between stages 1 and 2. More cases should be studied in order to further extend the context and increase validity of findings to SSAC in general. The sampling strategy must give consideration to SSAC (1) at different stages in the evolution of NHIS in SSAC (described in 7.1.1.1) (2) in which innovative new NHIS solutions are tested, (3) that face unique challenges, and (4) that could possibly overturn current hypotheses. Also, while South Africa and Rwanda are low and middle income countries (LMIC), the results of the study may also apply to other LMIC. However, the study did not deal with the broader scope of LMIC. The methodology designed for this study may be extended in a further research initiative to increase the scope of its applicability to LMIC in general.

The study did not deal with any detailed analysis of NHIS such as its engineering, procurement, nor concrete technical architectures. For example, an analysis of the challenges with developing NHIS in SSACS, as well as a framework to address them is still to emerge. An evidence-base of the benefits of NHIS implementations must be developed to justify public expenditure. Given their predominance in SSAC, the influence of paper-based health information systems is also acknowledged by the study. While this was not explicitly excluded from the results, the focus of the study was on the development of electronic HIS.

#### REFERENCES

- 1. World Health Organization (WHO), "Developing Health Management Information Systems; A Practical Guide For Developing Countries," 2004; <a href="http://www.wpro.who.int/health\_services/documents/developing\_health\_management\_information\_systems.pdf">http://www.wpro.who.int/health\_services/documents/developing\_health\_management\_information\_systems.pdf</a>, 16/04/2016.
- 2. L.L. Fragidis and P.D. Chatzoglou, "Challenges in implementing nationwide electronic health records: lessons learned and how should be implemented in Greece," *Proc. 10th International Conference on Information Communication Technologies in Health*, 2012, pp. 221-229; <a href="http://infoman.teikav.edu.gr/~fragidis/papers/2012">http://infoman.teikav.edu.gr/~fragidis/papers/2012</a> Challenges in implementing nationwide electron ic health record.pdf, 16/04/2016.
- 3. K.A. Stroetmann, J. Artmann and V. Stroetmann, "Developing national eHealth infrastructures results and lessons from Europe," *AMIA Annual Symposium*, 2011, pp. 1347-1354; <a href="http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3243126/pdf/1347\_amia\_2011\_proc.pdf">http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3243126/pdf/1347\_amia\_2011\_proc.pdf</a>, 16/04/2016.
- 4. E. Coiera, "Building a National Health IT System from the Middle Out," *Journal of the American Medical Informatics Association*, vol. 16, no. 3, 2009, pp. 271-273; DOI 10.1197/jamia.M3183; <a href="http://jamia.oxfordjournals.org/content/jaminfo/16/3/271.full.pdf">http://jamia.oxfordjournals.org/content/jaminfo/16/3/271.full.pdf</a>.
- 5. The World Bank, "The World Bank Data: Sub-Saharan Africa"; http://data.worldbank.org/region/SSA, 16/04/2016.
- 6. G. Habiba, C.D. Evina, M. Vouking and V.C. Tamo, *Are eHealth programs effective in LMIC*?, 2012; <a href="http://www.cdbph.org/documents/Rapid Response Effectiveness of e health programs in LMIC july2012.pdf">http://www.cdbph.org/documents/Rapid Response Effectiveness of e health programs in LMIC july2012.pdf</a>, 16/04/2016.
- 7. M.P. Todaro, *Diverse Structures and Common Characteristics of Developing Nations*, *Chapter 2, Economic Development in the Third World*, 1994, p. 27-66; <a href="http://www.c3l.uni-oldenburg.de/cde/OMDE625/Todaro/Todaro%20Chapter%202.pdf">http://www.c3l.uni-oldenburg.de/cde/OMDE625/Todaro/Todaro%20Chapter%202.pdf</a>, 16/04/2016.
- 8. P. Drury, "The ehealth agenda for developing countries," *World Hospitals and Health Services*, vol. 41, no. 4, 2010, pp. 38-40; <a href="http://www.ncbi.nlm.nih.gov/pubmed/16512063">http://www.ncbi.nlm.nih.gov/pubmed/16512063</a>, 01/01/2013.
- 9. The World Bank, "The World Bank Data: Country and Lending Groups"; http://data.worldbank.org/about/country-and-lending-groups, 16/04/2016.

- 10. ISO/TC 215 Health Informatics, "ISO/TR 14639-1: 2012 : Health informatics -- Capacity-based eHealth architecture roadmap -- Part 1: Overview of national eHealth initiatives," 2012; <a href="http://www.iso.org/iso/catalogue\_detail?csnumber=54902">http://www.iso.org/iso/catalogue\_detail?csnumber=54902</a>, 16/04/2016.
- 11. D. Moodley, A.W. Pillay and C.J. Seebregts, "Position Paper: Researching and Developing Open Architectures for National Health Information Systems in Developing African Countries," *Springer, Foundations of Health Informatics Engineering and Systems, Lecture Notes in Computer Science*, vol. 7151, 2012, pp. 129-139; <a href="http://link.springer.com/chapter/10.1007%2F978-3-642-32355-3\_8">http://link.springer.com/chapter/10.1007%2F978-3-642-32355-3\_8</a>, 16/04/2016.
- 12. P. Runeson and M. Höst, "Guidelines for conducting and reporting case study research in software engineering," *Springerlink, Empirical Software Engineering*, 2008; DOI 10.1007/s10664-008-9102-8; <a href="http://link.springer.com/article/10.1007/s10664-008-9102-8">http://link.springer.com/article/10.1007/s10664-008-9102-8</a>.
- 13. A. Sloman, "Types of Research in Computing Science Software Engineering and Artificial Intelligence," *University Of Birmingham, School of Computer Science*, 2001; <a href="http://www.cs.bham.ac.uk/axs/misc/cs-research.html">http://www.cs.bham.ac.uk/axs/misc/cs-research.html</a>, 16/04/2016.
- 14. A. Bundy, *The Need For Hypotheses In Informatics*, 2011; <a href="http://homepages.inf.ed.ac.uk/bundy/seminars/Hypotheses.pdf">http://homepages.inf.ed.ac.uk/bundy/seminars/Hypotheses.pdf</a>, 16/04/2016.
- 15. H. Hyppönen, J. Viitanen, J. Reponen, P. Doupi and T.L. Vesa Jormanainen, Jukka Vänskä, Ilkka Winblad, Päivi Hämäläinen, "Large-scale eHealth Systems: Providing Information to Support Evidence-based Management," *IARIA*, 2011, pp. 89-95; <a href="https://www.thinkmind.org/index.php?view=article&articleid=etelemed\_2011\_4\_10\_40093">https://www.thinkmind.org/index.php?view=article&articleid=etelemed\_2011\_4\_10\_40093</a>, 16/04/2016.
- 16. D. Castro, "Explaining International IT Application Leadership: Health IT," *The Information Technology and Innovation Foundation*, 2009; <a href="http://www.itif.org/files/2009-leadership-healthit.pdf">http://www.itif.org/files/2009-leadership-healthit.pdf</a>? ga=1.17865969.938830806.1450465675, 16/04/2016.
- 17. H. Lucas, "Information and communications technology for future health systems in developing countries," *Elsevier, Social Science & Medicine*, vol. 66, 2008, pp. 2122-2132; DOI 10.1016/j.socscimed.2008.01.033; http://www.sciencedirect.com/science/article/pii/S0277953608000580, 16/04/2016.
- 18. R. Heeks, D. Mundy and A. Salazar, "Why Health Care Information Systems Succeed or Fail," *SE Information Systems for Public Sector Management*, 1999; <a href="http://www.seed.manchester.ac.uk/medialibrary/IDPM/working\_papers/igov/igov\_wp09.pdf">http://www.seed.manchester.ac.uk/medialibrary/IDPM/working\_papers/igov/igov\_wp09.pdf</a>, 16/04/2016.

- 19. R. Heeks, "Information Systems and Developing Countries: Failure, Success, and Local Improvisations," *The Information Society*, vol. 18, no. 2, 2002, pp. 101–112; <a href="https://www.escholar.manchester.ac.uk/api/datastream?publicationPid=uk-ac-man-scw:1b601&datastreamId=POST-PEER-REVIEW-NON-PUBLISHERS.PDF">https://www.escholar.manchester.ac.uk/api/datastream?publicationPid=uk-ac-man-scw:1b601&datastreamId=POST-PEER-REVIEW-NON-PUBLISHERS.PDF</a>, 16/04/2016.
- 20. J. Braa, E. Monteiro and S. Sahay, "Networks Of Action: Sustainable Health Information Systems Across Developing Countries," *MIS Quarterly*, vol. 28, no. 3, 2004, pp. 337-362; <a href="http://www.idi.ntnu.no/~ericm/misq.pdf">http://www.idi.ntnu.no/~ericm/misq.pdf</a>, 16/04/2016.
- 21. C. AbouZahr and T. Boerma, "Health information systems: the foundations of public health," *Bulletin of the World Health Organization*, vol. 83, no. 8, 2005, pp. 578-583; <a href="http://www.scielosp.org/pdf/bwho/v83n8/v83n8a10.pdf">http://www.scielosp.org/pdf/bwho/v83n8/v83n8a10.pdf</a>, 16/04/2016.
- 22. Z. Morrison, A. Robertson, K. Cresswell, S. Crowe and A. Sheikh, "Understanding Contrasting Approaches to Nationwide Implementations of Electronic Health Record Systems: England, the USA and Australia," *Journal of Healthcare Engineering*, vol. 2, no. 1, 2011, pp. 25-41; DOI 10.1260/2040-2295.2.1.25; <a href="http://www.research.ed.ac.uk/portal/files/11135240/Understanding contrasting approaches to nationwide implementations of electronic health record systems.pdf">http://www.research.ed.ac.uk/portal/files/11135240/Understanding contrasting approaches to nationwide implementations of electronic health record systems.pdf</a>, 16/04/2016.
- 23. K.M. EISENHARDT and M.E. GRAEBNER, "Theory building from cases: opportunities and challenges," *Academy of Management Journal*, vol. 50, no. 1, 2007, pp. 25–32; <a href="https://aom.org/uploadedFiles/Publications/AMJ/Eisenhart.Graebner.2007.pdf">https://aom.org/uploadedFiles/Publications/AMJ/Eisenhart.Graebner.2007.pdf</a>, 2016/08/22.
- 24. K.M. EISENHARDT, "Building Theories from Case Study Research," *Academy of Management Journal*, vol. 14, no. 4, 1989, pp. 532-550; <a href="https://www.tu-chemnitz.de/wirtschaft/bwl5/forschung/forschungsseminar/downloads/15/eisenhardt1989.pdf">https://www.tu-chemnitz.de/wirtschaft/bwl5/forschung/forschungsseminar/downloads/15/eisenhardt1989.pdf</a>, 22/08/2016.
- 25. R.K. Yin, *Case Study Research: Design and Methods*, SAGE Publications, 2003, p. 181; <a href="https://books.google.co.za/books/about/Case\_Study\_Research.html?id=BWea\_9ZGQMwC">https://books.google.co.za/books/about/Case\_Study\_Research.html?id=BWea\_9ZGQMwC</a>, 16/04/2016.
- 26. G. Symon and C. Cassell, *Qualitative methods in organizational research: A practical guide*, SAGE Publications Ltd, 1994.
- 27. A. Armitage and D. Keeble-Allen, "Undertaking a Structured Literature Review or Structuring a Literature Review: Tales from the Field," *The Electronic Journal of Business Research Methods*, vol. 6, no. 2, 2008, pp. 103 114; <a href="http://www.ejbrm.com/">http://www.ejbrm.com/</a>, 2016/08/22.

- 28. C.M. Keet, "COMP700 Resources: Managing your Project," 2011; http://www.meteck.org/teaching/ThesisProjectManagement.pdf, 16/04/2016.
- 29. B.L. Berg, *Qualitative Research Methods for the Social Sciences, Chapter 11, An Introduction to Content Analysis*, The Free Library, 2007, p. 239-267; <a href="https://mthoyibi.files.wordpress.com/2011/05/qualitative-research-methods-for-the-social-sciences">https://mthoyibi.files.wordpress.com/2011/05/qualitative-research-methods-for-the-social-sciences</a> <a href="https://bruce-l-berg-2001.pdf">bruce-l-berg-2001.pdf</a>, 16/04/2016.
- 30. G. Dodig-Crnkovic, "Scientific Methods in Computer Science," 2002; http://www.mrtc.mdh.se/publications/0446.pdf, 22/08/2016.
- 31. C.H. Gattini, "Improving the Structure and Performance of National Health Information Systems: Operational Approach and Strategic Recommendations," *WHO, Technical Series on Information for Decision-Making PWR CHI/09/HA/03*, 2009; <a href="http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php?option=com\_docman&task=doc\_view&gid=104&Itemid="http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/chi./index.php"http://www.paho.org/c
- 32. World Economic Forum, "Global Health Data Charter," 2011; http://www3.weforum.org/docs/WEF HE GlobalHealthData Charter 2011.pdf, 16/04/2016.
- 33. D.E. Detmer, "Building the national health information infrastructure for personal health, health care services, public health, and research," *Medical Informatics and Decision Making*, vol. 3, 2003; <a href="http://www.biomedcentral.com/1472-6947/3/1">http://www.biomedcentral.com/1472-6947/3/1</a>, 16/04/2016.
- 34. M.C. Azubuike and J.E. Ehiri, "Health information systems in developing countries: benefits, problems, and prospects," *Journal of the Royal Society for the Promotion of Health*, no. 180-184, 1999; <a href="http://rsh.sagepub.com/content/119/3/180.long">http://rsh.sagepub.com/content/119/3/180.long</a>, 16/04/2016.
- 35. H.C. Kimaroa, "Strategies for Developing Human Resource Capacity to Support Sustainability of ICT based Health Information Systems: A Case Study from Tanzania," *The Electronic Journal of Information Systems in Developing Countries*, vol. 26, no. 2, 2006; <a href="http://www.ejisdc.org/ojs2/index.php/ejisdc/article/view/272">http://www.ejisdc.org/ojs2/index.php/ejisdc/article/view/272</a>, 16/04/2016.
- 36. M.R. Oak, "A review on barriers to implementing health informatics in developing countries," *Journal of Health Informatics in Developing Countries*, vol. 1, no. 1, 2007, pp. 19-22; <a href="http://www.jhidc.org/index.php/jhidc/article/view/3">http://www.jhidc.org/index.php/jhidc/article/view/3</a>, 16/04/2016.
- 37. C.N. Chaulagai, C.M. Moyo, J. Koot, H.B.M. Moyo, T.C. Sambakunsi, F.M. Khunga and P.D. Naphini, "Design and implementation of a health management information system in Malawi: issues, innovations and results," *Oxford Journals, Medicine & Health, Health Policy and Planning* vol. 20,

- no. 6, 2005, pp. 375-384; DOI 10.1093/heapol/czi044; http://heapol.oxfordjournals.org/content/20/6/375.full.pdf.
- 38. Health Metrics Network (HMN), "Framework and Standards for Country Health Information Systems Second Edition," 2008; <a href="http://apps.who.int/iris/bitstream/10665/43872/1/9789241595940\_eng.pdf">http://apps.who.int/iris/bitstream/10665/43872/1/9789241595940\_eng.pdf</a>, 13/04/2016.
- 39. F. Adebesin, R. Foster, P. Kotze and D.v. Greunen, "A review of interoperability standards in e-Health and imperatives for their adoption in Africa," *South African Computer Journal*, vol. 50, 2013, pp. 55-72; <a href="http://sacj.cs.uct.ac.za/index.php/sacj/article/view/176/73">http://sacj.cs.uct.ac.za/index.php/sacj/article/view/176/73</a>, 16/04/2016.
- 40. World Healh Assembly (WHA), "WHA58.28," 2005; www.who.int/healthacademy/media/WHA58-28-en.pdf, 16/04/2016.
- 41. C.P. Chandrasekhar and J. Ghosh, "Information and communication technologies and health in low income countries: the potential and the constraints," *Bulletin of the World Health Organization*, vol. 79, no. 9, 2001, pp. 850-855; http://www.who.int/bulletin/archives/79(9)850.pdf, 16/04/2016.
- 42. m. Alliance, "mHealth Knowledge"; mhealthalliance.org, 16/04/2016.
- 43. B.H. Trust, "Baobab Health"; baobabhealth.org, 16/04/2016.
- 44. Z. Omary, D. Lupiana, F. Mtenzi and B. Wu, "Analysis of the Challenges Affecting Ehealthcare Adoption in Developing Countries: A Case of Tanzania," *International Journal of Information Studies*, vol. 2, no. 1, 2010, pp. 38-50; http://www.istudies.net/ojs/index.php/journal/article/viewArticle/76, 16/04/2016.
- 45. R.E. Cibulskis and G. Hiawalyer, "Information systems for health sector monitoring in Papua New Guinea," *Policy and Practice*, vol. 80, 2002, pp. 752-758; <a href="http://www.ncbi.nlm.nih.gov/pmc/articles/PMC2567603/pdf/12378295.pdf">http://www.ncbi.nlm.nih.gov/pmc/articles/PMC2567603/pdf/12378295.pdf</a>, 16/04/2016.
- 46. E.H. Shortliffe and M.S. Blois, "The Computer Meets Medicine and Biology:Emergence of a Discipline," *Springer, Medical Informatics : Computer Applications in Health Care and Biomedicine (Health Informatics); 2nd Edition*, 2000; <a href="http://link.springer.com/chapter/10.1007%2F0-387-36278-9\_1">http://link.springer.com/chapter/10.1007%2F0-387-36278-9\_1</a>, 16/04/2016.
- 47. Health Metrics Network (HMN), "Analysis of World Health Organization African Region Health Information Systems based on the Health Metrics Network Assessment Tool," *WHO*, 2012.
- 48. P.E. Mbondji, D. Kebede, E.W. Soumbey-Alley, C. Zielinski, W. Kouvividila and P.-S. Lusamba-Dikassa, "Health information systems in Africa: analysis of resources, indicators, data

- management, dissemination and use," *Journal of the Royal Society of Medicine*, 2014; DOI 10.1177/0141076814528690; <a href="http://jrs.sagepub.com/content/107/1\_suppl/28.full.pdf">http://jrs.sagepub.com/content/107/1\_suppl/28.full.pdf</a>, 16/04/2016.
- 49. Y. Kabak, A. Dogac, İ. Köse, N. Akpinar, M. Gürel, Y. Arslan, . . . E. Sabur, "The Use of HL7 CDA in the National Health Information System (NHIS) of Turkey," *Software Research and Development Consultancy*, 2008; <a href="http://www.srdc.com.tr/share/publications/2008/FinalCDA.Turkey.IHIC08Formatted.pdf">http://www.srdc.com.tr/share/publications/2008/FinalCDA.Turkey.IHIC08Formatted.pdf</a>, 16/04/2016.
- 50. T.B. Üstün, S. Chatterji, A. Mechbal, C.J.L. Murray and WHS Collaborating Groups, "The World Health Surveys," *Health Systems Performance Assessment*, vol. Chapter 58, 2003, pp. 797-808; <a href="http://www.who.int/healthinfo/survey/hspa58.pdf">http://www.who.int/healthinfo/survey/hspa58.pdf</a>, 16/04/2016.
- 51. A. Aaviksoo, "National Health Information System," *Health Policy Monitor*, 2005; <a href="http://hpm.org/en/Surveys/PRAXIS\_-\_Estonia/06/National\_Health\_Information\_System.html">http://hpm.org/en/Surveys/PRAXIS\_-\_Estonia/06/National\_Health\_Information\_System.html</a>, 16/04/2016.
- 52. E.K. Kossi, J.I. Sæbo, O.H. Titlestad, R.R. Tohouri and J. Braa, "Comparing strategies to integrate health information systems following a data warehouse approach in four countries," *Special Issue: ICT and the Millennium Development Goals*, 2010; <a href="http://www.uio.no/studier/emner/matnat/ifi/INF3290/h10/undervisningsmateriale/ComparingStrategiesForHISintegration.pdf">http://www.uio.no/studier/emner/matnat/ifi/INF3290/h10/undervisningsmateriale/ComparingStrategiesForHISintegration.pdf</a>, 16/04/2016.
- T.B.J. Margunn Aanestad, "Building nation-wide information infrastructures in healthcare through modular implementation strategies," *Elsevier, Journal of Strategic Information Systems*, vol. 20, 2011; DOI 10.1016/j.jsis.2011.03.006; http://www.sciencedirect.com/science/article/pii/S0963868711000229, 16/04/2016.
- D. Chen, G. Doumeingts and F. Vernadat, "Architectures for enterprise integration and interoperability: Past, present and future," *Elsevier, Computers in Industry*, vol. 59, 2008; DOI 10.1016/j.compind.2007.12.016; http://www.sciencedirect.com/science/article/pii/S0166361508000365, 16/04/2016.
- 55. International Standards Organization (ISO), "ISO 14258:1998 Industrial automation systems Concepts and rules for enterprise models," 1998; <a href="http://www.iso.org/iso/iso\_catalogue/catalogue\_tc/catalogue\_detail.htm?csnumber=24020">http://www.iso.org/iso/iso\_catalogue/catalogue\_tc/catalogue\_detail.htm?csnumber=24020</a>, 16/04/2016.
- 56. E. Coiera, "Do we need a national electronic summary care record?," *Medical Journal of Australia*, vol. 194, no. 2, 2011, pp. 90-92; <a href="http://www.mja.com.au/public/issues/194\_02\_170111/coi10895\_fm.html">http://www.mja.com.au/public/issues/194\_02\_170111/coi10895\_fm.html</a>, 16/04/2016.

- 57. CSIR, "National Health Normative Standards Framework for Interoperability in eHealth in South Africa; Version 2.0," *Republic of South Africa Department of Health*, 2014; http://www.samed.org.za/Filemanager/userfiles/hnsf-complete-version.pdf, 16/04/2016.
- 58. N. Mostert-Phipps, D. Pottas and M. Korpela, "Improving continuity of care through the use of electronic records: a South African perspective," *South African Family Practice*, vol. 54, no. 4, 2012, pp. 326-331; <a href="http://www.safpj.co.za/index.php/safpj/article/view/1840">http://www.safpj.co.za/index.php/safpj/article/view/1840</a>, 16/04/2016.
- 59. R.V. Weeks, "Electronic Health Records: Managing the Transformation from a Paper-based to an Electronic System," *Journal of Contemporary Management*, vol. 10, 2013, pp. 135-155; <a href="http://repository.up.ac.za/bitstream/handle/2263/31892/Weeks\_Electronic%282013%29.pdf?sequence=1&isAllowed=y, 16/04/2016">http://repository.up.ac.za/bitstream/handle/2263/31892/Weeks\_Electronic%282013%29.pdf?sequence=1&isAllowed=y, 16/04/2016</a>.
- 60. Y.S. Kwak, "International Standards for Building Electronic Health Record (EHR)," *Proc. 7th International Workshop on Enterprise networking and Computing in Healthcare Industry HEALTHCOM*, IEEE Xplore, 2005, pp. 18-23; <a href="http://ieeexplore.ieee.org/xpl/login.jsp?tp=&arnumber=1500373&url=http%3A%2F%2Fieeexplore.ieee.org%2Fxpls%2Fabs\_all.jsp%3Farnumber%3D1500373">http://ieeexplore.ieee.org/xpl/login.jsp?tp=&arnumber=1500373&url=http%3A%2F%2Fieeexplore.ieee.org%2Fxpls%2Fabs\_all.jsp%3Farnumber%3D1500373</a>, 16/04/2016.
- 61. A.-m. Kleynhans, "Is South Africa ready for a national Electronic Health Record (EHR)?," *UNISA*, 2011; <a href="http://uir.unisa.ac.za/bitstream/handle/10500/6128/2011%20MBL3%20Research%20Report%20A-M%20Kleynhans.pdf?sequence=1">http://uir.unisa.ac.za/bitstream/handle/10500/6128/2011%20MBL3%20Research%20Report%20A-M%20Kleynhans.pdf?sequence=1</a>, 16/04/2016.
- 62. R. Rozenblum, Y. Jang, E. Zimlichman, C. Salzberg, M. Tamblyn, D. Buckeridge, . . . R. Tamblyn, "A qualitative study of Canada's experience with the implementation of electronic health information technology," *CMAJ*, vol. 183, no. 5, 2011, pp. E281-E288; <a href="http://www.cmaj.ca/content/183/5/E281.long">http://www.cmaj.ca/content/183/5/E281.long</a>, 16/04/2016.
- 63. J. Calvillo, I. Román, S. Rivas and L.M. Roa, "Easing the development of healthcare architectures following RM-ODP principles and healthcare standards," *Elsevier, Computer Standards & Interfaces*, vol. 35, 2011, pp. 329–337; http://www.sciencedirect.com/science/article/pii/S0920548911001346, 16/04/2016.
- 64. H. Deutsch and F. Turisco, "Accomplishing EHR/HIE (eHealth): Lessons From Europe," *Computer Sciences Corporation*, 2010, pp. 1-13; <a href="http://www.federalnewsradio.com/wp-content/uploads/pdfs/CSC\_Accomplishing\_EHR\_HIE\_eHealth\_Lessons\_from\_Europe.pdf">http://www.federalnewsradio.com/wp-content/uploads/pdfs/CSC\_Accomplishing\_EHR\_HIE\_eHealth\_Lessons\_from\_Europe.pdf</a>, 16/04/2016.

- 65. NEHTA, "NEHTA Interoperability Framework Version 2.0," 2007; <a href="https://www.nehta.gov.au/implementation-resources/ehealth-foundations/interoperability-framework">https://www.nehta.gov.au/implementation-resources/ehealth-foundations/interoperability-framework</a>, 16/04/2016.
- 66. NEHTA, "National E-Health Standards Catalogue; Supporting Standards Implementation Version 4.0," 2007; <a href="https://www.nehta.gov.au/implementation-resources">https://www.nehta.gov.au/implementation-resources</a>, 01/01/2013.
- 67. The (US) Chief Information Officers Council, "Federal Enterprise Architecture Framework Version 1.1," 1999; <a href="http://www.enterprise-architecture.info/Images/Documents/Federal%20EA%20Framework.pdf">http://www.enterprise-architecture.info/Images/Documents/Federal%20EA%20Framework.pdf</a>, 16/04/2016.
- 68. P.A.H. Williams and V. McCauley, "Sound Foundations: Leveraging International Standards for Australia's National Ehealth System," *European Journal For Biomedical Informatics*, vol. 8, no. 4, 2012, pp. en50-en55; http://www.ejbi.org/en/ejbi/artinfo/159-en-15.html, 16/04/2016.
- 69. Infoway, "EHRS Blueprint An interoperable EHR framework Version 2," 2006; <a href="https://www.infoway-inforoute.ca/en/component/edocman/resources/technical-documents/391-ehrs-blueprint-v2-full">https://www.infoway-inforoute.ca/en/component/edocman/resources/technical-documents/391-ehrs-blueprint-v2-full</a>, 16/04/2016.
- 70. National Information Structure Strategic Planning, "National IT strategies Denmark, England and Canada," *The National Board of Health and Welfare, Swedish Ministry of Health and Social Affairs*, 2009, pp. 1-33; <a href="https://www.socialstyrelsen.se/Lists/Artikelkatalog/Attachments/8374/2009-126-152">https://www.socialstyrelsen.se/Lists/Artikelkatalog/Attachments/8374/2009-126-152</a> 2009126152.pdf, 16/04/2016.
- 71. The Open Group, "TOGAF version 9.1," *The Open Group*, 2009; https://www.opengroup.org/architecture/togaf91/downloads.htm, 16/04/2016.
- 72. ISO/IEC, "Information technology Open Distributed Processing Reference model: Overview," 1998; <a href="https://www.itu.int/rec/T-REC-X.901-199708-I/en">https://www.itu.int/rec/T-REC-X.901-199708-I/en</a>, 16/04/2016.
- 73. B. Blobel, "Application of the component paradigm for analysis and design of advanced health system architectures," *Elsevier, International Journal of Medical Informatics*, vol. 60, 2000, pp. 281–301; <a href="http://www.sciencedirect.com/science/article/pii/S1386505600001040">http://www.sciencedirect.com/science/article/pii/S1386505600001040</a>, 16/04/2016.
- 74. M.J. Garfield and R.T. Watson, "Differences in national information infrastructures: the reflection of national cultures," *Elsevier, Journal of Strategic Information Systems*, vol. 6, 1998; <a href="http://www.sciencedirect.com/science/article/pii/S0963868798000122">http://www.sciencedirect.com/science/article/pii/S0963868798000122</a>, 16/04/2016.
- 75. Belize Minster Of Health, "Belize: National Health Information System: Strategic Plan 2010-2014," 2009;

 $\underline{http://health.gov.bz/www/attachments/305\_National\%20Health\%20Information\%20System\%20Strate}\\ \underline{gic\%20Plan\%202010-2014.pdf}.$ 

- 76. A. Robertson, D.W. Bates and A. Sheikh, "The rise and fall of England's National Programme for IT," *Journal of the Royal Society of Medicine*, 2011, pp. 434-435; <a href="http://jrs.sagepub.com/content/104/11/434.full">http://jrs.sagepub.com/content/104/11/434.full</a>, 16/04/2016.
- 77. A. Al-Aswad, S. Brownsell, R. Palmer and J.P. Nichol, "A Review Paper of the Current Status of Electronic Health Records Adoption Worldwide: The Gap between Developed and Developing Countries," *Journal of Health Informatics in Developing Countries*, vol. 7, no. 153-164, 2013; http://www.jhidc.org/index.php/jhidc/article/view/106, 16/04/2016.
- 78. E. Coiera, J. Aarts and C. Kulikowski, "The Dangerous Decade," *Journal of the American Medical Informatics Association*, vol. 19, 2012, pp. 05-Feb; DOI 10.1136/amiajnl-2011-000674; <a href="http://jamia.oxfordjournals.org/content/jaminfo/19/1/2.full.pdf">http://jamia.oxfordjournals.org/content/jaminfo/19/1/2.full.pdf</a>, 16/04/2016.
- 79. P. Waterson, "Health information technology and sociotechnical systems: A progress report on recent developments within the UK National Health Service (NHS)," *Elsevier, Applied Ergonomics*, vol. 45, 2013, pp. 150 161; DOI 10.1016/j.apergo.2013.07.004; <a href="http://www.sciencedirect.com/science/article/pii/S0003687013001440">http://www.sciencedirect.com/science/article/pii/S0003687013001440</a>, 16/04/2016.
- 80. E. Asamoah-Odei, D. Kebede, C. Zielinski, E.-W. Soumbey-Alley, M. Peixoto and M. Moeti, "Leveraging eHealth To Improve National Health Systems In The African Region," *WHO*, *The African Health Monitor*, no. 14, 2011, pp. 46-52; <a href="http://www.aho.afro.who.int/en/ahm/issue/14/reports/leveraging-ehealth-improve-national-health-systems-african-region">http://www.aho.afro.who.int/en/ahm/issue/14/reports/leveraging-ehealth-improve-national-health-systems-african-region</a>, 16/04/2016.
- 81. ISO/IEC, "Systems and software engineering Architecture description is an international standard for architecture descriptions of systems and software," 2011; http://en.wikipedia.org/wiki/ISO/IEC\_42010, 16/04/2016.
- 82. T.C. Bowden, "EHR Strategy: Top Down, Bottom Up or Middle Out?," *International Perspectives in Health Informatics*, vol. 164, 2011, pp. 138-142; DOI 10.3233/978-1-60750-709-3-138; http://ebooks.iospress.nl/publication/13886, 16/04/2016.
- N. Leon, H. Schneider and E. Daviaud, "Applying a framework for assessing the health system challenges to scaling up mHealth in South Africa," *Medical Informatics and Decision Making*, vol. 12, no. 123, 2012; http://www.biomedcentral.com/1472-6947/12/123;, 16/04/2016.
- 84. NAHIT, "The National Alliance for Health Information Technology Report to the Office of the National Coordinator for Health Information Technology on Defining Key Health Information

Technology Terms," *USA National Coordinator for Health Information Technology*, 2008, pp. 1-40; <a href="https://www.nachc.com/client/Key%20HIT%20Terms%20Definitions%20Final\_April\_2008.pdf">https://www.nachc.com/client/Key%20HIT%20Terms%20Definitions%20Final\_April\_2008.pdf</a>, 16/04/2016.

- 85. J. Halamka, J.M. Overhage, L. Ricciardi, W. Rishel, C. Shirky and C. Diamond, "Exchanging Health Information: Local Distribution, National Coordination," *Health Affairs*, vol. 24, no. 5, 2005, pp. 1170-1179; DOI 10.1377/hlthaff.24.5.1170; http://content.healthaffairs.org/content/24/5/1170.full.pdf, 16/04/2016.
- 86. R.S. Rudin, S.R. Simon, L.A. Volk, M. Tripathi and D. Bates, "Understanding the Decisions and Values of Stakeholders in Health Information Exchanges: Experiences From Massachusetts," *American Journal of Public Health*, vol. 99, no. 5, 2009, pp. 950-955; <a href="http://www.ncbi.nlm.nih.gov/pmc/articles/PMC2667860/pdf/950.pdf">http://www.ncbi.nlm.nih.gov/pmc/articles/PMC2667860/pdf/950.pdf</a>, 16/04/2016.
- 87. J. Covich, J. Diane R Jones, G. Morris and M. Bates, "Governance models for health information exchange," *Truven Health Analytics*, 2011, pp. 1-11; <a href="https://truvenhealth.com/portals/0/assets/GOV\_11558\_0712\_HIE\_Governance\_WP\_WEB.pdf">https://truvenhealth.com/portals/0/assets/GOV\_11558\_0712\_HIE\_Governance\_WP\_WEB.pdf</a>, 16/04/2016.
- 88. M. Lankhorst, *Enterprise Architecture at Work: Modelling, Communication and Analysis*, Springer, Science & Business Media, 2009, p. 352.
- 89. E. K, D. M, W. P, T. D and T. A., "Bottom-up and middle-out approaches to electronic patient information systems: a focus on healthcare pathways," *Informatics in Primary Care*, vol. 20, no. 1, 2012, pp. 51-56; <a href="http://www.ncbi.nlm.nih.gov/pubmed/23336835">http://www.ncbi.nlm.nih.gov/pubmed/23336835</a>, 16/04/2016.
- 90. J.M. Overhage, "Health Information Exchange: 'Lex Parsimoniae'," *Health Affairs*, vol. 26, no. 5, 2007; <a href="http://content.healthaffairs.org/cgi/pmidlookup?view=long&pmid=17670779">http://content.healthaffairs.org/cgi/pmidlookup?view=long&pmid=17670779</a>, 16/04/2016.
- 91. T. Bowden and E. Coiera, "Comparing New Zealand's 'Middle Out' health information technology strategy with other OECD nations," *International Journal of Medical Informatics*, vol. 82, no. 5, 2012; <a href="http://www.ijmijournal.com/article/S1386-5056(12)00243-2/abstract">http://www.ijmijournal.com/article/S1386-5056(12)00243-2/abstract</a>, 16/04/2016.
- 92. United States Department of Health and Human Services, "Federal Health It Strategic Plan 2015 2020," 2015; <a href="https://www.healthit.gov/sites/default/files/9-5-federalhealthitstratplanfinal\_0.pdf">https://www.healthit.gov/sites/default/files/9-5-federalhealthitstratplanfinal\_0.pdf</a>, 16/04/2016.
- 93. UK Department of Health, *The future of the National Programme for IT*, 2010; <a href="http://webarchive.nationalarchives.gov.uk/20130107105354/http://www.dh.gov.uk/en/MediaCentre/Pressreleases/DH\_119293">http://webarchive.nationalarchives.gov.uk/20130107105354/http://www.dh.gov.uk/en/MediaCentre/Pressreleases/DH\_119293</a>, 16/04/2016.

- 94. Integrated Health Systems Strengthening Project (IHSSP), "The Rwandan Health Management Information System: Improving Collection and Management of Health Service Data to Support Informed Decision Making," *Management Sciences for Health*, 2014, pp. 1-6; <a href="https://www.msh.org/resources/the-rwandan-health-management-information-system-improving-collection-and-management-of">https://www.msh.org/resources/the-rwandan-health-management-information-system-improving-collection-and-management-of</a>, 16/04/2016.
- 95. A. Wagstaff, "Poverty and health sector inequalities," *Bulletin of the World Health Organization*, vol. 80, no. 2, 2002; http://www.ncbi.nlm.nih.gov/pmc/articles/PMC2567730/pdf/11953787.pdf, 16/04/2016.
- 96. S. Stansfield, N. Orobaton, D. Lubinski, S. Uggowitzer and H. Mwanyika, "The Case for a National Health Information System Architecture; a Missing Link to Guiding National Development and Implementation," *WHO*, 2008; <a href="http://www.hrhresourcecenter.org/node/2600">http://www.hrhresourcecenter.org/node/2600</a>, 16/04/2015.
- 97. R. Foster, "eHealth on the African Continent," 2012; <a href="www.hiwiki.org">www.hiwiki.org</a>, 16/04/2016.
- 98. J. Braa and C. Hedberg, "The Struggle for District-Based Health Information Systems in South Africa," *The Information Society*, vol. 18, no. 2, 2002, pp. 113-127; https://pdfs.semanticscholar.org/8a65/dc637d39c14323dccd5cbcc08eed2553880e.pdf, 16/04/2016.
- 99. J. Braa and S. Sahay, "Integrated health information architecture: power for the users: design development and use," *Matrix Publishers*, 2012; <a href="https://www.mn.uio.no/ifi/english/research/networks/hisp/integrated-health-information-architecture/prelims.pdf">https://www.mn.uio.no/ifi/english/research/networks/hisp/integrated-health-information-architecture/prelims.pdf</a>, 16/04/2016.
- 100. A.D. Bakar, Y.H. Sheikh and A.B.M.D. Sultan, "Opportunities and Challenges of Open Source Software Integration in Developing Countries: Case of Zanzibar Health Sector," *Journal of Health Informatics in Developing Countries*, vol. 6, no. 2, 2012, pp. 443-453; <a href="http://www.jhidc.org/index.php/jhidc/article/view/84">http://www.jhidc.org/index.php/jhidc/article/view/84</a>, 16/04/2016.
- 101. C.J. Seebregts, B.W. Mamlin, P.G. Biondich, H.S. Fraser, B.A. Wolfe, D. Jazayeri, . . . OpenMRS Implementers Network, "The OpenMRS Implementers Network," *International Journal of Medical Informatics*, vol. 78, no. 11, 2009, pp. 711-720; <a href="http://www.sciencedirect.com/science/article/pii/S1386505608001652">http://www.sciencedirect.com/science/article/pii/S1386505608001652</a>, 16/04/2016.
- 102. B.W. Mamlin, P.G. Biondich, B.A. Wolfe, H. Fraser, D. Jazayeri, C. Allen, . . . W.M. Tierney, "Cooking up an open source EMR for developing countries: OpenMRS a recipe for successful collaboration," *Proc. AMIA Annual Symposium*, 2006, pp. 529-533; http://www.ncbi.nlm.nih.gov/pmc/articles/PMC1839638/, 16/04/2016.

- 103. J. Braa, A.S. Kanter, N. Lesh, R. Crichton, B. Jolliffe, J. Sæbø, . . . C.J. Seebregts, "Comprehensive Yet Scalable Health Information Systems for Low Resource Settings: A Collaborative Effort in Sierra Leone," *Proc. AMIA Annual Symposium*, 2010, pp. 372–376; <a href="http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3041283/">http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3041283/</a>, 16/04/2016.
- 104. South African National Department of Health (SANDOH), "District Health Management Information System (DHMIS) Policy," 2011; <a href="http://www.hst.org.za/publications/district-health-management-information-system-policy">http://www.hst.org.za/publications/district-health-management-information-system-policy</a>, 16/04/2016.
- 105. A. Garrib, N. Stoops, A. McKenzie, L. Dlamini, T. Govender, J. Rohde and K. Herbst, "An evaluation of the District Health Information System in rural South Africa," *South African Medical Journal*, vol. 98, no. 7, 2008, pp. 549-552; <a href="http://www.samj.org.za/index.php/samj/article/view/426">http://www.samj.org.za/index.php/samj/article/view/426</a>, 16/04/2016.
- 106. M. Kawonga, D. Blaauw and S. Fonn, "Aligning vertical interventions to health systems: a case study of the HIV monitoring and evaluation system in South Africa," *Health Research Policy and Systems*, vol. 10, no. 12, 2012; http://www.health-policy-systems.com/content/10/1/2, 16/04/2016.
- 107. W.M. Tierney, M. Achieng, E. Baker, A. Bell, P. Biondich, P. Braitstein, . . . T.T.-U.O. Consortium, "Experience Implementing Electronic Health Records in Three East African Countries," *IOS Press, MEDINFO 2010*, vol. 160, 2010, pp. 371-375; DOI 10.3233/978-1-60750-588-4-371; <a href="http://ebooks.iospress.nl/publication/13459">http://ebooks.iospress.nl/publication/13459</a>, 16/04/2016.
- 108. Rockefeller Foundation, From Silos to Systems: An Overview of eHealth's Transformative Power, 2010.
- 109. P. Littlejohns, J.C. Wyatt and L. Garvican, "Evaluating computerised health information systems: hard lessons still to be learnt," *BMJ*, vol. 326, 2006, pp. 860-863; DOI 10.1136/bmj.326.7394.860; <a href="http://bmj.com/cgi/content/full/326/7394/860">http://bmj.com/cgi/content/full/326/7394/860</a> http://www.bmj.com/content/bmj/326/7394/860.full.pdf, 16/04/2016.
- 110. Bill and Melinda Gates Foundation, "Operations Research on Improving Paper-based Information Systems for Child Health: Request for Application," 2015; <a href="http://www.gatesfoundation.org/How-We-Work/Quick-Links/Grants-Database/Grants/2015/11/OPP1135947">http://www.gatesfoundation.org/How-We-Work/Quick-Links/Grants-Database/Grants/2015/11/OPP1135947</a>, 16/04/2016.
- 111. D.M. Mugo and D. Nzuki, "Determinants of Electronic Health in Developing Countries," *International Journal of Arts and Commerce*, vol. 3, no. 3, 2014, pp. 49-60; <a href="http://www.ku.ac.ke/schools/business/images/stories/research/determinants\_of\_electronic\_health.pdf">http://www.ku.ac.ke/schools/business/images/stories/research/determinants\_of\_electronic\_health.pdf</a>, 16/04/2016.

- 112. C.J. Seebregts and e. al, "System Architectural Framework for Health in Low Resource Settings. A Systems Architecture Approach," 2012.
- 113. Vital Wave Consulting (VWC), "Health Information Systems in Developing Countries; A Landscape Analysis," *Peru Ministry of Health*, 2009; <a href="http://www.minsa.gob.pe/ogei/conferenciaops/Recursos/43.pdf">http://www.minsa.gob.pe/ogei/conferenciaops/Recursos/43.pdf</a>, 16/04/2016.
- 114. H.C. Kimaroa and J.L. Nhampossab, "The challenges of sustainability of health information systems in developing countries: comparative case studies of Mozambique and Tanzania," *Journal of Health Informatics in Developing Countries*, vol. 1, no. 1, 2007; <a href="http://www.jhidc.org/index.php/jhidc/article/view/6">http://www.jhidc.org/index.php/jhidc/article/view/6</a>, 16/04/2016.
- 115. J. Segole, "South African Government Interoperability Framework: Using Enterprise Architecture to achieve Interoperability," *Global Meeting on Government Interoperability Frameworks 2010, Rio de Janeiro*, 2010; <a href="http://www.gif4dev.net/wp-content/uploads/2010/05/South-Africa-Julius-Segole.pdf">http://www.gif4dev.net/wp-content/uploads/2010/05/South-Africa-Julius-Segole.pdf</a>, 16/04/2016.
- 116. M.C. Were and E.M. Meslin, "Ethics of Implementing Electronic Health Records in Developing Countries: Points to Consider," *AMIA Annual Symposium*, 2011, pp. 1499-1505; http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3243215/pdf/1499\_amia\_2011\_proc.pdf, 16/04/2016.
- 117. R.V. Weeks, "The implementation of an electronic patient healthcare record system: a South African case study," *Journal of Contemporary Management*, vol. 11, 2014, pp. 101 119; <a href="http://repository.up.ac.za/bitstream/handle/2263/40963/Weeks\_Implementation\_2014.pdf?sequence=1">http://repository.up.ac.za/bitstream/handle/2263/40963/Weeks\_Implementation\_2014.pdf?sequence=1</a> &isAllowed=y, 16/04/2016.
- 118. M.O. Akanbi, A.N. Ocheke, P.A. Agaba, C.A. Daniyam, E.I. Agaba, E.N. Okeke and C.O. Ukoli, "Use of Electronic Health Records in sub-Saharan Africa: Progress and challenges," *Journal of Tropical Medicine*, vol. 14, no. 1, 2012, pp. 06-Jan; <a href="http://www.ncbi.nlm.nih.gov/pmc/articles/PMC4167769/">http://www.ncbi.nlm.nih.gov/pmc/articles/PMC4167769/</a>, 16/04/2016.
- 119. A. Booz, Hamilton,, "Canada Health Infoway's 10 year Investment Strategy," *Canada Health Infoway*, 2005; <a href="https://www.infoway-inforoute.ca/en/component/edocman/17-canada-health-infoway-s-10-year-investment-strategy-executive-summary/view-document">https://www.infoway-inforoute.ca/en/component/edocman/17-canada-health-infoway-s-10-year-investment-strategy-executive-summary/view-document</a>, 16/04/2016.
- 120. A.D. Black, J. Car, C. Pagliari, C. Anandan, K. Cresswell, T. Bokun, . . . A. Sheikh, "The Impact of eHealth on the Quality and Safety of Health Care: A Systematic Overview," *PLOS Medicine*, vol. 8, no. 1, 2011; DOI 10.1371/journal.pmed.1000387; http://journals.plos.org/plosmedicine/article?id=10.1371/journal.pmed.1000387, 16/04/2016.

- 121. T. Lohman, "Australian e-health spending to top \$2 billion in 2010"; <a href="http://www.computerworld.com.au/article/343220/australian\_e-health\_spending\_top\_2\_billion\_2010/">http://www.computerworld.com.au/article/343220/australian\_e-health\_spending\_top\_2\_billion\_2010/</a>, 16/04/2016.
- 122. T. Hafner and J. Shiffman, "The emergence of global attention to health systems strengthening," *Health Policy and Planning*, vol. 28, 2013, pp. 41–50; DOI 10.1093/heapol/czs023; <a href="http://heapol.oxfordjournals.org/content/28/1/41.full.pdf">http://heapol.oxfordjournals.org/content/28/1/41.full.pdf</a>.
- 123. J.M. Kirigia, B.M. Nganda, C.N. Mwikisa and B. Cardoso, "Effects of global financial crisis on funding for health development in nineteen countries of the WHO African Region," *International Health and Human Rights*, vol. 11, no. 4, 2011; http://bmcinthealthhumrights.biomedcentral.com/articles/10.1186/1472-698X-11-4, 16/04/2016.
- 124. S. Mofleh, M. Wanous and P. Strachan, "Developing Countries and ICT Initiatives: Lessons learnt from Jordan's experience," *The Electronic Journal on Information Systems in Developing Countries*, vol. 34, no. 5, 2008; <a href="http://www.ejisdc.org/ojs2/index.php/ejisdc/article/view/482">http://www.ejisdc.org/ojs2/index.php/ejisdc/article/view/482</a>, 16/04/2016.
- 125. P. Shvaiko, A. Villafiorita, A. Zorer, L. Chemane, T.o. Fumo and a.J. Hinkkanen, "eGIF4M: eGovernment Interoperability Framework for Mozambique," *Springer, Electronic Government, Lecture Notes in Computer Science* vol. 5693, 2009, pp. 328–340; <a href="http://link.springer.com/chapter/10.1007%2F978-3-642-03516-6\_28">http://link.springer.com/chapter/10.1007%2F978-3-642-03516-6\_28</a>, 16/04/2016.
- 126. F. Bukachi and N. Pakenham-Walsh, "Information Technology for Health in Developing Countries," *Elsevier, Chest*, vol. 132, no. 5, 2007, pp. 1624-1630; <a href="http://linkinghub.elsevier.com/retrieve/pii/S0012-3692(15)51280-4">http://linkinghub.elsevier.com/retrieve/pii/S0012-3692(15)51280-4</a>, 16/04/2016.
- 127. World Health Organization (WHO), "Everybody's Business: Strengthening Health Systems to Improve Health Outcomes: WHO's Framework for Action," 2007; <a href="http://www.who.int/healthsystems/strategy/everybodys\_business.pdf">http://www.who.int/healthsystems/strategy/everybodys\_business.pdf</a>, 16/04/2016.
- 128. S. Sahay, M. Aanestad and E. Monteiro, "Configurable Politics and Asymmetric Integration: Health e-Infrastructures in India," *Journal of the Association for Information Systems*, vol. 10, no. 5, 2009, pp. 399-414; http://aisel.aisnet.org/jais/vol10/iss5/4/, 16/04/2016.
- 129. N. Archangel, "The critical issues affecting the introduction of Health Management Information Systems in developing countries in Africa," *International Institute for Communication and Development*, 2007; <a href="http://www.bibalex.org/search4dev/document/288388">http://www.bibalex.org/search4dev/document/288388</a>, 16/04/2016.
- 130. R. Canlas, "PHIS: The Philippine Health Information System: Critical Challenges and Solutions," *CCLFI*, 2010; <a href="http://www.cclfi.org/files/PHIS.pdf">http://www.cclfi.org/files/PHIS.pdf</a>, 16/04/2016.

- 131. V.D. Ndou, "e-Government for Developing Countries: Opportunities and Challenges," *The Electronic Journal of Information Systems in Developing Countries*, vol. 18, no. 1, 2004, pp. 1-24; http://www.ejisdc.org/Ojs2/index.php/ejisdc/article/view/110, 16/04/2016.
- 132. S. Syed-Abdul, J. Scholl, P. Lee, W.-S. Jian, D.-M. Liou and Y.-C. Li, "Study on the potential for delay tolerant networks by health workers in low resource settings," *Elsevier, Computer Methods Programs in Biomedicine*, vol. 107, no. 3, 2011, pp. 557-564; <a href="http://www.sciencedirect.com/science/article/pii/S0169260711003099">http://www.sciencedirect.com/science/article/pii/S0169260711003099</a>, 16/04/2016.
- 133. Y. Pillay, D. McCoy and B. Asia, "The District Health System in South Africa: Progress made and next steps," *Western Cape Government*, 2010, pp. 1-8; <a href="https://www.westerncape.gov.za/text/2003/district\_health\_system\_sa.pdf">https://www.westerncape.gov.za/text/2003/district\_health\_system\_sa.pdf</a>, 16/04/2016.
- 134. T.J. Bossert and J.C. Beauvais, "Decentralisation of health systems in Ghana, Zambia, Uganda and the Philippines: a comprehensive analysis of decision space," *Health Policy and Planning*, vol. 17, no. 14-31, 2002;

http://www1.worldbank.org/publicsector/decentralization/Feb2004Course/Background%20materials/Bossert2.pdf, 16/04/2016.

- 135. J. EFFAH and A. ABUOSI, "Standardizing a developing country health information system through proprietary software: Ghana's experience," *Journal of Health Informatics in Developing Countries*, vol. 7, no. 2, 2013, 01/01/2015.
- 136. J. Braa and H. Muquinge, "Building collaborative networks in Africa on health information systems and open source software development Experience from the HISP/BEANISH network," *Researchgate*, 2007, pp. 1-5;

https://www.researchgate.net/publication/228633133\_Building\_collaborative\_networks\_in\_Africa\_on\_health\_information\_systems\_and\_open\_source\_software\_development-experiences\_from\_the\_HISPBEANISH\_Network, 16/04/2016.

- 137. H. Mwanyika, D. Lubinski, R. Anderson, K. Chester, M. Makame, M. Steele and D.d. Savigny, "Rational Systems Design for Health Information Systems in Low-Income Countries: An Enterprise Architecture Approach," *Journal of Enterprise Architecture*, 2011, pp. 60-69; https://homes.cs.washington.edu/~anderson/docs/2011/hm2011.pdf, 16/04/2016.
- 138. R. Haux, "Health information systems past, present, future," *Elsevier, International Journal of Medical Informatics*, vol. 75, 2006, pp. 268-281; http://linkinghub.elsevier.com/retrieve/pii/S1386-5056(05)00159-0, 16/04/2016.

- World Health Organization (WHO), "Ehealth Solutions In The African Region: Current Context And Perspectives," 2010; <a href="http://apps.who.int/iris/handle/10665/19931">http://apps.who.int/iris/handle/10665/19931</a>, 16/04/2016.
- 140. D. Moodley, C.J. Seebregts, A.W. Pillay and T. Meyer, "An ontology-driven modeling platform for regulating eHealth interoperability in low resource settings," *Proc. Third International Symposium FHIES*, Springer, Foundations of Health Informatics Engineering and Systems, Lecture Notes in Computer Science, 2013, pp. 107-124; <a href="https://www.researchgate.net/publication/262912338">https://www.researchgate.net/publication/262912338</a> An ontology-driven modeling platform for regulating eHealth interoperability in low resource settings, 16/04/2016.
- 141. S.Y. Kwankama, "Successful partnerships for international collaboration in e-health: the need for organized national infrastructures," *Bulletin of the World Health Organization*, vol. 90, no. 5, 2012, pp. 395–397; <a href="http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3341703/pdf/BLT.12.103770.pdf">http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3341703/pdf/BLT.12.103770.pdf</a>, 16/04/2016.
- 142. G.B. Fanta, L. Pretorius and L. Erasmus, "An evaluation of ehealth systems implementation frameworks for sustainability in resource constrained environments: A literature review," *International Association for Management of Technology (IAMOT) 2015 Conference Proceedings*, 2015, pp. 1046 1063; <a href="http://www.iamot2015.com/2015proceedings/documents/P130.pdf">http://www.iamot2015.com/2015proceedings/documents/P130.pdf</a>, 16/04/2016.
- 143. D.d. Savigny and T. Adam, "Systems Thinking for Health Systems Strengthening," *WHO*, *Alliance for Health Policy and Systems Research*, 2009; <a href="http://www.who.int/alliance-https
- 144. T. Adam and D.d. Savigny, "Systems thinking for strengthening health systems in LMICs: need for a paradigm shift," *Health Policy and Planning*, no. 27, 2012, pp. iv1–iv3; DOI 10.1093/heapol/czs084; http://heapol.oxfordjournals.org/, 16/04/2016.
- 145. P. Saha, "Enterprise Architecture As Platform For Connected Government; Advancing the Whole of Government Enterprise Architecture Adoption with Strategic (Systems) Thinking," *Phase 2 Report, Singapore: NUS Institute of Systems Science*, 2010; http://unpan1.un.org/intradoc/groups/public/documents/unpan/unpan041801.pdf, 16/04/2016.
- 146. J. Lapalma, "3 Schools of Enterprise Architecture," *IEEE Computer Society, computer.org/ITPro*, 2011, pp. 37-43; <a href="http://www2.mitre.org/public/eabok/pdf/three\_schools\_of\_thought.pdf">http://www2.mitre.org/public/eabok/pdf/three\_schools\_of\_thought.pdf</a>, 16/04/2016.

- 147. B. Moulton, Z. Chaczko and M. Karatovic, "Updating Electronic Health Records with Information from Sensor Systems: Considerations Relating To Standards and Architecture Arising From the Development of a Prototype System," *Journal of Convergence Information Technology*, vol. 4, no. 4, 2009; DOI 10.4156/jcit.vol4.issue4.3; https://opus.lib.uts.edu.au/bitstream/10453/8967/1/2009001032OK.pdf, 16/04/2016.
- 148. W. Hammond, C. Bailey, P. Boucher, M. Spohr and P. Whitaker, "Connecting Information To Improve Health," *Health Affairs*, vol. 29, no. 2, 2010, pp. 284 288; DOI 10.1377/hlthaff.2009.0903; http://content.healthaffairs.org/content/29/2/284.full.pdf, 16/04/2016.
- 149. J.v. Gemert-Pijnen, S. Wynchank, H. Covvey and H. Ossebaard, "Improving the credibility of electronic health technologies," *Bulletin of the World Health Organization*, vol. 90, no. 5, 2012, pp. 321-400; http://www.who.int/bulletin/volumes/90/5/11-099804/en/, 16/04/2016.
- 150. M. Janssen, "Sociopolitical Aspects of Interoperability and Enterprise Architecture in E-Government," *The Journal of Industrial Relations*, vol. 30, no. 1, 2012; DOI 10.1177/0894439310392187; <a href="http://ssc.sagepub.com/content/30/1/24">http://ssc.sagepub.com/content/30/1/24</a>, 16/04/2016.
- 151. N. Anderson, "Top Down or Bottom Up Approaches to Change"; <a href="http://thecrispianadvantage.com/2010/08/15/top-down-ro-bottom-up-approaches-to-change/">http://thecrispianadvantage.com/2010/08/15/top-down-ro-bottom-up-approaches-to-change/</a>, 16/04/2016.
- 152. R. Miller and R. Cherinka, "Engineering a Complex Information Enterprise: A Case Study Architecting the Department of Defense Hourglass," *The MITRE Corporation* 2013, pp. 1-8; <a href="https://www.mitre.org/sites/default/files/pdf/07\_0215.pdf">https://www.mitre.org/sites/default/files/pdf/07\_0215.pdf</a>, 16/04/2016.
- 153. J.W. Ross, P. Weill and D.C. Robertson, "Enterprise Architecture as Strategy: Creating a foundation for business execution," *Harvard Business School Press*, 2006; <a href="http://books.google.co.za/books?id=ng3AbVQlEncC&printsec=frontcover&dq=Enterprise+Architecture+as+strategy&hl=en&sa=X&ei=DC3iT5KvGMXPhAe80oS\_Aw&ved=0CEEQ6wEwAA#v=onepage&q=Enterprise%20Architecture%20as%20strategy&f=false, 16/04/2016.
- 154. World Health Organization (WHO), "Who Country Cooperation Strategy 2008-2013: South Africa," 2009.
- 155. G. Tanna, "NHIS Development in South Africa," 2015.
- 156. D. McIntyre and B. Klugman, "The Human Face of Decentralisation and Integration of Health Services: Experience from South Africa," *Elsevier, Reproductive Health Matters*, vol. 11, no. 21, 2003, pp. 108–119; <a href="http://www.jstor.org/stable/3776675">http://www.jstor.org/stable/3776675</a>, 16/04/2016.

- 157. M. Mars and C. Seebregts, "Country Case Study for e-Health South Africa," *ehealth-connection.org*, 2008, pp. 1-46; <a href="http://ehealth-connection.org/files/resources/County%20Case%20Study%20for%20eHealth%20South%20Africa.pd">http://ehealth-connection.org/files/resources/County%20Case%20Study%20for%20eHealth%20South%20Africa.pd</a> <a href="http://ehealth-connection.org/files/resources/County%20Case%20Study%20for%20eHealth%20South%20Africa.pd">http://ehealth-connection.org/files/resources/County%20Case%20Study%20for%20eHealth%20South%20Africa.pd</a> <a href="https://ehealth.connection.org/files/resources/County%20Case%20Study%20for%20eHealth%20South%20Africa.pd">https://ehealth.connection.org/files/resources/County%20Case%20Study%20for%20eHealth%20South%20Africa.pd</a> <a href="https://ehealth.connection.org/files/resources/County%20Case%20Study%20for%20eHealth%20South%20Africa.pd">https://ehealth.connection.org/files/resources/County%20Case%20Study%20for%20eHealth%20South%20Africa.pd</a> <a href="https://ehealth.connection.org/files/resources/County%20Case%20Study%20for%20eHealth%20South%20Africa.pd">https://ehealth.connection.org/files/resources/County%20Case%20Study%20for%20eHealth%20South%20Africa.pd</a> <a href="https://ehealth.connection.org/files/resources/County%20Case%20Study%20for%20eHealth%20South%20Africa.pd">https://ehealth.connection.connect
- 158. K. Kautzkyi and S.M. Tollmani, "A Perspective on Primary Health Care in South Africa," *Health Systems Trust*, 2008, pp. 17-30; <a href="http://www.hst.org.za/uploads/files/chap2\_08.pdf">http://www.hst.org.za/uploads/files/chap2\_08.pdf</a>, 16/04/2016.
- 159. B. Gaede and M. Versteeg, "The state of the right to health in rural South Africa," *South African Health Review*, 2011, pp. 99-106; https://www.wits.ac.za/media/migration/files/healthr2011a11.pdf, 16/04/2016.
- 160. South African National Department of Health (SANDOH), "eHealth Strategy South Africa," 2012.
- 161. H. Schneider, H. Hlophe and D.v. Rensburg, "Community health workers and the response to HIV/AIDS in South Africa: tensions and prospects," *Health Policy and Planning*, vol. 23, 2008, pp. 179-187; DOI 10.1093/heapol/czn006; <a href="http://heapol.oxfordjournals.org/content/23/3/179.full.pdf">http://heapol.oxfordjournals.org/content/23/3/179.full.pdf</a>, 16/04/2016.
- 162. Y. Pillay and P. Barron, "The implementation of PHC re-engineering in South Africa," *Public Health Association of South Africa*, 2012, pp. 1-6; <a href="https://www.phasa.org.za/the-implementation-of-phc-re-engineering-in-south-africa/">https://www.phasa.org.za/the-implementation-of-phc-re-engineering-in-south-africa/</a>, 16/04/2016.
- 163. South African National Department of Health (SANDOH), "Strategic Plan 2014/15 2018/19," 2014; <a href="http://www.health-e.org.za/wp-content/uploads/2014/08/SA-DoH-Strategic-Plan-2014-to-2019.pdf">http://www.health-e.org.za/wp-content/uploads/2014/08/SA-DoH-Strategic-Plan-2014-to-2019.pdf</a>, 16/04/2016.
- 164. South African Department of Communications, "Health," *South Africa Year Book* 2013/14, pp. 225-238; <a href="http://www.gcis.gov.za/sites/www.gcis.gov.za/files/docs/resourcecentre/yearbook/2013-4Health.pdf">http://www.gcis.gov.za/sites/www.gcis.gov.za/files/docs/resourcecentre/yearbook/2013-4Health.pdf</a>, 16/04/2016.
- 165. World Health Organization (WHO), "WHO Country Coorperation Strategy at a Glance," 2013; <a href="http://www.who.int/countryfocus/cooperation\_strategy/ccsbrief\_southafrica\_en.pdf?ua=1">http://www.who.int/countryfocus/cooperation\_strategy/ccsbrief\_southafrica\_en.pdf?ua=1</a>, 16/04/2016.
- 166. L. Gilson and J. Daire, "Leadership and Governance within the South African Health System," *South African Health Review*, 2011, pp. 69-80; <a href="http://www.rmchsa.org/wp-content/resources/resources">http://www.rmchsa.org/wp-content/resources/resources</a> by theme/ClinicalGovernance/Leadership GovernanceSAHealthSystem. pdf, 16/04/2016.

- 167. D. Harrison, "An Overview of Health and Health care in South Africa 1994 2010: Priorities, Progress and Prospects for New Gains," *A Discussion Document Commissioned by the Henry J. Kaiser Family Foundation*, 2009, pp. 1-40; <a href="http://ftp.bhfglobal.com/files/bhf/overview1994-2010.pdf">http://ftp.bhfglobal.com/files/bhf/overview1994-2010.pdf</a>, 16/04/2016.
- 168. HMN, "Health and Vital Stats: Assessment of the Health Information System in South Africa," *Statistics South Africa*, 2009.
- 169. T.C. Masilela, R. Foster and M. Chetty, "The eHealth Strategy for South Africa 2012-2016: How Far Are We?," *South African Health Review* 2013/1014, pp. 15-24; <a href="http://www.hst.org.za/publications/ehealth-strategy-south-africa-2012-2016-how-far-are-we">http://www.hst.org.za/publications/ehealth-strategy-south-africa-2012-2016-how-far-are-we</a>, 16/04/2016.
- 170. C. Day, A. Gray and E. Budgell, "Health and Related Indicators," *South African Health Review*, 2011, pp. 119-248; <a href="http://www.hst.org.za/uploads/files/chap16\_08.pdf">http://www.hst.org.za/uploads/files/chap16\_08.pdf</a>, 16/04/2016.
- 171. P. Matshidze and L. Hanmer, "Health Information Systems in the Private Health Sector," *Health Systems Trust*, 2007, pp. 89-102; <a href="http://www.hst.org.za/uploads/files/chap6\_07.pdf">http://www.hst.org.za/uploads/files/chap6\_07.pdf</a>, 16/04/2016.
- 172. SARRAH, "SARRAH"; <a href="http://www.sarrahsouthafrica.org/SUPPORTFORHIVANDHEALTH/MONITORINGANDEVALUATION.aspx">http://www.sarrahsouthafrica.org/SUPPORTFORHIVANDHEALTH/MONITORINGANDEVALUATION.aspx</a>, 16/04/2016.
- 173. UnknownAuthor, "Review of health information systems (HIS) in selected countries VII: South Africa," *WHO*, 2005, pp. 1-17.
- 174. V. Shaw, E. Claasen and J. Brown, "The Changing Landscape of HIS in SA," *Journal of Health Informatics in Developing Countries*, 2013; DOI 10.12856/JHIA-2013-v1-i1-81; <a href="http://jhia-online.org/index.php/jhia/article/download/81/6">http://jhia-online.org/index.php/jhia/article/download/81/6</a>, 16/04/2016.
- 175. South African National Treasury, "2015 Public Sector Supply Chain Management Review," 2015; <a href="http://www.treasury.gov.za/publications/other/SCMR%20REPORT%202015.pdf">http://www.treasury.gov.za/publications/other/SCMR%20REPORT%202015.pdf</a>, 16/04/2016.
- 176. National Health Laboratory Service, "National Health Laboratory Service," <a href="www.nhls.ac.za">www.nhls.ac.za</a>, 16/04/2016.
- 177. Western Cape Government: Health, "Annual Report 2012/2013," 2013; <a href="http://www.wcpp.gov.za/sites/default/files/Health\_reduced%5B1%5D.pdf">http://www.wcpp.gov.za/sites/default/files/Health\_reduced%5B1%5D.pdf</a>, 13/04/2016.

- 178. J. Peter, P. Barron and Y. Pillay, "Using mobile technology to improve maternal, child and youth health and treatment of HIV patients," *South African Medical Journal*, vol. 106, no. 1, 2015, pp. 3-4; <a href="http://dx.doi.org/10.7196/samj.2016.v106i1.10209.">http://dx.doi.org/10.7196/samj.2016.v106i1.10209.</a>, 13/04/2016.
- 179. M. Wolmarans, W. Solomon, G. Tannai, M. Chetty, M. Domboiv, J. Venter and A. Parsonsii, "eHealth Programme reference implementation in primary health care facilities," *South African Health Review*, 2015, pp. 35-44; <a href="http://www.hst.org.za/publications/ehealth-programme-reference-implementation-primary-health-care-facilities">http://www.hst.org.za/publications/ehealth-programme-reference-implementation-primary-health-care-facilities</a>, 16/04/2016.
- 180. A. Coleman, "An Integrated Model to Share Patient Health Records in Public and Private Hospitals in South Africa," *Ethno Med*, vol. 7, no. 2, 2013, pp. 87-93; <a href="http://www.krepublishers.com/02-Journals/S-EM/EM-07-0-000-13-Web/S-EM-07-2-000-13-Abst-PDF/S-EM-07-2-087-13-256-Coleman-A/S-EM-07-2-087-13-256-Coleman-A-Tt.pdf">http://www.krepublishers.com/02-Journals/S-EM/EM-07-0-000-13-Web/S-EM-07-2-000-13-Abst-PDF/S-EM-07-2-087-13-256-Coleman-A/S-EM-07-2-087-13-256-Coleman-A-Tt.pdf</a>, 16/04/2016.
- 181. dhmis, "dhmis"; dd.dhmis.org.
- 182. R. Crichton, "The Open Health Information Mediator: an Architecture for Enabling Interoperability in Low to Middle Income Countries," *Jembi*, 2015; <a href="http://www.jembi.org/wpcontent/uploads/2012/07/thesis-ryan-11-03-2015-FINALv3\_2.compressed.pdf">http://www.jembi.org/wpcontent/uploads/2012/07/thesis-ryan-11-03-2015-FINALv3\_2.compressed.pdf</a>, 16/04/2016.
- 183. South African National Department of Health (SANDOH), *Towards A National Health System*, 1997; <a href="http://www.gov.za/sites/www.gov.za/files/17910\_gen667\_0.pdf">http://www.gov.za/sites/www.gov.za/files/17910\_gen667\_0.pdf</a>, 16/04/2016.
- 184. N. Mbananga, "The National Strategic Framework For EHR Implementation South Africa," 2006.
- 185. M.P. Matsoso and R. Fryatt, "National Health Insurance: The first 18 months," *South African Medical Journal*, vol. 103, no. 3, 2013, pp. 156-158; DOI 10.7196/SAMJ.6601; <a href="http://www.samj.org.za/index.php/samj/article/viewFile/6601/4919">http://www.samj.org.za/index.php/samj/article/viewFile/6601/4919</a>, 16/04/2016.
- 186. C.J. Seebregts, "NHIS Development in Sub-Saharan Africa (on going discussions)," 2015.
- 187. M. Osler, K. Hilderbrand, C. Hennessey, J. Arendse, E. Goemaere, N. Ford and A. Boulle, "A three-tier framework for monitoring antiretroviral therapy in high HIV burden settings," *Journal of the International AIDS Society*, vol. 17, no. 18908, 2014; <a href="http://www.jiasociety.org/index.php/jias/article/view/18908">http://www.jiasociety.org/index.php/jias/article/view/18908</a>, 16/04/2016.
- 188. GITOCS, "Government-Wide Enterprise Architecture (GWEA) Framework," *Government Information Technology Officer's Council (GITOC) of South Africa* 2009, pp. 1-62; <a href="http://gissa.org.za/special-interest-groups/open-source/foss-">http://gissa.org.za/special-interest-groups/open-source/foss-</a>

<u>documents/GWEA%20Framework%20Implementation%20Guide%20v1.2%20Jun%202010.pdf</u>, 16/04/2016.

- 189. K. Sullivan, W. Knaus and R. Marks, "An Ultra-Large-Scale Systems Approach to National-Scale Health Information Systems," *ACM*, 2010, pp. 365-367; <a href="https://pdfs.semanticscholar.org/c92a/eb89ab99aa48cd6aafc48ceb71f642e6026c.pdf">https://pdfs.semanticscholar.org/c92a/eb89ab99aa48cd6aafc48ceb71f642e6026c.pdf</a>, 16/04/2016.
- 190. World Health Organization (WHO), "WHO Country Cooperation Strategy 2009-2013 Rwanda," 2009; http://www.who.int/countryfocus/cooperation\_strategy/ccs\_rwa\_en.pdf, 16/04/2016.
- 191. IFAD, "Republic of Rwanda Country strategic opportunities programme," 2013, pp. 1-39; https://www.ifad.org/documents/10180/a15a8b27-c501-4e2a-bbbd-ee28d0e367a1, 16/04/2016.
- 192. P.C. Drobac, P. Basinga, J. Condo, P.E. Farmer, K.E. Finnegan, J.K. Hamon, . . . A. Binagwaho, "Comprehensive and integrated district health systems strengthening: the Rwanda Population Health Implementation and Training (PHIT) Partnership," *BMC Health Services Research*, vol. 13(Suppl 2), 2013; http://www.biomedcentral.com/1472-6963/13/S2/S5, 16/04/2016.
- 193. Rwanda Ministry of Health, PMNCH, WHO, World Bank, AHPSR and participants in the Rwanda multistakeholder policy review, "Success Factors for Women's and Children's Health: Rwanda," *WHO*, 2014; http://www.who.int/pmnch/knowledge/publications/rwanda country report.pdf, 16/04/2016.
- 194. S. Pavalam, M. Jawahar and F. Akorli, "Data Warehouse based Architecture for Electronic Health Records for Rwanda," *IEEE Xplore*, 2010, pp. 253-255; DOI 10.1109/ICEMT.2010.5657660; <a href="http://ieeexplore.ieee.org/xpl/login.jsp?tp=&arnumber=5657660&url=http%3A%2F%2Fieeexplore.ieee.org%2Fxpls%2Fabs\_all.jsp%3Farnumber%3D5657660">http://ieeexplore.ieee.org/xpl/login.jsp?tp=&arnumber=5657660&url=http%3A%2F%2Fieeexplore.ieeeorg%2Fxpls%2Fabs\_all.jsp%3Farnumber%3D5657660</a>, 16/04/2016.
- 195. R. Gakuba, "The National E-Health Strategic Plan 2009-2013," *Rwanda Ministry of Health*, 2009, pp. 1-53; <a href="https://www.isfteh.org/files/media/rwanda\_national\_ehealth\_strategy\_2009-2013.pdf">https://www.isfteh.org/files/media/rwanda\_national\_ehealth\_strategy\_2009-2013.pdf</a>, 16/04/2016.
- 196. Rwanda Ministry of Health, "Rwanda Health Sector Strategic Plan 2012–2018," 2012.
- 197. R. Pose and F. Samuels, "Rwanda's progress in health: Leadership, performance and insurance," *ODI publications*, 2011;

http://www.developmentprogress.org/sites/developmentprogress.org/files/rwanda\_report\_-\_master\_0.pdf, 16/04/2016.

- 198. L.M. Fox, N. Ravishankar, J. Squires, R.T. Williamson and D. Brinkerhoff, "Rwanda Health Governance Assessment," *USAID*, 2010, pp. 1-44; <a href="http://apps.who.int/medicinedocs/documents/s18413en.pdf">http://apps.who.int/medicinedocs/documents/s18413en.pdf</a>, 16/04/2016.
- 199. World Health Organization (WHO), "WHO Country Cooperation Strategy," 2009-2013; http://www.who.int/countryfocus/cooperation\_strategy/ccs\_rwa\_en.pdf, 16/04/2016.
- 200. M. Nisingizwe, H. Iyer, M. Gashayija, L. Hirschhorn, C. Amoroso, R. Wilson, . . . B. Hedt-Gauthier, "Toward utilization of data for program management and evaluation: quality assessment of five years of health management information system data in Rwanda," *Global Health Action*, vol. 7, no. 25829, 2014, pp. 1-5; DOI 10.3402/gha.v7.25829; http://www.globalhealthaction.net/index.php/gha/article/view/25829, 16/04/2016.
- 201. K. Leuchowius, "Report on the Health Care Sector and Business Opportunities in Rwanda," *Swecare Foundation*, 2014; <a href="http://www.swecare.se/Portals/swecare/Documents/Report-on-the-Health-Care-Sector-and-Business-Opportunities-in-Rwanda-Sep2014-vers2.pdf">http://www.swecare.se/Portals/swecare/Documents/Report-on-the-Health-Care-Sector-and-Business-Opportunities-in-Rwanda-Sep2014-vers2.pdf</a>, 16/04/2016.
- 202. H. Fraser, M.A. May and R. Wanchoo, "e-Health Rwanda Case Study," *The Rockefeller Foundation, ehealth-connection.org*, 2008, pp. 1-32; <a href="http://www.ehealth-connection.org/files/resources/Rwanda%20+%20Appendices.pdf">http://www.ehealth-connection.org/files/resources/Rwanda%20+%20Appendices.pdf</a>, 16/04/2016.
- 203. R. Crichton, D. Moodley, A.W. Pillay and C.J. Seebregts, "An Interoperability Architecture for the Health Information Exchange in Rwanda," *Proc. 2nd International Symposium on the Foundations of Health Information Engineering and Systems FHIES*, 2012; http://researchspace.csir.co.za/dspace/handle/10204/6453, 16/04/2016.
- 204. Rwanda Ministry of Health, "Health Sector Policy: Government of Rwanda," 2015; <a href="http://www.hrhresourcecenter.org/node/313">http://www.hrhresourcecenter.org/node/313</a>, 16/04/2016.
- 205. E. Rugomboka, "NHIS Development in Rwanda," 2015.
- 206. P. Rugumire, "New Information System Increasing Access to Health Data in Rwanda," *Global Health Action*, 2014; <a href="http://www.msh.org/news-events/stories/new-information-system-increasing-access-tohealth-data-in-rwanda">http://www.msh.org/news-events/stories/new-information-system-increasing-access-tohealth-data-in-rwanda</a>, 16/04/2016.
- 207. K. Fidèle, "Innovation developed in Rwanda to support social protection interventions," *International Labour Organisation*, 2011; <a href="http://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe">http://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe</a> <a href="http://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe">http://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe</a> <a href="https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe">https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe</a> <a href="https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe">https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe</a> <a href="https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe">https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe</a> <a href="https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe">https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe</a> <a href="https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe">https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe</a> <a href="https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe">https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe</a> <a href="https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.action;jsessionid=5817d0123cd27ce5fe</a> <a href="https://www.socialprotectionfloor.org/gimi/gess/RessourcePDF.actionfloor.org/gimi/gess/RessourcePDF.actionfloor.or

- 208. C. Lu, S. Tsai, J. Ruhumuriza, G. Umugiraneza, S. Kandamutsa, P.P. Salvatore, . . . F. Ngabo, "Tracking Rural Health Facility Financial Data in Resource-Limited Settings: A Case Study from Rwanda," *PLOS Medicine*, vol. 11, no. 12, 2014; DOI 10.1371/journal.pmed.1001763; <a href="http://www.plosmedicine.org/article/fetchObject.action?uri=info:doi/10.1371/journal.pmed.1001763&representation=PDF">http://www.plosmedicine.org/article/fetchObject.action?uri=info:doi/10.1371/journal.pmed.1001763&representation=PDF</a>, 16/04/2016.
- 209. A. Binagwaho and K.W. Scott, "Improving the world's health through the post-2015 development agenda: perspectives from Rwanda," *International Journal of Health Policy Management*, vol. 4, no. 4, 2015, pp. 203–205; DOI 10.15171/ijhpm.2015.46; http://www.ncbi.nlm.nih.gov/pmc/articles/PMC4380561/pdf/IJHPM-4-203.pdf, 16/04/2016.
- 210. OneNetwork, "Rwanda Demonstrates the Future of Public Health with Cloud-Based Healthcare Network"; <a href="http://www.onenetwork.com/2015/02/rwanda-demonstrates-future-public-health-cloud-based-healthcare-network/">http://www.onenetwork.com/2015/02/rwanda-demonstrates-future-public-health-cloud-based-healthcare-network/</a>, 16/04/2016.
- 211. F. Verbeke, L.Y. Ousmane, G. Karara and M. Nyssen, "Free And Open Source Software (Foss) For Electronic Health Record Management In Developing Countries," *Rwanda Medical Journal*, vol. 70, no. 2, 2013; http://www.bioline.org.br/abstract?id=rw13008, 16/04/2016.
- 212. Rapid SMS, "Rapid SMS"; www.rapidsms.moh.gov.rw, 16/04/2016.
- 213. Rwanda Ministry of ICT, "National ICT Strategy and Plan NICI 2015," 2012; <a href="https://www.enisa.europa.eu/activities/Resilience-and-CIIP/national-cyber-security-strategies-ncsss/RwandaNCSSNICI III.pdf">https://www.enisa.europa.eu/activities/Resilience-and-CIIP/national-cyber-security-strategies-ncsss/RwandaNCSSNICI III.pdf</a>, 16/04/2016.
- 214. R. English, T. Masilela, P. Barron and A. Schönfeldt, "Health Information Systems in South Africa," *South African Health Review*, 2011, pp. 81-90; <a href="http://www.hst.org.za/publications/south-african-health-review-2011">http://www.hst.org.za/publications/south-african-health-review-2011</a>, 16/04/2016.
- 215. D.A. McDonald and G. Ruiters, "Alternatives to privatisation; Public Options for Essential Services in Global South," 2012;

http://www.hsrcpress.ac.za/product.php?productid=2287&freedownload=1, 16/04/2016.